Kyoto University Research Info	mation Repository KYOTO UNIVERSITY
Title	The Crepant Transformation Conjecture for toric complete intersections
Author(s)	Coates, Tom; Iritani, Hiroshi; Jiang, Yunfeng
Citation	Advances in Mathematics (2018), 329: 1002-1087
Issue Date	2018-04-30
URL	http://hdl.handle.net/2433/233906
Right	© 2018 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).
Туре	Journal Article
Textversion	publisher





Contents lists available at ScienceDirect

Advances in Mathematics

www.elsevier.com/locate/aim

The Crepant Transformation Conjecture for toric complete intersections



霐

MATHEMATICS

Tom Coates^{a,*}, Hiroshi Iritani^b, Yunfeng Jiang^c

^a Department of Mathematics, Imperial College London, United Kingdom

^b Department of Mathematics, Kyoto University, Japan

 $^{\rm c}$ Department of Mathematics, University of Kansas, United States

A R T I C L E I N F O

Article history:

Received 14 November 2017 Accepted 15 November 2017 Available online 23 March 2018 Communicated by the Managing Editors

MSC:

primary 14N35 secondary 14A20, 14E16, 14F05, 53D45

Keywords: Toric Deligne–Mumford stacks Crepant Resolution Conjecture Mirror symmetry Quantum cohomology Fourier–Mukai transformation Mellin–Barnes method

ABSTRACT

Let X and Y be K-equivalent toric Deligne–Mumford stacks related by a single toric wall-crossing. We prove the Crepant Transformation Conjecture in this case, fully-equivariantly and in genus zero. That is, we show that the equivariant quantum connections for X and Y become gauge-equivalent after analytic continuation in quantum parameters. Furthermore we identify the gauge transformation involved, which can be thought of as a linear symplectomorphism between the Givental spaces for X and Y, with a Fourier–Mukai transformation between the K-groups of X and Y, via an equivariant version of the Gamma-integral structure on quantum cohomology. We prove similar results for toric complete intersections. We impose only very weak geometric hypotheses on X and Y: they can be non-compact, for example, and need not be weak Fano or have Gorenstein coarse moduli space. Our main tools are the Mirror Theorems for toric Deligne-Mumford stacks and toric complete intersections, and the Mellin-Barnes method for analytic continuation of hypergeometric functions.

© 2018 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

* Corresponding author.

E-mail addresses: t.coates@imperial.ac.uk (T. Coates), iritani@math.kyoto-u.ac.jp (H. Iritani), y.jiang@ku.edu (Y. Jiang).

https://doi.org/10.1016/j.aim.2017.11.017

0001-8708/© 2018 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

1. Introduction

A birational map $\varphi \colon X_+ \dashrightarrow X_-$ between smooth varieties, orbifolds, or Deligne– Mumford stacks is called a *K*-equivalence if there exists a smooth variety, orbifold, or Deligne–Mumford stack \widetilde{X} and projective birational morphisms $f_{\pm} \colon \widetilde{X} \to X_{\pm}$ such that $f_- = \varphi \circ f_+$ and $f_+^* K_{X_+} = f_-^* K_{X_-}$:

$$\begin{array}{c}
\widetilde{X} \\
f_{+} \\
\downarrow \\
X_{+} - - - \frac{\varphi}{-} \\
\end{array} (1.1)$$

In this case, the celebrated Crepant Transformation Conjecture of Y. Ruan predicts that the quantum (orbifold) cohomology algebras of X_+ and X_- should be related by analytic continuation in the quantum parameters. This conjecture has stimulated a great deal of interest in the connections between quantum cohomology (or Gromov-Witten theory) and birational geometry: see, for example, [9,10,17,18,20,22,23,27,40,44,52,55–58,61,67,70,74,75]. Ruan's original conjecture was subsequently refined, revised, and extended to higher genus Gromov–Witten invariants, first by Bryan–Graber [19] under some additional hypotheses, and then by Coates–Iritani–Tseng, Iritani, and Ruan in general [33,34,49]. Recall that a toric Deligne–Mumford stack X can be constructed as a GIT quotient $[\mathbb{C}^m/\!/_{\omega}K]$ of \mathbb{C}^m by an action of a complex torus K, where ω is an appropriate stability condition, and that wall-crossing in the space of stability conditions induces birational transformations between GIT quotients [36,71]. Our main result implies the CIT/Ruan version of the Crepant Transformation Conjecture in genus zero, in the case where X_+ and X_- are complete intersections in toric Deligne–Mumford stacks and $\varphi \colon X_+ \dashrightarrow X_-$ arises from a toric wall-crossing. We concentrate initially on the case where X_+ and X_- are toric, deferring the discussion of toric complete intersections to §1.3.

1.1. The toric case

We consider toric Deligne–Mumford stacks X_{\pm} of the form $[\mathbb{C}^m/\!/_{\omega}K]$, where K is a complex torus, and consider a K-equivalence $\varphi \colon X_+ \dashrightarrow X_-$ determined by a wallcrossing in the space of stability conditions ω . The action of $T = (\mathbb{C}^{\times})^m$ on \mathbb{C}^m descends to give (ineffective) actions of T on X_{\pm} , and we consider the T-equivariant Chen–Ruan cohomology groups $H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$ [25]. There is a T-equivariant big quantum product \star_{τ} on $H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$, parametrized by $\tau \in H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$ and defined in terms of T-equivariant Gromov–Witten invariants of X_{\pm} . The T-equivariant quantum connection is a pencil of flat connections:

$$\nabla = d + z^{-1} \sum_{i=0}^{N} (\phi_i \star_\tau) d\tau^i$$
(1.2)

on the trivial $H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$ -bundle over an open set in $H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$; here $z \in \mathbb{C}^{\times}$ is the pencil variable, $\tau \in H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$ is the co-ordinate on the base of the bundle, ϕ_0, \ldots, ϕ_N are a basis for $H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$, and τ^0, \ldots, τ^N are the corresponding co-ordinates of $\tau \in H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$, so that $\tau = \sum_{i=0}^N \tau^i \phi_i$.

Theorem 1.1. Let X_+ and X_- be toric Deligne–Mumford stacks, and let $\varphi \colon X_+ \dashrightarrow X_-$ be a K-equivalence that arises from a wall-crossing of GIT stability conditions. Then:

- (1) the equivariant quantum connections of X_{\pm} become gauge-equivalent after analytic continuation in τ , via a gauge transformation $\Theta(\tau, z)$: $H^{\bullet}_{\mathrm{CR},T}(X_{-}) \to H^{\bullet}_{\mathrm{CR},T}(X_{+})$ which is homogeneous of degree zero, regular at z = 0, and preserves the equivariant orbifold Poincaré pairing;
- (2) there exists a common toric blowup \widetilde{X} of X_{\pm} as in (1.1) such that gauge transformation Θ coincides with the Fourier-Mukai transformation

$$\mathbb{FM}: K^0_T(X_-) \to K^0_T(X_+) \qquad E \mapsto (f_+)_\star (f_-)^\star (E)$$

via the equivariant Gamma-integral structure introduced in §3 below.

Here:

- The Gamma-integral structure on equivariant quantum cohomology is an assignment, to each class $E \in K_T^0(X_{\pm})$ of *T*-equivariant vector bundles on X_{\pm} , of a flat section $\mathfrak{s}(E)$ for the equivariant quantum connection on X_{\pm} . This gives a lattice in the space of flat sections which is isomorphic to the integral equivariant *K*-group $K_T^0(X_{\pm})$. The flat section $\mathfrak{s}(E)$ is, roughly speaking, given by the Chern character of *E* multiplied by a characteristic class of X_{\pm} , called the $\widehat{\Gamma}$ -class, that is defined in terms of the Γ -function. Part (2) of Theorem 1.1 asserts that the flat section $\mathfrak{s}(E)$ analytically continues to $\mathfrak{s}(\mathbb{FM}(E))$.
- The gauge transformation $\Theta(\tau, z)$ will in general be non-constant: it depends on the parameter τ for the equivariant quantum product, and also on the parameter z appearing in the equivariant quantum connection. When written in terms of the integral structure, however, it becomes a constant, integral linear transformation.

Remark 1.2. Throughout this paper, when we consider K-equivalence (1.1) of Deligne– Mumford stacks X_{\pm} , $K_{X_{\pm}}$ means the canonical class as a stack; in general this is different from the (Q-Cartier) canonical divisor $K_{|X_{\pm}|}$ of the coarse moduli space $|X_{\pm}|$. In particular, we do not require the coarse moduli spaces $|X_{\pm}|$ to be Gorenstein.

Remark 1.3. Gonzalez and Woodward [44] have proved a very general wall-crossing formula for Gromov–Witten invariants under variation of GIT quotient, using gauged Gromov–Witten theory. Their result, which is a quantum version of Kalkman's wall-crossing formula, gives a complete description of how non-equivariant genus-zero Gromov–Witten invariants change under wall-crossing. Thus their theorem must imply the non-equivariant version of the first part of Theorem 1.1, and the first part of Theorem 1.4. Our methods are significantly less general — they apply only to toric stacks and toric complete intersections — but give a much more explicit relationship between the genus-zero Gromov–Witten theories.

Theorem 1.1 is slightly imprecisely stated: we give precise statements, once the necessary notation and definitions are in place, as Theorems 5.14, 6.1, and 6.3 below. We now explain how Theorem 1.1 implies the CIT/Ruan version of the Crepant Transformation Conjecture.

The CIT/Ruan version of the Crepant Transformation Conjecture is stated in terms of Givental's symplectic formalism for Gromov–Witten theory [43]. In our context, this associates to X_{\pm} the vector spaces $\mathcal{H}(X_{\pm}) := H^{\bullet}_{\mathrm{CR},T}(X_{\pm})((z^{-1}))$ equipped with a certain symplectic form, and encodes *T*-equivariant genus-zero Gromov–Witten invariants via a Lagrangian cone $\mathcal{L}_{\pm} \subset \mathcal{H}(X_{\pm})$. The Givental cone \mathcal{L}_{\pm} for X_{\pm} determines the big quantum product \star_{τ} on $H^{\bullet}_{\mathrm{CR},T}(X_{\pm})$, and vice versa. The CIT/Ruan Crepant Transformation Conjecture, made in the context of non-equivariant Gromov–Witten theory, asserts that there exists a $\mathbb{C}((z^{-1}))$ -linear grading-preserving symplectic isomorphism $\mathbb{U}: \mathcal{H}(X_{-}) \to \mathcal{H}(X_{+})$, such that after analytic continuation of \mathcal{L}_{\pm} we have $\mathbb{U}(\mathcal{L}_{-}) = \mathcal{L}_{+}$. See [33,34] for more details.

There are various subtle points in the notion of analytic continuation of the (infinitedimensional) cones \mathcal{L}_{\pm} , especially under the weak convergence hypotheses that we impose, and some necessary foundational material is missing. Thus we choose to state Theorem 1.1 in terms of the equivariant quantum connections for X_{\pm} rather than in terms of the Givental cones \mathcal{L}_{\pm} . The two formulations are very closely related, however, as we now explain. Let $L_{\pm}(\tau, z)$ denote a fundamental solution for the equivariant quantum connection ∇ , that is, a matrix with columns that give a basis of flat sections for ∇ . The assignment

$$\tau \mapsto L_{\pm}(\tau, z)^{-1} \mathcal{H}_{+} \qquad \tau \in H^{\bullet}_{\mathrm{CR}, T}(X_{\pm}) \qquad \text{where } \mathcal{H}_{+} := H^{\bullet}_{\mathrm{CR}, T}(X_{\pm}) \otimes \mathbb{C}[z]$$

gives the family of tangent spaces to the Givental cone \mathcal{L}_{\pm} . As emphasized in [33], this defines a variation of semi-infinite Hodge structure in the sense of Barannikov [5]. The Givental cone \mathcal{L}_{\pm} can be reconstructed from the semi-infinite variation as:

$$\mathcal{L}_{\pm} = \bigcup_{\tau} z L_{\pm}(\tau, z)^{-1} \mathcal{H}_{+}$$

Thus part (1) of Theorem 1.1 implies the CIT/Ruan-style Crepant Transformation Conjecture whenever it makes sense, with the symplectic transformation \mathbb{U} defined in terms

of the gauge transformation Θ by $\mathbb{U} = L_{+}^{-1}\Theta L_{-}$. The fact that \mathbb{U} is independent of τ follows from the fact that Θ is a gauge equivalence. The fact that \mathbb{U} is symplectic (or equivalently, the fact that Θ is pairing-preserving) follows from the identification, in part (2) of Theorem 1.1, of Θ with the Fourier–Mukai transformation FM. The Fourier–Mukai transformation is a derived equivalence and thus preserves the Mukai pairings on $K_T^0(X_{\pm})$; this implies, via the equivariant Hirzebruch–Riemann–Roch theorem, that Θ is pairing-preserving. The identification of Θ with FM also makes clear that the symplectic transformation \mathbb{U} has a well-defined non-equivariant limit, since the Fourier–Mukai transformation itself can be defined non-equivariantly.

In terms of the symplectic transformation \mathbb{U} , part (2) of Theorem 1.1 can be rephrased as the commutativity of the diagram

$$\begin{array}{ccc} K_T^0(X_-) & \xrightarrow{\mathbb{FM}} & K_T^0(X_+) \\ \\ & \tilde{\Psi}_- & & & & \\ & & & & \\ & & & \\ & & \widetilde{\mathcal{H}}(X_-) & \xrightarrow{\mathbb{U}} & \widetilde{\mathcal{H}}(X_+) \end{array}$$

where $\tilde{\mathcal{H}}(X_{\pm})$ is a variant of Givental's symplectic space and $\tilde{\Psi}_{\pm}$ are certain 'framing maps' built from the Gamma-integral structure: see Theorem 6.1. This identification of \mathbb{U} with a Fourier–Mukai transformation was proposed in [49]. Our results also imply Ruan's original conjecture that the quantum cohomology rings of X_{\pm} are (abstractly) isomorphic, and that the associated *F*-manifold structures are isomorphic. We refer the reader to [27,28,33,34,50] for discussions on the consequence of these conjectures and several concrete examples.

1.2. The Mellin-Barnes method and the work of Borisov-Horja

The main ingredients in the proof of Theorem 1.1 are the Mirror Theorem for toric stacks [26,29], which determines the equivariant quantum connection ∇ (or, equivalently, the Givental cone \mathcal{L}_{\pm}) in terms of a certain cohomology-valued hypergeometric function called the *I*-function, and the Mellin–Barnes method [6,21], which allows us to analytically continue the *I*-functions for X_{\pm} . From this point of view, the symplectic transformation \mathbb{U} arises as the matrix which intertwines the two *I*-functions (see Theorem 6.1):

$$\mathbb{U}I_{-}=I_{+}.$$

On the other hand, components of the *I*-function give hypergeometric solutions to the Gelfand–Kapranov–Zelevinsky (GKZ) system of differential equations. The analytic continuation of solutions to the GKZ system has been studied by Borisov–Horja [12]. They showed that, under an appropriate identification of the spaces of GKZ solutions with the

K-groups of the corresponding toric Deligne–Mumford stacks, the analytic continuation of solutions to a GKZ system is induced by a Fourier–Mukai transformation between the K-groups. Our computation may be viewed as a straightforward generalization of theirs. The differences from their situation are:

- (a) we work with a fully equivariant version, that is, the parameters β_j in the GKZ system are arbitrary and we use the equivariant K-groups (here β_j corresponds to the equivariant parameter);
- (b) we compute analytic continuation of the *I*-function corresponding to the *big* quantum cohomology; in terms of the GKZ system, we do not assume that lattice vectors in the set¹ A lie on a hyperplane of height one.

Since we work equivariantly, we can use the fixed point basis in localized equivariant cohomology to calculate the analytic continuation of the *I*-functions. It turns out that analytic continuation via the Mellin–Barnes method becomes much easier to handle in the fully equivariant setting, because we only need to evaluate residues at *simple* poles.² It is also straightforward to compute the Fourier–Mukai transformation in terms of the fixed point basis in the localized equivariant *K*-group, and hence to see that analytic continuation coincides with Fourier–Mukai.

Regarding part (b) above, we choose A to be the set $\{b_1, \ldots, b_m\} \subset \mathbb{N}$ of ray vectors of an extended stacky fan [11,53]. Since we do not restrict ourselves to the weak Fano case, and since we work with Jiang's extended stacky fans, the generic rank of the GKZ system can be bigger than the rank of $H^{\bullet}_{CR,T}(X_{\pm})$. To remedy this, we treat one special variable analytically and work formally in the other variables. In fact, the big *I*-functions are not necessarily convergent in all of the variables, and we analytically continue the *I*-function with respect to one specific variable y_r . This amounts to considering an adic completion of the Borisov–Horja better-behaved GKZ system [14] with respect to the other variables. The analytic continuation in Theorem 1.1 occurs across a "global Kähler moduli space" $\widetilde{\mathcal{M}}^{\circ}$ which is treated as an analytic space in one direction and as a formal scheme in the other directions.

1.3. The toric complete intersection case

Let $\varphi: X_+ \dashrightarrow X_-$ be a *K*-equivalence between toric Deligne–Mumford stacks that arises from a toric wall-crossing, as in §1.1. Let \tilde{X} be the common toric blow-up of X_{\pm} and let \overline{X}_0 denote the common blow-down; \overline{X}_0 here is a (singular) toric variety, not a stack.

¹ Recall that Gelfand–Kapranov–Zelevinsky defined the GKZ system in terms of a finite set $A \subset \mathbb{Z}^d$. They called it the A-hypergeometric system.

 $^{^2}$ For an example of the complexities caused by non-simple poles, see the orbifold flop calculation in $[27,\, \$7].$



Consider a direct sum of semiample line bundles $E_0 \to \overline{X}_0$, and pull this back to give vector bundles $E_+ \to X_+$, $\widetilde{E} \to \widetilde{X}$, and $E_- \to X_-$. Let s_+ , \widetilde{s} , and s_- be sections of, respectively, E_+ , \widetilde{E} , and E_- that are compatible via f_+ and f_- (so $f_+^*s_+ = \widetilde{s} = f_-^*s_-$) such that the zero loci of s_{\pm} intersect the flopping locus of φ transversely. Let Y_+ , \widetilde{Y} , and Y_- denote the substacks defined by the zero loci of, respectively, s_+ , \widetilde{s} , and s_- . In this situation there is a commutative diagram:



where the vertical maps are inclusions, the bottom triangle is (1.1), and the squares are Cartesian. The K-equivalence $\varphi \colon X_+ \dashrightarrow X_-$ induces a K-equivalence $\varphi \colon Y_+ \dashrightarrow Y_-$. We now consider the Crepant Transformation Conjecture for $\varphi \colon Y_+ \dashrightarrow Y_-$.

Since the complete intersections Y_{\pm} will not in general be *T*-invariant we consider non-equivariant Gromov–Witten invariants and the non-equivariant quantum product. (Our assumptions on X_{\pm} ensure that the non-equivariant theory makes sense.) Denote by $H^{\bullet}_{amb}(Y_{\pm})$ the image im $\iota^{\star}_{\pm} \subset H^{\bullet}_{CR}(Y_{\pm})$, where $\iota_{\pm} \colon Y_{\pm} \to X_{\pm}$ is the inclusion map. If $\tau \in H^{\bullet}_{amb}(Y_{\pm})$ then the big quantum product \star_{τ} preserves the ambient part $H^{\bullet}_{amb}(Y_{\pm}) \subset$ $H^{\bullet}_{CR}(Y_{\pm})$. We can therefore define a quantum connection on the ambient part:

$$\nabla = d + z^{-1} \sum_{i=0}^{N} (\phi_i \star_{\tau}) d\tau^i$$

This is a pencil of flat connections on the trivial $H^{\bullet}_{amb}(Y_{\pm})$ -bundle over an open set in $H^{\bullet}_{amb}(Y_{\pm})$ where, as in (1.2), $z \in \mathbb{C}^{\times}$ is the pencil variable, $\tau \in H^{\bullet}_{amb}(Y_{\pm})$ is the coordinate on the base of the bundle, ϕ_0, \ldots, ϕ_N are a basis for $H^{\bullet}_{amb}(Y_{\pm})$, and τ^0, \ldots, τ^N are the corresponding co-ordinates of τ .

In §7.1 below we construct an ambient version of the Gamma-integral structure, which is an assignment to each class E in the ambient part of K-theory

$$K^0_{\rm amb}(Y_{\pm}) = \operatorname{im} \iota_{\pm}^{\star} \subset K^0(Y_{\pm})$$

of a flat section $\mathfrak{s}(E)$ for the quantum connection on the ambient part $H^{\bullet}_{\mathrm{amb}}(Y_{\pm})$. This gives a lattice in the space of flat sections which is isomorphic to the ambient part of (integral) K-theory $K^{0}_{\mathrm{amb}}(Y_{\pm})$.

Theorem 1.4. Let $\varphi: Y_+ \dashrightarrow Y_-$ be a K-equivalence between toric complete intersections as above. Then:

(1) the quantum connections on the ambient parts $H^{\bullet}_{amb}(Y_{\pm}) \subset H^{\bullet}_{CR}(Y_{\pm})$ become gaugeequivalent after analytic continuation in τ , via a gauge transformation

$$\Theta_Y(\tau, z) \colon H^{\bullet}_{\mathrm{amb}}(Y_-) \to H^{\bullet}_{\mathrm{amb}}(Y_+)$$

which is homogeneous of degree zero and regular at z = 0. If Y is compact then Θ_Y preserves the orbifold Poincaré pairing;

(2) when expressed in terms of the ambient integral structure, the gauge transformation Θ_Y coincides with the Fourier–Mukai transformation

$$\mathbb{FM}: K^0_{\mathrm{amb}}(Y_-) \to K^0_{\mathrm{amb}}(Y_+) \qquad E \mapsto (F_+)_*(F_-)^*(E)$$

given by the top triangle in (1.4).

As before, Theorem 1.4 is slightly imprecisely stated: precise statements can be found as Theorems 7.2, 7.9, and 7.11 below. Arguing as in §1.1 shows that Theorem 1.4 implies the CIT/Ruan version of the Crepant Transformation Conjecture for $\varphi: Y_+ \dashrightarrow Y_$ whenever it makes sense, with the corresponding map

$$\mathbb{U}_Y \colon \mathcal{H}_{\mathrm{amb}}(Y_-) \to \mathcal{H}_{\mathrm{amb}}(Y_+)$$

between the ambient parts of the Givental spaces for Y_{\pm} being given by:

$$\mathbb{U}_Y = (L_+^{\mathrm{amb}})^{-1} \Theta_Y L_-^{\mathrm{amb}}$$

where L^{amb}_{\pm} are the fundamental solutions for the quantum connections on the ambient parts $H^{\bullet}_{\text{amb}}(Y_{\pm})$.

The proof of Theorem 1.4 relies on the Mirror Theorem for toric complete intersections [30], and on *non-linear Serre duality* [31,41,42,73], which relates the quantum cohomology of Y_{\pm} to the quantum cohomology of the total space of the dual bundles E_{\pm}^{\vee} . Since E_{\pm}^{\vee} is toric, it can be analyzed using Theorem 1.1.

Remark 1.5. The idea of using non-linear Serre duality to analyze wall-crossing has been developed independently by Lee–Priddis–Shoemaker [59], in the context of the Landau–Ginzburg/Calabi–Yau correspondence.

Example 1.6. A mirror Y to the quintic 3-fold arises [7,21,46] as a crepant resolution of an anticanonical hypersurface in $X = \left[\mathbb{P}^4/(\mathbb{Z}/5\mathbb{Z})^3\right]$. A mirror theorem for Y has been proved by Lee–Shoemaker [60]. The variety Y is a Calabi–Yau 3-fold with $h^{1,1}(Y) = 101$. There are many birational models of Y as toric hypersurfaces, corresponding to the many different lattice triangulations of the boundary of the fan polytope for X. Theorem 1.4 implies that the quantum connections (and quantum cohomology algebras) of all of these birational models become isomorphic after analytic continuation over the Kähler moduli space (which is 101-dimensional), and that the isomorphisms involved arise from Fourier–Mukai transformations.

1.4. A note on hypotheses

Since we work with *T*-equivariant Gromov–Witten invariants of the toric Deligne– Mumford stacks X_{\pm} , we do not need to assume that the coarse moduli spaces $|X_{\pm}|$ of X_{\pm} are projective. We insist instead that $|X_{\pm}|$ is semi-projective, i.e. that $|X_{\pm}|$ is projective over the affinization $\operatorname{Spec}(H^0(|X_{\pm}|, \mathcal{O}))$, and also that X_{\pm} contains at least one torus fixed point. These conditions are equivalent to demanding that X_{\pm} is obtained as the GIT quotient $[\mathbb{C}^m/\!/_{\omega}K]$ of a vector space by the linear action of a complex torus K; they ensure that the equivariant quantum cohomology of X_{\pm} admits a non-equivariant limit. In particular, therefore, the non-equivariant version of the Crepant Transformation Conjecture follows automatically from Theorem 1.1.

We do not assume, either, that the stacks X_{\pm} or Y_{\pm} satisfy any sort of positivity or weak Fano condition; put differently, we do not impose any additional convergence hypotheses on the *I*-functions for X_{\pm} and Y_{\pm} . This extra generality is possible because of our hybrid formal/analytic approach, where we single out one variable y_r and analytically continue in that variable alone. The same technique allows us to describe the analytic continuation of *big* quantum cohomology (or its ambient part), as opposed to small quantum cohomology. In general, obtaining convergence results for big quantum cohomology is hard.

1.5. The hemisphere partition function

Recently there was some progress in physics in the exact computation of hemisphere partition functions for gauged linear sigma models. Hori–Romo [48] explained why the Mellin–Barnes analytic continuation of hemisphere partition functions should be compatible with brane transportation [47] in the B-brane category. In the language of this paper, the hemisphere partition function corresponds to a component of the K-theoretic flat section $\mathfrak{s}(E)$, and brane transportation corresponds to the Fourier–Mukai transformation. Theorem 1.1 thus confirms the result of Hori–Romo. Note that the relevant equivalence between B-brane categories should depend on a choice of a path of analytic continuation, and that the Fourier–Mukai transformation in Theorem 1.1 corresponds to a specific choice of path (see Fig. 1).

1.6. Plan of the paper

We fix notation for equivariant Gromov–Witten invariants and equivariant quantum cohomology in §2, and introduce the equivariant Gamma-integral structure in §3. We establish notation for toric Deligne–Mumford stacks in §4. In §5 we study K-equivalences $\varphi: X_+ \dashrightarrow X_-$ of toric Deligne–Mumford stacks arising from wall-crossing, constructing global versions of the equivariant quantum connections for X_{\pm} . We prove the Crepant Transformation Conjecture for toric Deligne–Mumford stacks (Theorem 1.1) in §6, and the Crepant Transformation Conjecture for toric complete intersections (Theorem 1.4) in §7.

1.7. Notation

We use the following notation throughout the paper.

- X denotes a general smooth Deligne–Mumford stack in \$2 and \$3; it denotes a smooth toric Deligne–Mumford stack in \$4 and later.
- $T = (\mathbb{C}^{\times})^m$.
- $R_T = H^{\bullet}_T(\mathrm{pt}, \mathbb{C}).$
- $\lambda_j \in H^2_T(\mathrm{pt}, \mathbb{C}) = \mathrm{Lie}(T)^*$ is the character of $T = (\mathbb{C}^{\times})^m$ given by projection to the *j*th factor, so that $R_T = \mathbb{C}[\lambda_1, \ldots, \lambda_m]$.
- S_T is the localization of R_T with respect to the set of non-zero homogeneous elements.
- $\mathbb{Z}[T] = K_T^{\bullet}(\mathrm{pt})$, so that $\mathbb{Z}[T] = \mathbb{Z}[e^{\pm\lambda_1}, \dots, e^{\pm\lambda_m}].$
- $\mu_l = \{z \in \mathbb{C}^{\times} : z^l = 1\}$ is a cyclic group of order l.

2. Equivariant quantum cohomology

In this section we establish notation for various objects in equivariant Gromov–Witten theory. We introduce equivariant Chen–Ruan cohomology in §2.2, equivariant Gromov–Witten invariants in §2.3, equivariant quantum cohomology in §2.4, Givental's symplectic formalism in §2.5, and the equivariant quantum connection in §2.6.

2.1. Smooth Deligne–Mumford stacks with torus action

Let X be a smooth Deligne–Mumford stack of finite type over \mathbb{C} equipped with an action of an algebraic torus $T \cong (\mathbb{C}^{\times})^m$. Let |X| denote the coarse moduli space of X and let IX denote the inertia stack $X \times_{|X|} X$ of X: a point on IX is given by a pair (x, g) with $x \in X$ and $g \in \operatorname{Aut}(x)$. We write

$$IX = \bigsqcup_{v \in \mathsf{B}} X_v$$

for the decomposition of IX into connected components. We assume the following conditions:

- (1) the coarse moduli space |X| is semi-projective, i.e. is projective over the affinization Spec $H^0(|X|, \mathcal{O}) = \text{Spec } H^0(X, \mathcal{O});$
- (2) all the *T*-weights appearing in the *T*-representation $H^0(X, \mathcal{O})$ are contained in a strictly convex cone in $\text{Lie}(T)^*$, and the *T*-invariant subspace $H^0(X, \mathcal{O})^T$ is \mathbb{C} ;
- (3) the inertia stack IX is equivariantly formal, that is, the *T*-equivariant cohomology $H_T^{\bullet}(IX;\mathbb{C})$ is a free module over $R_T := H_T^{\bullet}(\mathrm{pt};\mathbb{C})$ and one has a (non-canonical) isomorphism of R_T -modules $H_T^{\bullet}(IX;\mathbb{C}) \cong H^{\bullet}(IX;\mathbb{C}) \otimes_{\mathbb{C}} R_T$.

These conditions allow us to define Gromov–Witten invariants of X and also the equivariant (Dolbeault) index of coherent sheaves on X. The first and second conditions together imply that the fixed set X^T is compact. The third condition seems to be closely related to the first two, but it implies for example the localization of equivariant cohomology: the restriction $H_T^{\bullet}(IX; \mathbb{C}) \to H_T^{\bullet}(IX^T; \mathbb{C})$ to the T-fixed locus is injective and becomes an isomorphism after localization (see [45]). Later we shall restrict to the case where X is a toric Deligne–Mumford stack, where conditions (1)–(3) automatically hold, but the definitions in this section make sense for general X satisfying these conditions.

2.2. Equivariant Chen-Ruan cohomology

Let $H^{\bullet}_{\mathrm{CR},T}(X)$ denote the even part of the *T*-equivariant orbifold cohomology group of Chen and Ruan. It is defined as the even degree part of the *T*-equivariant cohomology

$$H^{k}_{\mathrm{CR},T}(X) = \bigoplus_{v \in \mathsf{B}: k-2\iota_{v} \in 2\mathbb{Z}} H^{k-2\iota_{v}}_{T}(X_{v};\mathbb{C})$$

of the inertia stack IX. The grading of $H^{\bullet}_{CR,T}(X)$ is shifted from that of $H^{\bullet}_{T}(IX)$ by the so-called *age* or *degree shifting number* $\iota_v \in \mathbb{Q}$ [24]; note that we consider only the even degree classes in $H^{\bullet}_{T}(IX)$. (For toric stacks, all cohomology classes on IX are of even degree.) Equivariant formality of IX gives that $H^{\bullet}_{CR,T}(X)$ is a free module over R_T . We write

$$(\alpha, \beta) = \int_{IX} \alpha \cup \operatorname{inv}^* \beta, \qquad \alpha, \beta \in H^{\bullet}_{\operatorname{CR}, T}(X)$$

for the equivariant orbifold Poincaré pairing: here inv: $IX \to IX$ denotes the involution on the inertia stack IX that sends a point (x,g) with $x \in X, g \in \operatorname{Aut}(x)$ to (x,g^{-1}) . Since X is not necessarily proper, the equivariant integral on the right-hand side here is defined via the Atiyah–Bott localization formula [3] and takes values in the localization S_T of R_T with respect to the multiplicative set of non-zero homogeneous elements³ in R_T .

³ Note that $R_T \subsetneq S_T \subsetneq \operatorname{Frac}(R_T)$; we use S_T instead of $\operatorname{Frac}(R_T)$ since we need a grading on S_T later.

2.3. Equivariant Gromov–Witten invariants

Let $X_{g,n,d}$ denote the moduli space of degree-*d* stable maps to *X* from genus *g* orbifold curves with *n* marked points [1,2]; here $d \in H_2(|X|;\mathbb{Z})$. The moduli space carries a *T*-action and a virtual fundamental cycle $[X_{g,n,d}]^{\text{vir}} \in A_{\bullet,T}(X_{g,n,d};\mathbb{Q})$. There are *T*-equivariant evaluation maps $\text{ev}_i \colon X_{g,n,d} \to \overline{IX}, 1 \leq i \leq n$, to the rigidified inertia stack \overline{IX} (see [2]). Let $\psi_i \in H_T^2(X_{g,n,d})$ denote the psi-class at the *i*th marked point, i.e. the equivariant first Chern class of the *i*th universal cotangent line bundle $L_i \to X_{g,n,d}$. For $\alpha_1, \ldots, \alpha_n \in H_{\text{CR},T}^{\bullet}(X)$ and non-negative integers k_1, \ldots, k_n , the *T*-equivariant Gromov-Witten invariant is defined to be:

$$\left\langle \alpha_1 \psi^{k_1}, \dots, \alpha_n \psi^{k_n} \right\rangle_{g,n,d}^X = \int\limits_{[X_{g,n,d}]^{\text{vir}}} \prod_{i=1}^n (\text{ev}_i^* \alpha_i) \psi_i^{k_i}$$
(2.1)

where we regard α_i as a class in $H^{\bullet}_T(\overline{IX})$ via the canonical isomorphism $H^{\bullet}_T(\overline{IX}) \cong H^{\bullet}_T(IX)$. The moduli space here is not necessarily proper: the right-hand side is again defined via the Atiyah–Bott localization formula and so belongs to S_T . Conditions (1) and (2) in §2.1 ensure that the *T*-fixed locus $X^T_{g,n,d}$ in the moduli space is compact, and thus that the right-hand side of (2.1) is well-defined.

2.4. Equivariant quantum cohomology

Consider the cone NE(X) $\subset H_2(|X|, \mathbb{R})$ generated by classes of effective curves and set NE(X)_Z := { $d \in H_2(|X|, \mathbb{Z}) : d \in NE(X)$ }. For a ring R, define R[[Q]] to be the ring of formal power series with coefficients in R:

$$R\llbracket Q \rrbracket = \left\{ \sum_{d \in \operatorname{NE}(X)_{\mathbb{Z}}} a_d Q^d : a_d \in R \right\}$$

so that Q is a so-called Novikov variable [62, III 5.2.1]. Let $\phi_0, \phi_1, \ldots, \phi_N$ be a homogeneous basis for $H^{\bullet}_{\operatorname{CR},T}(X)$ over R_T and let $\tau^0, \tau^1, \ldots, \tau^N$ be the corresponding linear co-ordinates. We assume that $\phi_0 = 1$ and $\phi_1, \ldots, \phi_r \in H^2_T(X)$ are degree-two untwisted classes that induce a \mathbb{C} -basis of $H^2(X; \mathbb{C}) \cong H^2_T(X)/H^2_T(\operatorname{pt})$. We write $\tau = \sum_{i=0}^N \tau^i \phi_i$ for a general element of $H^{\bullet}_{\operatorname{CR},T}(X)$. The equivariant quantum product \star_{τ} at $\tau \in H^{\bullet}_{\operatorname{CR},T}(X)$ is defined by the formula

$$(\phi_i \star_{\tau} \phi_j, \phi_k) = \sum_{d \in \operatorname{NE}(X)_{\mathbb{Z}}} \sum_{n=0}^{\infty} \frac{Q^d}{n!} \langle \phi_i, \phi_j, \phi_k, \tau, \dots, \tau \rangle_{0, n+3, d}^X$$

or, equivalently, by

T. Coates et al. / Advances in Mathematics 329 (2018) 1002-1087

$$\phi_i \star_{\tau} \phi_j = \sum_{d \in \text{NE}(X)_{\mathbb{Z}}} \sum_{n=0}^{\infty} \frac{Q^d}{n!} \operatorname{inv}^* \operatorname{ev}_{3,*} \left(\operatorname{ev}_1^*(\phi_i) \operatorname{ev}_2^*(\phi_j) \prod_{l=4}^{n+3} \operatorname{ev}_l^*(\tau) \cap [X_{0,n+3,d}]^{\text{vir}} \right).$$
(2.2)

Conditions (1) and (2) in §2.1 ensure that $ev_3 : X_{0,n+3,d} \to \overline{IX}$ is proper, and thus that the push-forward along ev_3 is well-defined without inverting equivariant parameters. It follows that:

$$\phi_i \star_\tau \phi_j \in H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_T} R_T \llbracket \tau, Q \rrbracket$$

where $R_T[\![\tau, Q]\!] = R_T[\![\tau^0, \ldots, \tau^N]\!][\![Q]\!]$. The product \star_{τ} defines an associative and commutative ring structure on $H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_T} R_T[\![\tau, Q]\!]$. The non-equivariant limit of \star_{τ} exists, and this limit defines the non-equivariant quantum cohomology $(H^{\bullet}_{\mathrm{CR}}(X) \otimes_{\mathbb{C}} \mathbb{C}[\![\tau, Q]\!], \star_{\tau})$.

Remark 2.1. The divisor equation [2, Theorem 8.3.1] implies that exponentiated H^2 -variables and the Novikov variable Q play the same role: one has

$$(\phi_i \star_\tau \phi_j, \phi_k) = \sum_{d \in \operatorname{NE}(X)_{\mathbb{Z}}} \sum_{n=0}^{\infty} \frac{Q^d e^{\langle \sigma, d \rangle}}{n!} \langle \phi_i, \phi_j, \phi_k, \tau', \dots, \tau' \rangle_{0, n+3, d}^X$$

where $\tau = \sigma + \tau'$ with $\sigma = \sum_{i=1}^{r} \tau^{i} \phi_{i}$ and $\tau' = \tau_{0} \phi_{0} + \sum_{i=r+1}^{N} \tau^{i} \phi_{i}$. The String Equation [2, Theorem 8.3.1] implies that the right-hand side here is in fact independent of τ_{0} .

2.5. Givental's Lagrangian cone

Let $S_T((z^{-1}))$ denote the ring of formal Laurent series in z^{-1} with coefficients in S_T . Givental's symplectic vector space is the space

$$\mathcal{H} = H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_T} S_T((z^{-1})) \llbracket Q \rrbracket$$

equipped with the non-degenerate $S_T[Q]$ -bilinear alternating form:

$$\Omega(f,g) = -\operatorname{Res}_{z=\infty}(f(-z),g(z))dz$$

with $f, g \in \mathcal{H}$. The space is equipped with a standard polarization

$$\mathcal{H} = \mathcal{H}_+ \oplus \mathcal{H}_-$$

where

$$\mathcal{H}_{+} := H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_{T}} S_{T}[z]\llbracket Q \rrbracket \qquad \text{and} \qquad \mathcal{H}_{-} := z^{-1} H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_{T}} S_{T}\llbracket z^{-1} \rrbracket \llbracket Q \rrbracket$$

are isotropic subspaces for Ω . The standard polarization identifies \mathcal{H} with the cotangent bundle of \mathcal{H}_+ . The genus-zero descendant Gromov-Witten potential is a formal function $\mathcal{F}_X^0: (\mathcal{H}_+, -z1) \to S_T[\![Q]\!]$ defined on the formal neighbourhood of $-z \cdot 1$ in \mathcal{H}_+ and taking values in $S_T[\![Q]\!]$:

$$\mathcal{F}_X^0(-z\mathbf{1} + \mathbf{t}(z)) = \sum_{d \in \operatorname{NE}(X)_{\mathbb{Z}}} \sum_{n=0}^{\infty} \frac{Q^d}{n!} \langle \mathbf{t}(\psi), \dots, \mathbf{t}(\psi) \rangle_{0,n,d}^X$$

Here $\mathbf{t}(z) = \sum_{n=0}^{\infty} t_n z^n$ with $t_n \in H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_T} S_T[\![Q]\!]$. Let $\{\phi^i\} \subset H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_T} S_T$ denote the basis Poincaré dual to $\{\phi_i\}$, so that $(\phi_i, \phi^j) = \delta^j_i$.

Definition 2.2 ([29,43]). Givental's Lagrangian cone $\mathcal{L}_X \subset (\mathcal{H}, -z1)$ is the graph of the differential $d\mathcal{F}_X^0: \mathcal{H}_+ \to T^*\mathcal{H}_+ \cong \mathcal{H}$. It consists of points of \mathcal{H} of the form:

$$-z1 + \mathbf{t}(z) + \sum_{d \in \operatorname{NE}(X)_{\mathbb{Z}}} \sum_{n=0}^{\infty} \sum_{i=0}^{N} \frac{Q^{d}}{n!} \left\langle \frac{\phi_{i}}{-z - \psi}, \mathbf{t}(\psi), \dots, \mathbf{t}(\psi) \right\rangle_{0, n+1, d} \phi^{i}$$
(2.3)

where $1/(-z - \psi)$ in the correlator should be expanded as the power series $\sum_{k=0}^{\infty} \psi^k (-z)^{-k-1}$ in z^{-1} . In a more formal language, we define the notion of a 'point on \mathcal{L}_X ' as follows. Let $x = (x_1, \ldots, x_n)$ be formal parameters. An $S_T[[Q, x]]$ -valued point on \mathcal{L}_X is an element of $\mathcal{H}[[x]]$ of the form (2.3) with $\mathbf{t}(z) \in \mathcal{H}_+[[x]]$ satisfying

$$\mathbf{t}(z)|_{Q=x=0} = 0.$$

It should be thought of as a formal family of elements on \mathcal{L}_X parametrized by x.

The submanifold \mathcal{L}_X encodes all genus-zero Gromov–Witten invariants (2.1). It has the following special geometric properties [43]: *it is a cone, and a tangent space* T of \mathcal{L}_X *is tangent to* \mathcal{L}_X *exactly along* zT. Knowing Givental's Lagrangian cone \mathcal{L}_X is equivalent to knowing the data of the quantum product \star_{τ} , i.e. \mathcal{L}_X can be reconstructed from \star_{τ} and vice versa. See Remark 2.5.

2.6. The equivariant quantum connection and its fundamental solution

Let $v \in H^{\bullet}_{\mathrm{CR},T}(X)$. The equivariant quantum connection

$$\nabla_{v} \colon H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_{T}} R_{T}[z]\llbracket \tau, Q \rrbracket \to z^{-1} H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_{T}} R_{T}[z]\llbracket \tau, Q \rrbracket$$

is defined by

$$\nabla_v f(\tau) = \partial_v f(\tau) + z^{-1} v \star_\tau f(\tau)$$

where $\partial_v f(\tau) = \frac{d}{ds} f(\tau + sv)|_{s=0}$ is the directional derivative. We write ∇_i for ∇_{ϕ_i} and ∇f for $\sum_{i=0}^{N} (\nabla_i f) d\tau^i$. The associativity of \star_{τ} implies that the connection ∇ is flat, that is, $[\nabla_i, \nabla_j] = 0$ for all i, j. Let ρ denote the equivariant first Chern class (in the untwisted sector):

$$\rho := c_1^T(TX) \in H^2_T(X) \subset H^2_{\operatorname{CR},T}(X)$$

For homogeneous $\phi \in H^{\bullet}_{\mathrm{CR},T}(X)$, we write deg ϕ for the age-shifted (real) degree of ϕ , so that $\phi \in H^{\mathrm{deg}\,\phi}_{\mathrm{CR},T}(X)$. The equivariant Euler vector field \mathcal{E} and the grading operator $\mu \in \mathrm{End}_{\mathbb{C}}(H^{\bullet}_{\mathrm{CR},T}(X))$ are defined by

$$\mathcal{E} := \sum_{i=1}^{m} \lambda_i \frac{\partial}{\partial \lambda_i} + \sum_{i=0}^{N} \left(1 - \frac{\deg \phi_i}{2} \right) \tau^i \frac{\partial}{\partial \tau^i} + \partial_\rho$$

$$\mu(\phi) := \left(\frac{\deg \phi}{2} - \frac{\dim_{\mathbb{C}} X}{2} \right) \phi$$
(2.4)

where $\lambda_1, \ldots, \lambda_m \in H^2_T(\text{pt})$ are generators of R_T (see §1.7). The grading operator on $H^{\bullet}_{\text{CR},T}(X) \otimes_{R_T} R_T[z][\tau, Q]$ is defined by

$$\operatorname{Gr}(f(\tau, z)\phi) = \left(\left(z\frac{\partial}{\partial z} + \mathcal{E}\right)f(\tau, z)\right)\phi + f(\lambda, \tau, z)\mu(\phi)$$

where $\phi \in H^{\bullet}_{\mathrm{CR},T}(X)$ and $f(\lambda, \tau, z) \in R_T[z][[\tau, Q]]$. The quantum connection is compatible with the grading operator in the sense that $[\mathrm{Gr}, \nabla_i] = \nabla_{[\mathcal{E}, \partial_{\tau^i}]} = (\frac{1}{2} \deg \phi_i - 1) \nabla_i,$ $i = 0, \ldots, N$. This follows from the virtual dimension formula for the moduli space of stable maps.

Notation 2.3. Let $v \in H^2_T(X)$ be a degree-two class in the untwisted sector. The action of v on $H^{\bullet}_{CR,T}(X)$ is defined by $v \cdot \alpha = q^*(v) \cup \alpha$, where $q: IX \to X$ is the natural projection. (This coincides with the action of v via the Chen–Ruan cup product.)

Consider the flat section equations for ∇ , and a fundamental solution

$$L(\tau, z) \in \operatorname{End}_{R_T}(H^{\bullet}_{\operatorname{CR}, T}(X)) \otimes_{R_T} R_T((z^{-1})) \llbracket \tau, Q \rrbracket$$

determined by the following conditions:

$$\nabla_i L(\tau, z)\phi = 0 \qquad \text{for } i = 0, \dots, N \qquad \text{(flatness)} \qquad (2.5)$$
$$\left(vQ\frac{\partial}{\partial Q} - \partial_v\right)L(\tau, z)\phi = L(\tau, z)\frac{v}{z}\phi \qquad \text{for } v \in H^2_T(X) \qquad \text{(divisor equation)}$$

(2.6)

T. Coates et al. / Advances in Mathematics 329 (2018) 1002-1087

$$L(\tau, z)|_{\tau=Q=0} = \mathrm{id}$$
 (initial condition)
(2.7)

Here $\phi \in H^{\bullet}_{\operatorname{CR},T}(X)$ and $vQ\frac{\partial}{\partial Q}$ with $v \in H^2_T(X)$ acts on Novikov variables as $Q^d \mapsto \langle v, d \rangle Q^d$ (it acts by zero when $v \in H^2_T(\operatorname{pt}) \subset H^2_T(X)$). The flatness equation fixes $L(\tau, z)$ up to right multiplication by an endomorphism-valued function g(z; Q) in z and Q; the divisor equation implies that the ambiguity g(z; Q) is independent of Q and commutes with $v \cup$, $v \in H^2_T(X)$; finally the initial condition fixes $L(\tau, z)$ uniquely. The fundamental solution satisfying these conditions can be written explicitly in terms of (descendant) Gromov–Witten invariants:

$$L(\tau, z)\phi_{i} = \phi_{i} + \sum_{j=0}^{N} \sum_{d \in NE(X)_{\mathbb{Z}}} \sum_{\substack{n=0\\(n \ge 1 \text{ if } d=0)}}^{\infty} \frac{Q^{d}}{n!} \left\langle \frac{\phi_{i}}{-z - \psi}, \tau, \dots, \tau, \phi_{j} \right\rangle_{0, n+2, d}^{X} \phi^{j}$$
(2.8)

This is defined over R_T (without inverting equivariant parameters) because it can be rewritten in terms of the push-forward along the last evaluation map ev_{n+2} as in (2.2). A straightforward equivariant generalization of [41, Corollary 6.3], [66, Proposition 2], [49, Proposition 2.4] gives:

Proposition 2.4. The fundamental solution $L(\tau, z)$ in (2.8) satisfies the conditions (2.5)–(2.7). Furthermore it satisfies:

$$L(\tau, z) = \mathrm{id} + O(z^{-1}) \qquad (\text{regularity at } z = \infty)$$

Gr $L(\tau, z)\phi = L(\tau, z)\left(\mu - \frac{\rho}{z}\right)\phi \qquad (\text{homogeneity})$
 $(\alpha, \beta) = (L(\tau, -z)\alpha, L(\tau, z)\beta) \qquad (\text{unitarity})$

where $\phi, \alpha, \beta \in H^{\bullet}_{\mathrm{CR},T}(X)$.

Remark 2.5 ([43]). The fundamental solution $L(\tau, z)$ is determined by the quantum product \star_{τ} via differential equations (2.5)–(2.7). Then $\tau \mapsto T_{\tau} = L(\tau, -z)^{-1}\mathcal{H}_{+}$ gives a versal family of tangent spaces to Givental's cone \mathcal{L}_X . The cone \mathcal{L}_X is reconstructed as $\mathcal{L}_X = \bigcup_{\tau} zT_{\tau}$.

We now study ∇ -flat sections $s(\tau, z)$ that are homogeneous of degree zero: Gr $(s(\tau, z)) = 0$. By Proposition 2.4, if a flat section $L(\tau, z)f(z)$ is homogeneous of degree zero, then:

$$\left(z\frac{\partial}{\partial z} + \mu - \frac{\rho}{z}\right)f(z) = 0$$

This differential equation has the fundamental solution:

$$z^{-\mu}z^{\rho} = z^{\rho/z}z^{-\mu} = \exp(\rho\log(z)/z)z^{-\mu}$$

that belongs to $\operatorname{End}_{R_T}(H^{\bullet}_{\operatorname{CR},T}(X)) \otimes_{R_T} R_T[\log z]((z^{-1/k}))$ for some $k \in \mathbb{N}$; here k is chosen so that all the eigenvalues of $k\mu$ are integers. Note that homogeneous flat sections can be multi-valued in z (as they contain $\log z$). We have:

Corollary 2.6. The sections $s_i(\tau, z) = L(\tau, z)z^{-\mu}z^{\rho}\phi_i$, i = 0, ..., N satisfy $\nabla s_i(\tau, z) =$ Gr $s_i(\tau, z) = 0$ and give a basis of homogeneous flat sections. They belong to $H^{\bullet}_{CR,T}(X) \otimes_{R_T} R_T[\log z]((z^{-1/k}))[\tau, Q]$ for a sufficiently large $k \in \mathbb{N}$.

3. Equivariant Gamma-integral structure

In this section we introduce one of the main ingredients of our result: an integral structure for equivariant quantum cohomology. This is a $K_T^0(\text{pt})$ -lattice in the space of flat sections for the equivariant quantum connection on X which is isomorphic to the integral equivariant K-group $K_T^0(X)$: it generalizes the integral structure for non-equivariant quantum cohomology constructed by Iritani [49] and Katzarkov–Kontsevich–Pantev [54]. Similar structures have been studied by Okounkov–Pandharipande [65] in the case where X is a Hilbert scheme of points in \mathbb{C}^2 , and by Brini–Cavalieri–Ross [16] in the case where X is a 3-dimensional toric Calabi–Yau stack. We define the integral structure in §3.1. In §3.2 we observe that the quantum product, flat sections for the quantum connection, and integral structure continue to make sense when the Novikov variable Q (see §2.4) is specialized to Q = 1.

The integral structure is defined in terms of a *T*-equivariant characteristic class of *X* called the $\hat{\Gamma}$ -class. One of the key points in this section is that the $\hat{\Gamma}$ -class behaves like a square root of the Todd class: see equation (3.4). When combined with the Hirzebruch–Riemann–Roch formula, this leads to one of the fundamental properties of the integral structure: that the so-called framing map is pairing-preserving (Proposition 3.2 below).

3.1. The equivariant Gamma class and the equivariant Gamma-integral structure

Let $K_T^0(X)$ denote the Grothendieck group of *T*-equivariant vector bundles on *X*. We write $H_T^{\bullet\bullet}(IX) := \prod_p H_T^{2p}(IX)$. We introduce an orbifold Chern character map $\tilde{ch}: K_T^0(X) \to H_T^{\bullet\bullet}(IX)$ as follows. Let $IX = \bigsqcup_{v \in \mathbf{B}} X_v$ be the decomposition of the inertia stack *IX* into connected components, let $q_v: X_v \to X$ be the natural map, and let *E* be a *T*-equivariant vector bundle on *X*. The stabilizer g_v along X_v acts on the vector bundle $q_v^*E \to X_v$, giving an eigenbundle decomposition

$$q_v^* E = \bigoplus_{0 \le f < 1} E_{v,f} \tag{3.1}$$

where g_v acts on $E_{v,f}$ by $\exp(2\pi i f)$. The equivariant Chern character is defined to be

T. Coates et al. / Advances in Mathematics 329 (2018) 1002-1087

$$\widetilde{\mathrm{ch}}(E) = \bigoplus_{v \in \mathsf{B}} \sum_{0 \le f < 1} e^{2\pi \mathrm{i}f} \mathrm{ch}^T(E_{v,f})$$

where $\operatorname{ch}^{T}(E_{v,f}) \in H_{T}^{\bullet\bullet}(X_{v})$ is the *T*-equivariant Chern character. Let $\delta_{v,f,i}$, $1 \leq i \leq \operatorname{rank}(E_{v,f})$ be the *T*-equivariant Chern roots of $E_{v,f}$, so that $c^{T}(E_{v,f}) = \prod_{i}(1 + \delta_{v,f,i})$. These Chern roots are not actual cohomology classes, but symmetric polynomials in the Chern roots make sense as equivariant cohomology classes on X_{v} . The *T*-equivariant orbifold Todd class $\widetilde{\operatorname{Td}}(E) \in H_{T}^{\bullet\bullet}(IX)$ is defined to be:

$$\widetilde{\mathrm{Td}}(E) = \bigoplus_{v \in \mathsf{B}} \left(\prod_{0 < f < 1} \prod_{i=1}^{\mathrm{rank}(E_{v,f})} \frac{1}{1 - e^{-2\pi \mathrm{i}f} e^{-\delta_{v,f,i}}} \right) \prod_{i=1}^{\mathrm{rank}(E_{v,0})} \frac{\delta_{v,0,i}}{1 - e^{-\delta_{v,0,i}}}$$

We write $\widetilde{\mathrm{Td}}_X = \widetilde{\mathrm{Td}}(TX)$ for the orbifold Todd class of the tangent bundle.

Recall that, because we are assuming condition (2) from §2.1, all of the *T*-weights of $H^0(X, \mathcal{O})$ lie in a strictly convex cone in $\text{Lie}(T)^*$. After changing the identification of T with $(\mathbb{C}^{\times})^m$ if necessary, we may assume that this cone is contained within the cone spanned by the standard characters $\lambda_1, \ldots, \lambda_m$ of $H^2_T(\text{pt}) = \text{Lie}(T)^*$ defined in §1.7. As is explained in [32], under conditions (1)–(2) in §2.1 there is a well-defined equivariant Euler characteristic

$$\chi(E) := \sum_{i=0}^{\dim X} (-1)^i \operatorname{ch}^T \left(H^i(X, E) \right)$$
(3.2)

taking values in

$$\mathbb{Z}\llbracket e^{\lambda} \rrbracket [e^{-\lambda}]_{\mathrm{rat}} := \left\{ f \in \mathbb{Z}\llbracket e^{\lambda_1}, \dots, e^{\lambda_m} \rrbracket [e^{-\lambda_1}, \dots, e^{-\lambda_m}] : \begin{array}{c} f \text{ is the Laurent expansion} \\ \text{of a rational function in} \\ e^{\lambda_1}, \dots, e^{\lambda_m} \text{ at} \\ e^{\lambda_1} = \dots = e^{\lambda_m} = 0 \end{array} \right\}$$

and we expect that the following equivariant Hirzebruch–Riemann–Roch (HRR) formula should hold:

$$\chi(E) = \int_{IX} \widetilde{\operatorname{ch}}(E) \cup \widetilde{\operatorname{Td}}_X$$
(3.3)

(This holds for toric Deligne–Mumford stacks [32].) Formula (3.3) should be interpreted with care. The right-hand side is defined via the localization formula, and lies in a completion \hat{S}_T of S_T :

$$\widehat{S}_T := \left\{ \sum_{n \in \mathbb{Z}} a_n : a_n \in S_T, \deg a_n = n, \text{ there exists } n_0 \in \mathbb{Z} \\ \text{such that } a_n = 0 \text{ for all } n < n_0 \right\}$$

There is an inclusion of rings $\mathbb{Z}[\![e^{\lambda}]\!][e^{-\lambda}]_{\mathrm{rat}} \hookrightarrow \widehat{S}_T$ given by Laurent expansion at $\lambda_1 = \cdots = \lambda_m = 0$ (see [32]), and (3.3) asserts that $\chi(E)$ coincides with the right-hand side after this inclusion.

We now introduce a lattice in the space of homogeneous flat sections for the quantum connection which is identified with the equivariant K-group of X. The key ingredient in the definition is the characteristic class, called the *Gamma class*, defined as follows. Let E be a vector bundle on X and consider the bundles $E_{v,f} \to X_v$ and their equivariant Chern roots $\delta_{v,f,i}$, $i = 1, \ldots, \operatorname{rank}(E_{v,f})$ as above (see (3.1)). The equivariant Gamma class $\widehat{\Gamma}(E) \in H_T^{\bullet}(IX)$ is defined to be:

$$\widehat{\Gamma}(E) = \bigoplus_{v \in \mathbf{B}} \prod_{0 \le f < 1} \prod_{i=1}^{\operatorname{rank}(E_{v,f})} \Gamma(1 - f + \delta_{v,f,i})$$

Here the Γ -function on the right-hand side should be expanded as a Taylor series at 1-f, and then evaluated at $\delta_{v,f,i}$. The identity $\Gamma(1-z)\Gamma(1+z) = 2\pi i z e^{-\pi i z}/(1-e^{-2\pi i z})$ implies that

$$\begin{split} \left[\widehat{\Gamma}(E^*) \cup \widehat{\Gamma}(E)\right]_v &= \prod_{i,f} \Gamma(1 - \overline{f} - \delta_{v,f,i}) \Gamma(1 - f + \delta_{v,f,i}) \\ &= (2\pi i)^{\operatorname{rank}((q_v^* E)^{\operatorname{mov}})} \left[e^{-\pi i (\operatorname{age}(q^* E) + c_1(q^* E))} (2\pi i)^{\frac{\operatorname{deg}_0}{2}} \widetilde{\operatorname{Td}}(E) \right]_{\operatorname{inv}(v)} \end{split}$$

$$(3.4)$$

where \cup is the cup product on IX, $[\cdots]_v$ denotes the component in $H^{\bullet}_T(X_v)$, $0 \leq \overline{f} < 1$ is the fractional part of -f, $(q_v^*E)^{\mathrm{mov}} = \bigoplus_{f \neq 0} E_{v,f}$ is the moving part of q_v^*E , $q: IX \to X$ is the natural projection, $\operatorname{age}(q^*E): IX \to \mathbb{Q}$ is the locally constant function given by $\operatorname{age}(q^*E)|_{X_v} = \sum_f f \operatorname{rank}(E_{v,f})$, $\operatorname{deg}_0: H^{\bullet\bullet}_T(IX) \to H^{\bullet\bullet}_T(IX)$ is the degree operator defined by $\operatorname{deg}_0(\phi) = 2p\phi$ for $\phi \in H^{2p}_T(IX)$, and $\operatorname{inv}(v) \in \mathsf{B}$ corresponds to the component $X_{\operatorname{inv}(v)}$ of IX defined by $\operatorname{inv}(X_v) = X_{\operatorname{inv}(v)}$. Note that deg_0 means the degree as a class on IX, not the age-shifted degree as an element of $H^{\bullet\bullet}_{\operatorname{CR},T}(X)$.

Definition 3.1. Define the *K*-group framing

$$\mathfrak{s} \colon K_T^0(X) \to H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_T} R_T[\log z]((z^{-1/k}))\llbracket Q, \tau \rrbracket$$

by the formula:

$$\mathfrak{s}(E)(\tau,z) = \frac{1}{(2\pi)^{\dim X/2}} L(\tau,z) z^{-\mu} z^{\rho} \left(\widehat{\Gamma}_X \cup (2\pi \mathfrak{i})^{\frac{\deg_0}{2}} \operatorname{inv}^* \widetilde{\operatorname{ch}}(E)\right)$$

where $k \in \mathbb{N}$ is as in Corollary 2.6 and $\widehat{\Gamma}_X \cup$ is the cup product in $H_T^{\bullet\bullet}(IX)$. Corollary 2.6 shows that the image of \mathfrak{s} is contained in the space of Gr-degree zero flat sections. Note that $z^{-\mu}$ maps $H_{CR,T}^{\bullet\bullet}(X)$ into $H_{CR,T}^{\bullet}(X) \otimes_{R_T} R_T((z^{-1/k}))$.

For *T*-equivariant vector bundles E, F on X, let $\chi(E, F) \in \mathbb{Z}\llbracket e^{\lambda} \rrbracket [e^{-\lambda}]_{rat}$ denote the equivariant Euler pairing defined by:

$$\chi(E,F) := \sum_{i=0}^{\dim X} (-1)^i \operatorname{ch}^T \left(\operatorname{Ext}^i(E,F) \right)$$
(3.5)

We use a z-modified version $\chi_z(E, F)$ that is given by replacing equivariant parameters λ_j in $\chi(E, F)$ with $2\pi i \lambda_j/z$:

$$\chi_z(E,F) := (2\pi i z^{-1})^{\sum_{i=1}^m \lambda_i \partial_{\lambda_i}} \chi(E,F) \in \mathbb{Z}\llbracket e^{2\pi i \lambda/z} \rrbracket [e^{-2\pi i \lambda/z}]_{\text{rat}}$$
(3.6)

Proposition 3.2 (cf. [49, Proposition 2.10]). Suppose that the equivariant HRR formula (3.3) holds. For $E, F \in K_T^0(X)$, we have

$$(\mathfrak{s}(E)(\tau, e^{-\pi \mathfrak{i}}z), \mathfrak{s}(F)(\tau, z)) = \chi_z(E, F).$$

Proof. Set $\Psi(E) = \widehat{\Gamma}_X \cup (2\pi i)^{\frac{\deg_0}{2}}$ inv^{*} $\widetilde{ch}(E)$. Using the unitarity in Proposition 2.4, we have

$$\left(\mathfrak{s}(E)(\tau, e^{-\pi \mathbf{i}}z), \mathfrak{s}(F)(\tau, z)\right) = \frac{1}{(2\pi)^{\dim X}} \left(z^{-\mu} e^{\pi \mathbf{i}\mu} z^{\rho} e^{-\pi \mathbf{i}\rho} \Psi(E), z^{-\mu} z^{\rho} \Psi(F) \right).$$
(3.7)

Write $\lambda \partial_{\lambda} = \sum_{i=1}^{m} \lambda_i \partial_{\lambda_i}$. Using $(z^{-\mu} \alpha, z^{-\mu} \beta) = z^{-\lambda \partial_{\lambda}}(\alpha, \beta), e^{\pi i \mu} \rho = -\rho e^{\pi i \mu}, (z^{-\rho} \alpha, z^{\rho} \beta) = (\alpha, \beta)$, we have

$$(3.7) = \frac{z^{-\lambda\partial_{\lambda}}}{(2\pi)^{\dim X}} \left(e^{\pi \mathbf{i}\rho} e^{\pi \mathbf{i}\mu} \Psi(E), \Psi(F) \right)$$

$$= \frac{z^{-\lambda\partial_{\lambda}}}{(2\pi)^{\dim X}} \int_{IX} \left(e^{\pi \mathbf{i}q^*\rho} e^{\pi \mathbf{i}\mu} \widehat{\Gamma}_X(2\pi \mathbf{i})^{\frac{\deg_0}{2}} \operatorname{inv}^* \widetilde{\operatorname{ch}}(E) \right)$$

$$\cup \operatorname{inv}^* \left(\widehat{\Gamma}_X(2\pi \mathbf{i})^{\frac{\deg_0}{2}} \operatorname{inv}^* \widetilde{\operatorname{ch}}(F) \right)$$

$$= \frac{z^{-\lambda\partial_{\lambda}}}{(2\pi)^{\dim X}} \sum_{v \in \mathsf{B}_{X_v}} \int_{v \in \mathsf{R}_{X_v}} e^{\pi \mathbf{i}(\iota_v - \frac{\dim X}{2})} \left[\widehat{\Gamma}_X^* \widehat{\Gamma}_X \right]_{\operatorname{inv}(v)} (2\pi \mathbf{i})^{\frac{\deg_0}{2}} \left[\widetilde{\operatorname{ch}}(E^*) \widetilde{\operatorname{ch}}(F) \right]_v$$

$$= z^{-\lambda\partial_{\lambda}} \sum_{v \in \mathsf{B}} \frac{1}{(2\pi \mathbf{i})^{\dim X_v}} \int_{X_v} (2\pi \mathbf{i})^{\frac{\deg_0}{2}} \left[\widetilde{\operatorname{ch}}(E^* \otimes F) \cup \widetilde{\operatorname{Td}}_X \right]_v$$

where we set $\widehat{\Gamma}_X^* = \widehat{\Gamma}(T^*X)$ and used equation (3.4) in the last line. The last expression equals $\chi_z(E, F)$ by the HRR formula (3.3). \Box

Remark 3.3. Okounkov–Pandharipande [65] and Braverman–Maulik–Okounkov [15] introduced shift operators S_i on quantum cohomology, which induce the shift $\lambda_i \to \lambda_i + z$ of

equivariant parameters (see [63, Chapter 8] for a detailed description). Our K-theoretic flat sections $\mathfrak{s}(E)$ are invariant under the shift operators, and our main result suggests that shift operators for toric stacks should be defined globally on the secondary toric variety.

3.2. Specialization of Novikov variables

In this section we show that the quantum product, the flat sections for the quantum connection, and the K-group framing remain well-defined after the specialization Q = 1 of the Novikov variable Q. Recall that τ^0, \ldots, τ^N are co-ordinates on $H^{\bullet}_{\mathrm{CR},T}(X)$ dual to a homogeneous R_T -basis $\{\phi_0, \ldots, \phi_N\}$ of $H^{\bullet}_{\mathrm{CR},T}(X)$, and that:

- $\phi_0 = 1;$
- $\phi_1, \ldots, \phi_r \in H^2_T(X);$
- ϕ_1, \ldots, ϕ_r descend to a basis of $H^2(X) = H^2_T(X)/H^2_T(\text{pt})$.

Without loss of generality we may assume that the images of ϕ_1, \ldots, ϕ_r in $H^2(X)$ are nef and integral.

It is clear from Remark 2.1 that the specialization Q = 1 of the quantum product is well-defined, and we have:

$$\phi_i \star_{\tau} \phi_j \Big|_{Q=1} \in H^{\bullet}_{\mathrm{CR},T}(X) \otimes_{R_T} R_T \llbracket e^{\tau^1}, \dots, e^{\tau^r}, \tau^{r+1}, \dots, \tau^N \rrbracket$$

As discussed in Remark 2.1, the product $\phi_i \star_{\tau} \phi_j$ is independent of τ^0 . It is explained in [51, §2.5] that the specialization Q = 1 makes sense for $L(\tau, z)$, and:

$$L(\tau, z)\Big|_{Q=1} \in \operatorname{End}\left(H^{\bullet}_{\operatorname{CR}, T}(X)\right) \otimes_{R_T} R_T[\tau^0, \tau^1, \dots, \tau^r] \llbracket z^{-1} \rrbracket \llbracket e^{\tau^1}, \dots, e^{\tau^r}, \tau^{r+1}, \dots, \tau^N \rrbracket$$

The specialization Q = 1 for homogeneous flat sections $\mathfrak{s}(E)$ in Definition 3.1 (as well as the homogeneous flat sections s_i in Corollary 2.6) also makes sense and we have

$$\mathfrak{s}(E)(\tau, z)\Big|_{Q=1} \in H^{\bullet}_{\mathrm{CR}, T}(X) \otimes_{R_T} R_T[\tau^0, \tau^1, \dots, \tau^r, \log z]((z^{-1/k}))[\![e^{\tau^1}, \dots, e^{\tau^r}, \tau^{r+1}, \dots, \tau^N]\!]$$

where $k \in \mathbb{N}$ is such that all the eigenvalues of $k\mu$ are integral.

4. Toric Deligne–Mumford stacks as GIT quotients

In the rest of this paper we consider toric Deligne–Mumford stacks X with semiprojective coarse moduli space such that the torus-fixed set X^T is non-empty. This is the class of stacks that arise as GIT quotients of a complex vector space by the action of a complex torus. In this section we establish notation and describe basic properties of these quotients. Good introductions to this material include [4, §VII], [35] and [11].

4.1. GIT data

Consider the following data:

- $K \cong (\mathbb{C}^{\times})^r$, a connected torus of rank r;
- $\mathbb{L} = \operatorname{Hom}(\mathbb{C}^{\times}, K)$, the cocharacter lattice of K;
- $D_1, \ldots, D_m \in \mathbb{L}^{\vee} = \operatorname{Hom}(K, \mathbb{C}^{\times})$, characters of K.

The characters D_1, \ldots, D_m define a map from K to the torus $T = (\mathbb{C}^{\times})^m$, and hence define an action of K on \mathbb{C}^m .

Notation 4.1. For a subset I of $\{1, 2, \ldots, m\}$, write \overline{I} for the complement of I, and set

$$\mathcal{L}_{I} = \left\{ \sum_{i \in I} a_{i} D_{i} : a_{i} \in \mathbb{R}, \, a_{i} > 0 \right\} \subset \mathbb{L}^{\vee} \otimes \mathbb{R},$$
$$(\mathbb{C}^{\times})^{I} \times \mathbb{C}^{\overline{I}} = \left\{ (z_{1}, \dots, z_{m}) : z_{i} \neq 0 \text{ for } i \in I \right\} \subset \mathbb{C}^{m}.$$

We set $\angle_{\emptyset} := \{0\}.$

Definition 4.2. Consider now a *stability condition* $\omega \in \mathbb{L}^{\vee} \otimes \mathbb{R}$, and set:

$$\mathcal{A}_{\omega} = \left\{ I \subset \{1, 2, \dots, m\} : \omega \in \angle_I \right\}$$
$$U_{\omega} = \bigcup_{I \in \mathcal{A}_{\omega}} (\mathbb{C}^{\times})^I \times \mathbb{C}^{\overline{I}}$$
$$X_{\omega} = \left[U_{\omega} / K \right]$$

The square brackets here indicate that X_{ω} is the stack quotient of U_{ω} (which is *K*-invariant) by *K*. We call X_{ω} the *toric stack associated to the GIT data* $(K; \mathbb{L}; D_1, \ldots, D_m; \omega)$. We refer to elements of \mathcal{A}_{ω} as *anticones*, for reasons which will become clear in §4.2 below.

Assumption 4.3. We assume henceforth that:

(1) $\{1, 2, ..., m\} \in \mathcal{A}_{\omega};$ (2) for each $I \in \mathcal{A}_{\omega}$, the set $\{D_i : i \in I\}$ spans $\mathbb{L}^{\vee} \otimes \mathbb{R}$ over \mathbb{R} .

These are assumptions on the stability condition ω . The first ensures that X_{ω} is nonempty; the second ensures that X_{ω} is a Deligne–Mumford stack. Under these assumptions, \mathcal{A}_{ω} is closed under enlargement of sets, i.e. if $I \in \mathcal{A}_{\omega}$ and $I \subset J$ then $J \in \mathcal{A}_{\omega}$. Let $S \subset \{1, 2, ..., m\}$ denote the set of indices i such that $\{1, ..., m\} \setminus \{i\} \notin \mathcal{A}_{\omega}$. It is easy to see that the characters $\{D_i : i \in S\}$ are linearly independent and that every element of \mathcal{A}_{ω} contains S as a subset. Therefore we can write

$$\mathcal{A}_{\omega} = \{ I \sqcup S : I \in \mathcal{A}_{\omega}' \}$$

$$U_{\omega} \cong U_{\omega}' \times (\mathbb{C}^{\times})^{|S|}$$

$$(4.1)$$

for some $\mathcal{A}'_{\omega} \subset 2^{\{1,\ldots,m\}\setminus S}$ and an open subset U'_{ω} of $\mathbb{C}^{m-|S|}$. The toric stack X_{ω} can be also written as the quotient $[U'_{\omega}/G]$ of U'_{ω} for $G = \operatorname{Ker}(K \to (\mathbb{C}^{\times})^{|S|})$: this corresponds to the original construction of toric Deligne–Mumford stacks by Borisov–Chen–Smith [11].

The space of stability conditions $\omega \in \mathbb{L}^{\vee} \otimes \mathbb{R}$ satisfying Assumption 4.3 has a wall and chamber structure. The chamber C_{ω} to which ω belongs is given by

$$C_{\omega} = \bigcap_{I \in \mathcal{A}_{\omega}} \angle_{I},\tag{4.2}$$

and $X_{\omega} \cong X_{\omega'}$ as long as $\omega' \in C_{\omega}$. The GIT quotient $X_{\omega'}$ changes when ω' crosses a codimension-one boundary of C_{ω} . We call C_{ω} the *extended ample cone*; as we will see in §4.5 below, it is the product of the ample cone for X_{ω} with a simplicial cone.

4.2. GIT data and stacky fans

In the foundational work of Borisov–Chen–Smith [11], toric DM stacks are defined in terms of *stacky fans*. Jiang [53] introduced the notion of an *extended* stacky fan, which is a stacky fan with extra data. Our GIT data above are in one-to-one correspondence with extended stacky fans satisfying certain conditions, as we now explain.

An S-extended stacky fan is a quadruple $\Sigma = (\mathbf{N}, \Sigma, \beta, S)$, where:

- N is a finitely generated abelian group⁴;
- Σ is a rational simplicial fan in $\mathbf{N} \otimes \mathbb{R}$;
- $\beta: \mathbb{Z}^m \to \mathbf{N}$ is a homomorphism; we write $b_i = \beta(e_i) \in \mathbf{N}$ for the image of the *i*th standard basis vector $e_i \in \mathbb{Z}^m$, and write \overline{b}_i for the image of b_i in $\mathbf{N} \otimes \mathbb{R}$;
- $S \subset \{1, \ldots, m\}$ is a subset,

such that:

- each one-dimensional cone of Σ is spanned by \overline{b}_i for a unique $i \in \{1, \ldots, m\} \setminus S$, and each \overline{b}_i with $i \in \{1, \ldots, m\} \setminus S$ spans a one-dimensional cone of Σ ;
- for $i \in S$, \overline{b}_i lies in the support $|\Sigma|$ of the fan.

 $^{^4\,}$ Note that ${\bf N}$ may have torsion.

The vectors b_i for $i \in S$ are called *extended vectors*. Stacky fans as considered by Borisov–Chen–Smith correspond to the cases where $S = \emptyset$. For an extended stacky fan $(\mathbf{N}, \Sigma, \beta, S)$, the *underlying stacky fan* is the triple $(\mathbf{N}, \Sigma, \beta')$ where $\beta' : \mathbb{Z}^{m-|S|} \to \mathbf{N}$ is obtained from β by deleting the columns corresponding to $S \subset \{1, \ldots, m\}$. The toric Deligne–Mumford stack associated to an extended stacky fan $(\mathbf{N}, \Sigma, \beta, S)$ depends only on the underlying stacky fan.

To obtain an extended stacky fan from our GIT data, consider the exact sequence:

$$0 \longrightarrow \mathbb{L} \longrightarrow \mathbb{Z}^m \xrightarrow{\beta} \mathbf{N} \longrightarrow 0 \tag{4.3}$$

where the map from \mathbb{L} to \mathbb{Z}^m is given by (D_1, \ldots, D_m) and $\beta \colon \mathbb{Z}^m \to \mathbf{N}$ is the cokernel of the map $\mathbb{L} \to \mathbb{Z}^m$. Let $b_i = \beta(e_i) \in \mathbf{N}$ and $\overline{b}_i \in \mathbf{N} \otimes \mathbb{R}$ be as above and, given a subset I of $\{1, \ldots, m\}$, let σ_I denote the cone in $\mathbf{N} \otimes \mathbb{R}$ generated by $\{\overline{b}_i : i \in I\}$. The extended stacky fan $\Sigma_{\omega} = (\mathbf{N}, \Sigma_{\omega}, \beta, S)$ corresponding to our data consists of the group \mathbf{N} and the map β defined above, together with a fan Σ_{ω} in $\mathbf{N} \otimes \mathbb{R}$ and S given by⁵:

$$\Sigma_{\omega} = \{ \sigma_I : \overline{I} \in \mathcal{A}_{\omega} \}, \\ S = \{ i \in \{1, \dots, m\} : \overline{\{i\}} \notin \mathcal{A}_{\omega} \}.$$

The quotient construction in [53, §2] coincides with that in Definition 4.2, and therefore X_{ω} is the toric Deligne–Mumford stack corresponding to Σ_{ω} . Extended stacky fans $(\mathbf{N}, \Sigma_{\omega}, \beta, S)$ corresponding to GIT data satisfy the following conditions:

- (1) the support $|\Sigma_{\omega}|$ of the fan is convex and full-dimensional;
- (2) there is a strictly convex piecewise-linear function $f: |\Sigma_{\omega}| \to \mathbb{R}$ that is linear on each cone of Σ_{ω} ;
- (3) the map $\beta \colon \mathbb{Z}^m \to \mathbf{N}$ is surjective.

The first two conditions are geometric constraints on X_{ω} : they are equivalent to saying that the corresponding toric stack X_{ω} is semi-projective and has a torus fixed point. The third condition can be always achieved by adding enough extended vectors.

Conversely, given an extended stacky fan $\Sigma = (\mathbf{N}, \Sigma, \beta, S)$ satisfying the conditions (1)–(3) just stated, we can obtain GIT data as follows. Define a free \mathbb{Z} -module \mathbb{L} by the exact sequence (4.3) and define $K := \mathbb{L} \otimes \mathbb{C}^{\times}$. The dual of (4.3) is an exact sequence:

$$0 \longrightarrow \mathbf{N}^{\vee} \longrightarrow (\mathbb{Z}^m)^{\vee} \longrightarrow \mathbb{L}^{\vee}$$

$$(4.4)$$

and we define the character $D_i \in \mathbb{L}^{\vee}$ of K to be the image of the *i*th standard basis vector in $(\mathbb{Z}^m)^{\vee}$ under the third arrow $(\mathbb{Z}^m)^{\vee} \to \mathbb{L}^{\vee}$. Set:

⁵ This is why we refer to the elements of \mathcal{A}_{ω} as anticones.

$$\mathcal{A}_{\omega} = \{ I \subset \{1, 2, \cdots, m\} : S \subset I, \ \sigma_{\overline{I}} \text{ is a cone of } \Sigma \}$$

and take the stability condition $\omega \in \mathbb{L}^{\vee} \otimes \mathbb{R}$ to lie in $\bigcap_{I \in \mathcal{A}_{\omega}} \angle_{I}$; the condition (2) ensures that this intersection is non-empty. This specifies the data in Definition 4.2.

4.3. Torus-equivariant cohomology

The action of $T = (\mathbb{C}^{\times})^m$ on U_{ω} descends to a $\mathcal{Q} := T/K$ -action on X_{ω} . We also consider an ineffective *T*-action on X_{ω} induced by the projection $T \to \mathcal{Q}$. The \mathcal{Q} -equivariant and *T*-equivariant cohomology of X_{ω} are modules over $R_{\mathcal{Q}} := H^{\bullet}_{\mathcal{Q}}(\mathrm{pt}; \mathbb{C})$ and $R_T := H^{\bullet}_T(\mathrm{pt}; \mathbb{C})$ respectively. By the exact sequence (4.3), the Lie algebra of \mathcal{Q} is identified with $\mathbf{N} \otimes \mathbb{C}$ and $R_{\mathcal{Q}} \cong \mathrm{Sym}^{\bullet}(\mathbf{N}^{\vee} \otimes \mathbb{C})$. Let $\lambda_i \in R_T$ be the equivariant first Chern class of the irreducible *T*-representation given by the projection $T \cong (\mathbb{C}^{\times})^m \to \mathbb{C}^{\times}$ to the *i*th factor. Then $R_T = \mathbb{C}[\lambda_1, \ldots, \lambda_m]$. It is well-known that:

$$H^{\bullet}_{\mathcal{Q}}(X_{\omega};\mathbb{C}) = R_{\mathcal{Q}}[u_1,\ldots,u_m]/(\mathfrak{I}+\mathfrak{J})$$

$$(4.5)$$

where u_i is the Q-equivariant class Poincaré-dual to the toric divisor:

$$\{(z_1, \dots, z_m) \in U_\omega : z_i = 0\}/K$$
(4.6)

and \mathfrak{I} and \mathfrak{J} are the ideals of additive and multiplicative relations:

$$\begin{aligned} \mathfrak{I} &= \big\langle \chi - \sum_{i=1}^m \langle \chi, b_i \rangle \, u_i : \chi \in \mathbf{N}^{\vee} \otimes \mathbb{C} \big\rangle, \\ \mathfrak{J} &= \big\langle \prod_{i \notin I} u_i : I \notin \mathcal{A}_{\omega} \big\rangle. \end{aligned}$$

Note that $u_i = 0$ for $i \in S$ because the corresponding divisor (4.6) is empty (see equation (4.1)). Indeed, this relation is contained in the ideal \mathfrak{J} . The *T*-equivariant cohomology is given by the extension of scalars:

$$H^{\bullet}_{T}(X_{\omega}) \cong H^{\bullet}_{\mathcal{O}}(X_{\omega}) \otimes_{R_{\mathcal{O}}} R_{T}$$

where the algebra homomorphism $R_{\mathcal{Q}} \to R_T$ is given by $\chi \mapsto \sum_{i=1}^m \langle \chi, b_i \rangle \lambda_i$ for $\chi \in \mathbf{N}^{\vee} \otimes \mathbb{C}$.

Remark 4.4. We note that the assumptions at the beginning of §2 are satisfied for toric Deligne–Mumford stacks obtained from GIT data. First, all the Q-weights appearing in the Q-representation $H^0(X_{\omega}, \mathcal{O})$ are contained in the strictly convex cone $|\Sigma_{\omega}|^{\vee} =$ $\{\chi \in \mathbf{N}^{\vee} \otimes \mathbb{R} : \langle \chi, v \rangle \geq 0 \text{ for all } v \in |\Sigma_{\omega}|\}$. Second, X_{ω} is equivariantly formal since the cohomology group of X_{ω} is generated by Q-invariant cycles [45]. Because each component of IX_{ω} is again a toric stack given by certain GIT data (see §4.8), we have that IX_{ω} is also equivariantly formal. The same conclusions hold for the *T*-action.

4.4. Second cohomology and homology

There is a commutative diagram:

with exact rows and columns. Note that we have $H^2_{\mathcal{Q}}(\mathrm{pt};\mathbb{R}) \cong \mathbb{N}^{\vee} \otimes \mathbb{R}$, $H^2_T(\mathrm{pt};\mathbb{R}) \cong \mathbb{R}^m$, $H^2_K(\mathrm{pt};\mathbb{R}) \cong \mathbb{L}^{\vee} \otimes \mathbb{R}$. The top row of (4.7) is identified with the exact sequence (4.4) tensored with \mathbb{R} . By (4.5), $H^2_{\mathcal{Q}}(X_{\omega};\mathbb{R})$ is freely generated by the classes $u_i, i \in \{1,\ldots,m\} \setminus S$ of toric divisors, and hence $H^2_{\mathcal{Q}}(X_{\omega};\mathbb{R}) \cong \mathbb{R}^{m-|S|}$. The leftmost column is identified with the exact sequence

$$0 \longrightarrow \mathbf{N}^{\vee} \otimes \mathbb{R} \longrightarrow \mathbb{R}^{m-|S|} \longrightarrow \mathbb{L}^{\vee} \otimes \mathbb{R} / \sum_{i \in S} \mathbb{R} D_i \longrightarrow 0$$

induced by (4.4). In particular we have

$$H^2(X_\omega; \mathbb{R}) \cong \mathbb{L}^{\vee} \otimes \mathbb{R} / \sum_{i \in S} \mathbb{R} D_i$$

where the non-equivariant limit of u_i is identified with the class of D_i . The homology group $H_2(X_{\omega}; \mathbb{R})$ is identified with $\bigcap_{i \in S} \operatorname{Ker}(D_i)$ in $\mathbb{L} \otimes \mathbb{R}$. The square at the upper left of (4.7) is a pushout and we have:

$$H_T^2(X_{\omega};\mathbb{R}) \cong \bigoplus_{i \in \{1,\dots,m\} \setminus S} \mathbb{R}u_i \oplus \bigoplus_{i=1}^m \mathbb{R}\lambda_i \Big/ \big\langle \sum_{i=1}^m \langle \chi, b_i \rangle (u_i - \lambda_i) \colon \chi \in \mathbf{N}^{\vee} \otimes \mathbb{R} \big\rangle.$$

It follows that the middle row of (4.7) splits canonically: we have a well-defined homomorphism⁶

⁶ More precisely $(-\theta)$ gives a splitting of the middle row of (4.7).

$$\theta \colon \mathbb{L}^{\vee} \otimes \mathbb{R} \cong H^2_K(\mathrm{pt}; \mathbb{R}) \longrightarrow H^2_T(X_{\omega}; \mathbb{R})$$

$$(4.8)$$

such that $\theta(D_i) = u_i - \lambda_i$ and that

$$H^2_T(X_\omega; \mathbb{R}) \cong H^2_\mathcal{Q}(X_\omega; \mathbb{R}) \oplus \theta(\mathbb{L}^{\vee} \otimes \mathbb{R}).$$

The class $\theta(p)$ can be written as the *T*-equivariant first Chern class of a certain line bundle L(p) associated to p (see §6.3.2). One advantage of working with *T*-equivariant cohomology instead of Q-equivariant cohomology is the existence of this canonical splitting.

We also introduce a canonical splitting of the projection $\mathbb{L}^{\vee} \otimes \mathbb{R} \to \mathbb{L}^{\vee} \otimes \mathbb{R} / \sum_{i \in S} \mathbb{R} D_i \cong H^2(X_{\omega}, \mathbb{R})$. This is equivalent to choosing a complementary subspace of $H_2(X_{\omega}; \mathbb{R})$ in $\mathbb{L} \otimes \mathbb{R}$. Take $j \in S$. The corresponding extended vector $\overline{b_j} \in \mathbb{N} \otimes \mathbb{R}$ lies in the support of the fan. Let $\sigma_{I_j} \in \Sigma$, $I_j \subset \{1, \ldots, m\} \setminus S$ be the minimal cone⁷ containing $\overline{b_j}$ and write $\overline{b_j} = \sum_{i \in I_j} c_{ij} \overline{b_i}$ for some $c_{ij} \in \mathbb{R}_{>0}$. By the exact sequence (4.3), there exists an element $\xi_j \in \mathbb{L} \otimes \mathbb{Q}$ such that

$$D_{i} \cdot \xi_{j} = \begin{cases} 1 & \text{if } i = j; \\ -c_{ij} & \text{if } i \in I_{j}; \\ 0 & \text{if } i \notin I_{j} \cup \{j\}. \end{cases}$$
(4.9)

Note that one has $D_i \cdot \xi_j = \delta_{ij}$ for $i, j \in S$. Hence $\{\xi_i\}_{i \in S}$ spans a complementary subspace of $H_2(X_{\omega}; \mathbb{R}) = \bigcap_{j \in S} \operatorname{Ker}(D_j) \subset \mathbb{L} \otimes \mathbb{R}$ and defines a splitting:

$$\mathbb{L} \otimes \mathbb{R} \cong H_2(X_\omega; \mathbb{R}) \oplus \bigoplus_{j \in S} \mathbb{R}\xi_j,$$
(4.10)

or, for the dual space,

$$\mathbb{L}^{\vee} \otimes \mathbb{R} \cong \bigcap_{j \in S} \operatorname{Ker}(\xi_j) \oplus \bigoplus_{j \in S} \mathbb{R}D_j$$
(4.11)

with $\bigcap_{j \in S} \operatorname{Ker}(\xi_j) \cong H^2(X_{\omega}; \mathbb{R}).$

The equivariant first Chern class of TX_{ω} is given by:

$$\rho = c_1^{\mathcal{Q}}(TX_{\omega}) = c_1^T(TX_{\omega}) = \sum_{i \in \{1, \dots, m\} \setminus S} u_i$$

⁷ Minimality is not essential here. Let $\sigma_I \in \Sigma$, $I \subset \{1, \ldots, m\}$, be any cone containing $\overline{b_j}$ and write $\overline{b_j} = \sum_{i \in I} c_{ij} \overline{b_i}$ for some $c_{ij} \in \mathbb{R}_{\geq 0}$. In our setting, the vectors $\{b_i : i \in I\}$ are linearly independent for any choice of cone σ_I , and so the coefficients c_{ij} here are unique. In particular, therefore, we have that $c_{ij} = 0$ for $i \in I \setminus I_j$.

4.5. Ample cone and Mori cone

Let D'_i denote the image of D_i in $\mathbb{L}^{\vee} \otimes \mathbb{R} / \sum_{i \in S} \mathbb{R} D_i \cong H^2(X_{\omega}; \mathbb{R})$. This is the non-equivariant Poincaré dual of the toric divisor (4.6), that is, the non-equivariant limit of u_i . The cone of ample divisors of X_{ω} is given by

$$C'_{\omega} = \bigcap_{I \in \mathcal{A}'_{\omega}} \angle'_{I}$$

where \mathcal{A}'_{ω} was introduced in equation (4.1) and $\angle'_I := \sum_{i \in I} \mathbb{R}_{>0} D'_i$ is an open cone in $\mathbb{L}^{\vee} \otimes \mathbb{R} / \sum_{i \in S} \mathbb{R} D_i$ (cf. Notation 4.1). Under the splitting (4.11) of $\mathbb{L}^{\vee} \otimes \mathbb{R}$, the extended ample cone C_{ω} defined in equation (4.2) also splits [49, Lemma 3.2]:

$$C_{\omega} \cong C'_{\omega} \times \left(\sum_{i \in S} \mathbb{R}_{>0} D_i\right) \subset H^2(X_{\omega}; \mathbb{R}) \times \bigoplus_{i \in S} \mathbb{R} D_i.$$

$$(4.12)$$

The Mori cone is the dual cone of C'_{ω} :

$$\operatorname{NE}(X_{\omega}) = C'_{\omega} = \{ d \in H_2(X_{\omega}; \mathbb{R}) : \eta \cdot d \ge 0 \text{ for all } \eta \in C'_{\omega} \}$$

4.6. Fixed points and isotropy groups

Fixed points of the *T*-action on X_{ω} are in one-to-one correspondence with minimal anticones, that is, with $\delta \in \mathcal{A}_{\omega}$ such that $|\delta| = r$. A minimal anticone δ corresponds to the *T*-fixed point:

$$\{(z_1,\ldots,z_n)\in U_\omega: z_i=0 \text{ if } i\notin \delta\}/K$$

We now describe the isotropy of the Deligne–Mumford stack X_{ω} , i.e. those elements $g \in K$ such that the action of g on U_{ω} has fixed points. Recall that there are canonical isomorphisms $K \cong \mathbb{L} \otimes \mathbb{C}^{\times}$ and $\text{Lie}(K) \cong \mathbb{L} \otimes \mathbb{C}$, via which the exponential map $\text{Lie}(K) \to K$ becomes $\text{id} \otimes \exp(2\pi i -) : \mathbb{L} \otimes \mathbb{C} \to \mathbb{L} \otimes \mathbb{C}^{\times}$. The kernel of the exponential map is $\mathbb{L} \subset \mathbb{L} \otimes \mathbb{C}$. Define $\mathbb{K} \subset \mathbb{L} \otimes \mathbb{Q}$ to be the set of $f \in \mathbb{L} \otimes \mathbb{Q}$ such that:

$$I_f := \left\{ i \in \{1, 2, \dots, m\} : D_i \cdot f \in \mathbb{Z} \right\} \in \mathcal{A}_{\omega}$$

$$(4.13)$$

The lattice \mathbb{L} acts on \mathbb{K} by translation, and elements $g \in K$ such that the action of g on U_{ω} has fixed points correspond, via the exponential map, to elements of \mathbb{K}/\mathbb{L} .

4.7. Floors, ceilings, and fractional parts

For a rational number q, we write:

- $\lfloor q \rfloor$ for the largest integer n such that $n \leq q$;
- $\lceil q \rceil$ for the smallest integer n such that $q \leq n$; and
- $\langle q \rangle$ for the fractional part $q \lfloor q \rfloor$ of q.

4.8. The inertia stack and Chen-Ruan cohomology

Recall the definition of the inertia stack IX_{ω} from §2.1. Components of IX_{ω} are indexed by elements of \mathbb{K}/\mathbb{L} : the component X_{ω}^{f} of IX_{ω} corresponding to $f \in \mathbb{K}/\mathbb{L}$ consists of the points (x, g) in IX_{ω} such that $g = \exp(2\pi i f)$. Recall the set I_{f} defined in (4.13). The component X_{ω}^{f} in the inertia stack IX_{ω} is the toric Deligne–Mumford stack with GIT data given by K, \mathbb{L} , and ω exactly as for X_{ω} , and characters $D_{i} \in \mathbb{L}^{\vee}$ for $i \in I_{f}$. We have:

$$X^f_{\omega} = [\mathbb{C}^{I_f} \cap U_{\omega}/K].$$

The inclusion $\mathbb{C}^{I_f} \subset \mathbb{C}^m$ exhibits X^f_{ω} as a closed substack of the toric stack X_{ω} . According to Borisov–Chen–Smith [11], components of the inertia stack of X_{ω} are indexed by elements of the set $Box(X_{\omega})$:

$$\operatorname{Box}(X_{\omega}) = \left\{ v \in \mathbf{N} : \overline{v} = \sum_{i \notin I} c_i \overline{b_i} \text{ in } \mathbf{N} \otimes \mathbb{R} \text{ for some } I \in \mathcal{A} \text{ and } 0 \le c_i < 1 \right\}$$

In fact, we have an isomorphism [49, \$3.1.3]:

$$\mathbb{K}/\mathbb{L} \cong \operatorname{Box}(X_{\omega}) \qquad [f] \mapsto v_f = \sum_{i=1}^m \lceil -(D_i \cdot f) \rceil b_i \in \mathbf{N}.$$
(4.14)

When $j \in S$ and $b_j \in Box(X_{\omega})$, the element $-\xi_j \in \mathbb{L} \otimes \mathbb{Q}$ defined in (4.9) belongs to \mathbb{K} and corresponds to b_j .

The age ι_f of the component $X^f_{\omega} \subset IX_{\omega}$ is $\sum_{i \notin I_f} \langle D_i \cdot f \rangle$. The *T*-equivariant Chen-Ruan cohomology of X_{ω} is, as we saw in §2.2, the *T*-equivariant cohomology of the inertia stack IX_{ω} with age-shifted grading:

$$H^{\bullet}_{\mathrm{CR},T}(X_{\omega};\mathbb{Q}) = \bigoplus_{f \in \mathbb{K}/\mathbb{L}} H^{\bullet-2\iota_f}_T(X^f_{\omega};\mathbb{Q})$$

This contains the *T*-equivariant cohomology of X_{ω} as a summand, corresponding to the element $0 \in \mathbb{K}/\mathbb{L}$; furthermore the fact that each X^f_{ω} is a closed substack of X_{ω} implies that $H^{\bullet}_{\mathrm{CR},T}(X_{\omega};\mathbb{Q})$ is naturally a module over $H^{\bullet}_{T}(X_{\omega};\mathbb{Q})$. We write $\mathbf{1}_{f}$ for the unit class in $H^{0}_{T}(X^{f}_{\omega};\mathbb{Q})$, regarded as an element of $H^{2\iota_{f}}_{\mathrm{CR},T}(X_{\omega};\mathbb{Q})$.

Recall that the component X^f_{ω} of the inertia stack is the toric Deligne–Mumford stack with GIT data $(K; \mathbb{L}; \omega; D_i : i \in I_f)$. In particular, therefore, the anticones for X^f_{ω} are given by $\{I \in \mathcal{A}_{\omega} : I \subset I_f\}$. *T*-fixed points on the inertia stack IX_{ω} are indexed by pairs (δ, f) where δ is a minimal anticone in $\mathcal{A}_{\omega}, f \in \mathbb{K}/\mathbb{L}$, and $D_i \cdot f \in \mathbb{Z}$ for all $i \in \delta$. The pair (δ, f) determines a *T*-fixed point on the component X_{ω}^f of the inertia stack: the *T*-fixed point that corresponds to the minimal anticone $\delta \subset I_f$.

5. Wall-crossing in toric Gromov–Witten theory

In this section we consider crepant birational transformations $X_+ \dashrightarrow X_-$ between toric Deligne–Mumford stacks which arise from variation of GIT. We use the Mirror Theorem for toric Deligne–Mumford stacks [26,29] to construct a global equivariant quantum connection over (a certain part of) the secondary toric variety for X_{\pm} ; this gives an analytic continuation of the equivariant quantum connections for X_+ and X_- .

5.1. Birational transformations from wall-crossing

Recall that our GIT data in §4.1 consist of a torus $K \cong (\mathbb{C}^{\times})^r$, the lattice $\mathbb{L} = \text{Hom}(\mathbb{C}^{\times}, K)$ of \mathbb{C}^{\times} -subgroups of K, and characters $D_1, \ldots, D_m \in \mathbb{L}^{\vee}$. Recall further that a choice of stability condition $\omega \in \mathbb{L}^{\vee} \otimes \mathbb{R}$ satisfying Assumption 4.3 determines a toric Deligne–Mumford stack $X_{\omega} = [U_{\omega}/K]$. The space $\mathbb{L}^{\vee} \otimes \mathbb{R}$ of stability conditions is divided into chambers by the closures of the sets \angle_I , |I| = r-1, and the Deligne–Mumford stack X_{ω} depends on ω only via the chamber containing ω . For any stability condition ω satisfying Assumption 4.3, the set U_{ω} contains the big torus $T = (\mathbb{C}^{\times})^m$, and thus for any two such stability conditions ω_1, ω_2 there is a canonical birational map $X_{\omega_1} \dashrightarrow X_{\omega_2}$, induced by the identity transformation between $T/K \subset X_{\omega_1}$ and $T/K \subset X_{\omega_2}$. Our setup is as follows. Let C_+ , C_- be chambers in $\mathbb{L}^{\vee} \otimes \mathbb{R}$ that are separated by a hyperplane wall W, so that $W \cap \overline{C_+}$ is a facet of $\overline{C_+}, W \cap \overline{C_-}$ is a facet of $\overline{C_-}$, and $W \cap \overline{C_+} = W \cap \overline{C_-}$. Choose stability conditions $\omega_+ \in C_+, \omega_- \in C_-$ satisfying Assumption 4.3 and set $X_+ := X_{\omega_+}, X_- := X_{\omega_-}$, and

$$\mathcal{A}_{\pm} := \mathcal{A}_{\omega_{\pm}} = \left\{ I \subset \{1, 2, \dots, m\} : \omega_{\pm} \in \angle_I \right\}$$

Then $C_{\pm} = \bigcap_{I \in \mathcal{A}_{\pm}} \angle_I$. Let $\varphi \colon X_+ \dashrightarrow X_-$ be the birational transformation induced by the toric wall-crossing and suppose that

$$\sum_{i=1}^{m} D_i \in W$$

As we will see below this amounts to requiring that φ is crepant. Let $e \in \mathbb{L}$ denote the primitive lattice vector in W^{\perp} such that e is positive on C_{+} and negative on C_{-} .

Remark 5.1. The situation considered here is quite general. We do not require X_+ , X_- to have projective coarse moduli space (they are required to be semi-projective). We do

not require that X_+ , X_- are weak Fano, or that they satisfy the extended weak Fano condition in [49, §3.1.4]. In other words, we do not require $\sum_{i=1}^{m} D_i \in W$ to lie in the boundary $W \cap \overline{C_+} = W \cap \overline{C_-}$ of the extended ample cones.

Choose ω_0 from the relative interior of $W \cap \overline{C_+} = W \cap \overline{C_-}$. The stability condition ω_0 does not satisfy our Assumption 4.3, but we can still consider:

$$\mathcal{A}_0 := \mathcal{A}_{\omega_0} = \{ I \subset \{1, \dots, m\} : \omega_0 \in \angle_I \}$$

and the corresponding toric (Artin) stack $X_0 := X_{\omega_0} = [U_{\omega_0}/K]$ as given in Definition 4.2. Here X_0 is not Deligne–Mumford, as the \mathbb{C}^{\times} -subgroup of K corresponding to $e \in \mathbb{L}$ (the defining equation of the wall W) has a fixed point in U_{ω_0} . The stack X_0 contains both X_+ and X_- as open substacks and the canonical line bundles of X_+ and $X_$ are the restrictions of the same line bundle $L_0 \to X_0$ given by the character $-\sum_{i=1}^m D_i$ of K. The condition $\sum_{i=1}^m D_i \in W$ ensures that L_0 comes from a Q-Cartier divisor on the underlying singular toric variety $\overline{X}_0 = \mathbb{C}^m /\!\!/ \omega_0 K$ associated to the fan Σ_{ω_0} . On the other hand, in §6.3, we shall construct a toric Deligne–Mumford stack \widetilde{X} equipped with proper birational morphisms $f_{\pm} \colon \widetilde{X} \to X_{\pm}$ such that the diagram (1.3) commutes. Then $f_{\pm}^*(K_{X_+})$ and $f_{-}^*(K_{X_-})$ coincide since they are the pull-backs of a Q-Cartier divisor on \overline{X}_0 . This is what is meant by the birational map φ being crepant.⁸

$$M_{\pm} = \{ i \in \{1, \dots, m\} : \pm D_i \cdot e > 0 \},\$$

$$M_0 = \{ i \in \{1, \dots, m\} : D_i \cdot e = 0 \}.$$

Our assumptions imply that both M_+ and M_- are non-empty. The following lemma is easy to check:

Lemma 5.2. Set:

$$\mathcal{A}_0^{\text{thin}} := \{ I \in \mathcal{A}_0 : I \subset M_0 \}$$
$$\mathcal{A}_0^{\text{thick}} := \{ I \in \mathcal{A}_0 : I \cap M_+ \neq \emptyset, I \cap M_- \neq \emptyset \}.$$

Then one has $M_0 \in \mathcal{A}_0^{\text{thin}}$ and

$$\mathcal{A}_{0} = \mathcal{A}_{0}^{\text{thin}} \sqcup \mathcal{A}_{0}^{\text{thick}},$$
$$\mathcal{A}_{\pm} = \mathcal{A}_{0}^{\text{thick}} \sqcup \left\{ I \sqcup J : \emptyset \neq J \subset M_{\pm}, I \in \mathcal{A}_{0}^{\text{thin}} \right\}.$$

Remark 5.3. Let Σ_{\pm} be the fans of X_{\pm} . In terms of fans, a toric wall-crossing can be described as a *modification along a circuit* [12,38], where 'circuit' means a minimal

 $^{^{8}\,}$ This notion is also called K-equivalence: see the Introduction.

linearly dependent set of vectors. In our wall-crossing, the relevant circuit is $\{b_i : i \in M_+ \cup M_-\}$: we have $\sum_{i \in M_+ \cup M_-} (D_i \cdot e)b_i = 0$, and every proper subset of $\{b_i : i \in M_+ \cup M_-\}$ is linearly independent. The partition of the circuit $M_+ \cup M_-$ into M_+ and M_- is determined by the sign of the coefficients in a relation among $\{b_i : i \in M_+ \cup M_-\}$. The modification along the circuit $M_+ \cup M_-$ turns the fan Σ_+ into Σ_- : it removes every cone σ_I of Σ_+ such that I contains M_- but not M_+ and introduces cones of the form σ_K where $K = (I \cup M_+) \setminus J$ for any non-empty subset $J \subset M_-$. This description matches with Lemma 5.2, §4.2, and §4.1.

There are three types of possible crepant toric wall-crossings: (I) X_+ and X_- are isomorphic in codimension one ("flop"), (II) φ induces a morphism $X_+ \rightarrow |X_-|$ or $X_- \rightarrow |X_+|$ contracting a divisor to a toric subvariety ("crepant resolution") and (III) the rigidifications⁹ X_+^{rig} , X_-^{rig} are isomorphic (only the gerbe structures change; we call it a "gerbe flop"). Define:

$$S_{\pm} = \{i \in \{1, \dots, m\} : \overline{\{i\}} \notin \mathcal{A}_{\pm}\}.$$

Proposition 5.4. The intersection $S_0 := S_+ \cap S_-$ is contained in M_0 . Moreover, one and only one of the following holds:

- (II-i) there exists $i \in \{1, ..., m\}$ such that $S_{-} = S_{+} \sqcup \{i\}$, $M_{-} = \{i\}$ and $\sharp(M_{+}) \ge 2$;
- (II-ii) there exists $i \in \{1, ..., m\}$ such that $S_+ = S_- \sqcup \{i\}$, $M_+ = \{i\}$ and $\sharp(M_-) \ge 2$;
- (III) there exist $i_+, i_- \in \{1, \dots, m\}$ such that $S_+ = S_0 \sqcup \{i_+\}, S_- = S_0 \sqcup \{i_-\}, M_+ = \{i_+\}$ and $M_- = \{i_-\}.$

Proof. First we show that $S_0 \subset M_0$. Take $i \in S_0$. Suppose that $i \in M_+$. Since $M_0 \in \mathcal{A}_0^{\text{thin}}$, we have $M_0 \cup M_- \in \mathcal{A}_-$ by Lemma 5.2. Thus $\overline{\{i\}} = M_0 \cup M_- \cup (M_+ \setminus \{i\})$ also belongs to \mathcal{A}_- . This contradicts the fact that $i \in S_-$. Thus we have $i \notin M_+$, and similarly that $i \notin M_-$. Hence $i \in M_0$. We have shown that $S_0 \subset M_0$.

Next we claim that:

(a) if S₋ \ S₊ is non-empty, then we have \$\\$(S₋ \ S₊) = 1 and M₋ = S₋ \ S₊;
(b) if S₋ ⊂ S₊, then \$\\$(M₋) ≥ 2.

Take $i \in S_{-} \setminus S_{+}$. We have $\overline{\{i\}} \in \mathcal{A}_{+} \setminus \mathcal{A}_{-}$. Lemma 5.2 implies that an element of $\mathcal{A}_{+} \setminus \mathcal{A}_{-}$ is of the form $I \sqcup J$ with $\emptyset \neq J \subset M_{+}$ and $I \subset M_{0}$, and in particular does not intersect with M_{-} . This implies that $\{i\} = M_{-}$. Therefore $S_{-} \setminus S_{+} = M_{-}$ consists of only one element. This proves (a). Conversely, if $M_{-} = \{i\}$, it follows from Lemma 5.2

⁹ See e.g. [37].

that $\overline{\{i\}} \in \mathcal{A}_+ \setminus \mathcal{A}_-$ and thus $i \in S_- \setminus S_+$. This proves (b). The same claim holds if we exchange + and -. It follows that one and only one of (I), (II-i), (II-ii), (III) happens. \Box

Proposition 5.5. The loci of indeterminacy of φ and φ^{-1} are the toric substacks

$$\bigcap_{j \in M_{-}} \{z_{j} = 0\} \subset X_{+} \qquad and \qquad \bigcap_{j \in M_{+}} \{z_{j} = 0\} \subset X_{-}$$

respectively. With cases as in Proposition 5.4, we have:

- (I) X_+ and X_- are isomorphic in codimension one;
- (II-i) φ induces a morphism $\varphi \colon X_+ \to |X_-|$ that contracts the divisor $\{z_i = 0\}$ to the subvariety $\bigcap_{i \in M_+} \{z_j = 0\}$;
- (II-ii) a statement similar to (II-i) with + and interchanged;
- (III) φ induces an isomorphism $X_{+}^{\operatorname{rig}} \cong X_{-}^{\operatorname{rig}}$ between the rigidifications.

Proof. One can check that $U_{\omega_+} \cap U_{\omega_-} = U_{\omega_+} \setminus \bigcap_{i \in M_-} \{z_i = 0\} = U_{\omega_-} \setminus \bigcap_{i \in M_+} \{z_i = 0\}$ using Lemma 5.2. The geometric picture in each case can be seen from the stacky fans: (I) the sets of one-dimensional cones are the same; (II-i) the fan Σ_- is obtained by deleting the ray $\mathbb{R}_{\geq 0}\overline{b}_i$ from Σ_+ ; $\sigma_{M_+} \in \Sigma_-$ is a minimal cone containing \overline{b}_i ; φ contracts the toric divisor $\{z_i = 0\}$ to the closed subvariety associated with σ_{M_+} ; (II-ii) similar; (III) the stacky fan Σ_- is obtained from Σ_+ by replacing b_{i_-} with b_{i_+} ; one has $(D_{i_+} \cdot e)b_{i_+} = -(D_{i_-} \cdot e)b_{i_-}$ by (4.3) and $D_{i_+} \cdot e + D_{i_-} \cdot e = 0$; thus b_{i_+} and b_{i_-} differ only by a torsion element in **N**. \Box

Example 5.6.

(I) Let $a_1, \ldots, a_k, b_1, \ldots, b_l$ be positive integers such that $a_1 + \cdots + a_k = b_1 + \cdots + b_l$. Consider the GIT data given by $\mathbb{L}^{\vee} = \mathbb{Z}$, $D_1 = a_1, \ldots, D_k = a_k$, $D_{k+1} = -b_1, \ldots, D_{k+l} = -b_l$. If $k, l \ge 2$, we have a flop between

$$X_{+} = \bigoplus_{i=1}^{k} \mathcal{O}_{\mathbb{P}(a_1,\dots,a_k)}(-b_i) \quad \text{and} \quad X_{-} = \bigoplus_{j=1}^{l} \mathcal{O}_{\mathbb{P}(b_1,\dots,b_l)}(-a_j).$$

- (II) Consider the case as in (I) but with l = 1. Setting $d = a_1 + \cdots + a_k = b_1$, we have that $X_+ = \mathcal{O}_{\mathbb{P}(a_1,\ldots,a_k)}(-d)$ is a crepant (partial) resolution of $|X_-| = \mathbb{C}^k / \mu_d$ where μ_d acts on \mathbb{C}^k with weights (a_1,\ldots,a_k) .
- (III) Consider the GIT data given by $\mathbb{L}^{\vee} = \mathbb{Z}^2$, $D_1 = (1,0)$, $D_2 = (1,2)$, $D_3 = (0,2)$. Take ω_+ from the chamber $\{(x,y) : 0 < y < 2x\}$ and ω_- from the chamber $\{(x,y) : 0 < 2x < y\}$. Then we have a "gerbe flop" between $X_+ = \mathbb{P}(2,2)$ and $X_- = \mathbb{P}^1 \times B\boldsymbol{\mu}_2$.

5.2. Decompositions of extended ample cones

Recall the decomposition (4.11) of the vector space $\mathbb{L}^{\vee} \otimes \mathbb{R}$ and the decomposition (4.12) of the extended ample cone. In the case at hand, we have two (possibly different) decompositions of $\mathbb{L}^{\vee} \otimes \mathbb{R}$ associated to the GIT quotients X_+ and X_- :

$$\mathbb{L}^{\vee} \otimes \mathbb{R} = \bigcap_{j \in S_{\pm}} \operatorname{Ker}(\xi_j^{\pm}) \oplus \bigoplus_{j \in S_{\pm}} \mathbb{R}D_j$$
(5.1)

where elements $\xi_j^{\pm} \in \mathbb{L} \otimes \mathbb{R}$, $j \in S_{\pm}$ are as in (4.9) and $\bigcap_{j \in S_{\pm}} \operatorname{Ker}(\xi_i^+) \cong H^2(X_{\pm}, \mathbb{R})$. Under these decompositions, one has

$$C_{\pm} = C'_{\pm} \times \sum_{j \in S_{\pm}} \mathbb{R}_{>0} D_j$$

where $C'_{\pm} \subset \bigcap_{i \in S_{\pm}} \operatorname{Ker}(\xi_i^{\pm}) \cong H^2(X_{\pm}; \mathbb{R})$ is the ample cone of X_{\pm} . Let $\overline{C_W} := W \cap \overline{C_+} = W \cap \overline{C_-}$ be a common facet of C_+ and C_- , and write C_W for the relative interior of $\overline{C_W}$. We now show that these decompositions of the cones C_+ , C_- are compatible along the wall.

Proposition 5.7. We have $\xi_i^+|_W = \xi_i^-|_W$ for $i \in S_0 = S_+ \cap S_-$ and $\xi_i^{\pm}|_W = 0$ for $i \in S_{\pm} \setminus S_{\pm}$. Set $\xi_i^W = \xi_i^+|_W = \xi_i^-|_W \in \text{Hom}(W,\mathbb{R})$. Then we have

$$W' := W \cap \bigcap_{i \in S_+} \operatorname{Ker}(\xi_i^+) = W \cap \bigcap_{i \in S_-} \operatorname{Ker}(\xi_i^-) = \bigcap_{i \in S_0} \operatorname{Ker}(\xi_i^W)$$

and so the decompositions (5.1) restrict to the same decomposition of W:

$$W = W' \oplus \bigoplus_{i \in S_0} \mathbb{R}D_i.$$
(5.2)

Under this decomposition of W, the cone C_W decomposes as

$$C_W = C'_W \times \sum_{i \in S_0} \mathbb{R}_{>0} D_i$$

for some cone C'_W in W'. With cases as in Proposition 5.4, we have:

- (I) C'_W is a common facet of C'_+ and C'_- ;
- (II-i) $C'_W = C'_-, C'_W$ is a facet of C'_+ and $C_- = C_W + \mathbb{R}_{>0}D_i$;
- (II-ii) $C'_W = C'_+, C'_W$ is a facet of C'_- and $C_+ = C_W + \mathbb{R}_{>0}D_i$;
- (III) $C'_W = C'_+ = C'_-, C_+ = C_W + \mathbb{R}_{>0}D_{i_+}$ and $C_- = C_W + \mathbb{R}_{>0}D_{i_-}$.

Proof. It suffices to show that $\xi_i^+|_W = \xi_i^-|_W$ for $i \in S_0$ and that $\xi_i^\pm|_W = 0$ for $i \in S_{\pm} \setminus S_{\pm}$. The rest of the statements follow easily. Suppose that $i \in S_0$. Recall the

definition of ξ_i^{\pm} in (4.9). Let $\sigma_I \in \Sigma_+$ be the minimal cone containing $\overline{b_i}$. Then $\overline{I} \in \mathcal{A}_+$. If $\overline{I} \in \mathcal{A}_-$, we have $\xi_i^+ = \xi_i^-$ by the definition of ξ_i^{\pm} . Suppose that $\overline{I} \notin \mathcal{A}_-$. By Lemma 5.2, I contains M_- but not M_+ . We have a relation of the form:

$$\overline{b_i} = \sum_{j \in I} c_j \overline{b_j} \tag{5.3}$$

with $c_j > 0$. By adding to the right-hand side of (5.3) a suitable positive multiple of the relation

$$\sum_{j \in M_+} (D_j \cdot e)\overline{b_j} - \sum_{j \in M_-} (-D_j \cdot e)\overline{b_j} = 0$$

given by $e \in \mathbb{L}$ via (4.3), we obtain a relation of the form

$$\overline{b_i} = \sum_{j \in I'} c'_j \overline{b_j}$$

such that $c'_j > 0$ and $I' = (I \cup M_+) \setminus J$ with $\emptyset \neq J \subset M_-$. Then $\overline{I'} \in \mathcal{A}_-$ by Lemma 5.2 (see also Remark 5.3). Note that $c_j = c'_j$ if $j \in I \cap M_0 = I' \cap M_0$. This implies that $D_j \cdot \xi_i^+ = D_j \cdot \xi_i^-$ for all $j \in M_0$. Since $\{D_j : j \in M_0\}$ spans W, we have $\xi_i^+|_W = \xi_i^-|_W$.

Now suppose that $i \in S_+ \setminus S_-$. Then $M_+ = \{i\}$ by Proposition 5.4 and we have a relation $(D_i \cdot e)\overline{b_i} = \sum_{j \in M_-} (-D_j \cdot e)\overline{b_j}$ given by $e \in \mathbb{L}$. This implies that $\overline{b_i}$ is contained in the cone σ_{M_-} of Σ_+ , and the definition of ξ_i^+ implies that $D_j \cdot \xi_i^+ = 0$ for all $j \in M_0$. Thus $\xi_i^+|_W = 0$. The case where $i \in S_- \setminus S_+$ is similar. \Box

5.3. Global extended Kähler moduli

Our next goal is to describe a global 'moduli space' $\widetilde{\mathcal{M}}$ and a flat connection over $\widetilde{\mathcal{M}}$, together with two neighbourhoods in $\widetilde{\mathcal{M}}$ such that the restriction of the flat connection to one of the neighbourhoods (respectively to the other neighbourhood) is isomorphic to the equivariant quantum connection for X_+ (respectively for X_-). Thus the equivariant quantum connections for X_+ and X_- can be analytically continued to each other. Roughly speaking, the space $\widetilde{\mathcal{M}}$ will be a covering of a neighbourhood of a certain curve in the *secondary toric variety* for X_{\pm} ; in this section we introduce notation for and local co-ordinates on this secondary toric variety.

The wall and chamber structure of $\mathbb{L}^{\vee} \otimes \mathbb{R}$ described in §5.1 defines a fan in $\mathbb{L}^{\vee} \otimes \mathbb{R}$, called the *secondary fan* or *Gelfand–Kapranov–Zelevinsky (GKZ) fan*. The toric variety associated to the GKZ fan is called the secondary toric variety. We consider the subfan of the GKZ fan consisting of the cones $\overline{C_+}$, $\overline{C_-}$ and their faces, and consider the toric variety \mathcal{M} associated to this fan. (Thus \mathcal{M} is an open subset of the secondary toric variety.) In the context of mirror symmetry, \mathcal{M} arises as the moduli space of Landau–Ginzburg models mirror to X_{\pm} . It contains the torus fixed points P_+ and P_- associated to the cones
C_+ and C_- , which are called the *large radius limit points* for X_+ and X_- . More precisely, because we want to impose only very weak convergence hypotheses on the equivariant quantum products for X_{\pm} , we restrict our attention to the formal neighbourhood of the torus-invariant curve $\mathcal{C} \subset \mathcal{M}$ connecting P_+ and P_- : \mathcal{C} is the closed toric subvariety associated to the cone $\overline{C_W} = W \cap \overline{C_+} = W \cap \overline{C_-}$.

Our secondary toric variety \mathcal{M} is covered by two open charts

$$\operatorname{Spec} \mathbb{C}[C_+^{\vee} \cap \mathbb{L}] \quad \text{and} \quad \operatorname{Spec} \mathbb{C}[C_-^{\vee} \cap \mathbb{L}]$$
 (5.4)

that are glued along Spec $\mathbb{C}[C_W^{\vee} \cap \mathbb{L}]$. Since the cones C_{\pm} are not necessarily simplicial, \mathcal{M} is in general singular. For our purpose, it is convenient to use a lattice structure different from \mathbb{L} and to work with a smooth cover \mathcal{M}_{reg} of \mathcal{M} . We will define the cover \mathcal{M}_{reg} by choosing suitable co-ordinates. As in §4.6, consider the subsets $\mathbb{K}_{\pm} \subset \mathbb{L} \otimes \mathbb{Q}$:

$$\mathbb{K}_{\pm} := \left\{ f \in \mathbb{L} \otimes \mathbb{Q} : \left\{ i \in \{1, 2, \dots, m\} : D_i \cdot f \in \mathbb{Z} \right\} \in \mathcal{A}_{\pm} \right\}$$

and define $\widetilde{\mathbb{L}}_+$ (respectively $\widetilde{\mathbb{L}}_-$) to be the \mathbb{Z} -submodule of $\mathbb{L} \otimes \mathbb{Q}$ generated by \mathbb{K}_+ (respectively by \mathbb{K}_-). Note that $\widetilde{\mathbb{L}}_+$ and $\widetilde{\mathbb{L}}_-$ are free (because they are submodules of $\mathbb{L} \otimes \mathbb{Q}$, which is torsion free) of rank equal to the rank of \mathbb{L} ; they are overlattices of \mathbb{L} .

Lemma 5.8. Set $\widetilde{\mathbb{L}}_{\pm}^{\vee} = \operatorname{Hom}(\widetilde{\mathbb{L}}_{\pm}, \mathbb{Z}) \subset \mathbb{L}^{\vee}$. We have $D_j \in \widetilde{\mathbb{L}}_{\pm}^{\vee}$ if $j \in S_{\pm}$. The decomposition (5.1) of $\mathbb{L}^{\vee} \otimes \mathbb{R}$ is compatible with the integral lattice $\widetilde{\mathbb{L}}_{\pm}^{\vee}$: one has

$$\widetilde{\mathbb{L}}_{\pm}^{\vee} = \left(H^2(X_{\pm}; \mathbb{R}) \cap \widetilde{\mathbb{L}}_{\pm}^{\vee} \right) \oplus \bigoplus_{j \in S_{\pm}} \mathbb{Z}D_j$$
(5.5)

where we regard $H_2(X_{\pm}; \mathbb{R})$ as a subspace of $\mathbb{L}^{\vee} \otimes \mathbb{R}$ via the isomorphism $H^2(X_{\pm}; \mathbb{R}) \cong \bigcap_{j \in S_{\pm}} \operatorname{Ker}(\xi_j^{\pm})$. The lattices $\widetilde{\mathbb{L}}_+^{\vee}$ and $\widetilde{\mathbb{L}}_-^{\vee}$ are compatible along the wall; one has (see equation (5.2)):

$$W \cap \widetilde{\mathbb{L}}_{+}^{\vee} = W \cap \widetilde{\mathbb{L}}_{-}^{\vee} = (W' \cap \widetilde{\mathbb{L}}_{\pm}^{\vee}) \oplus \bigoplus_{j \in S_0} \mathbb{Z}D_j.$$
(5.6)

Proof. Equation (5.5) holds for both X_+ and X_- and we omit the subscript \pm in what follows. Since every element in \mathcal{A} contains S, we have $D_j \cdot f \in \mathbb{Z}$ for all $j \in S$ and $f \in \mathbb{K}$. This shows that $D_j \in \widetilde{\mathbb{L}}^{\vee}$ for $j \in S$. Thus $\widetilde{\mathbb{L}}^{\vee} \supset (H^2(X; \mathbb{R}) \cap \widetilde{\mathbb{L}}^{\vee}) \oplus \bigoplus_{j \in S} \mathbb{Z}D_j$. Conversely, for $v \in \widetilde{\mathbb{L}}^{\vee}$, one has $v \cdot \xi_i \in \mathbb{Z}$ for all $i \in S$ because $\xi_i \in \mathbb{K}$. Then $w = v - \sum_{i \in S} (v \cdot \xi_i) D_i$ lies in $\bigcap_{j \in S} \operatorname{Ker}(\xi_j) \cap \widetilde{\mathbb{L}}^{\vee}$ and $v = w + \sum_{i \in S} (v \cdot \xi_i) D_i$.

Next we prove (5.6). First we claim that for every element $f \in \mathbb{K}_+ \setminus \mathbb{K}_-$, there exists $\alpha \in \mathbb{Q}$ such that $f + \alpha e \in \mathbb{K}_-$. This follows easily from the definition of \mathbb{K}_\pm and Lemma 5.2. It follows from the claim that for any $f \in \widetilde{\mathbb{L}}_+$, there exists $\alpha \in \mathbb{Q}$ such that $f + \alpha e \in \widetilde{\mathbb{L}}_-$. Suppose that $v \in W \cap \widetilde{\mathbb{L}}_-^{\vee}$. For any $f \in \widetilde{\mathbb{L}}_+$, taking $\alpha \in \mathbb{Q}$ as above, one has $v \cdot f = v \cdot (f + \alpha e) \in \mathbb{Z}$. Therefore $v \in W \cap \widetilde{\mathbb{L}}_+^{\vee}$. This shows that $W \cap \widetilde{\mathbb{L}}_-^{\vee} \subset W \cap \widetilde{\mathbb{L}}_+^{\vee}$.

The reverse inclusion follows similarly. The second equality in (5.6) follows from (5.5) and Proposition 5.7. \Box

Remark 5.9. We have $H_2(X_{\pm}; \mathbb{R}) \cap \widetilde{\mathbb{L}}_{\pm} = H_2(|X_{\pm}|; \mathbb{Z}).$

Set $\ell_{\pm} = \dim H^2(X_{\pm}; \mathbb{R}) = r - \sharp(S_{\pm})$ and $\ell = \dim W' = r - 1 - \sharp(S_0)$. We have $\ell \leq \min\{\ell_+, \ell_-\}$. With cases as in Proposition 5.4, we have:

(I) $\ell_{+} = \ell_{-} = \ell + 1;$ (II-i) $\ell_{+} = \ell + 1, \ell_{-} = \ell;$ (II-ii) $\ell_{-} = \ell + 1, \ell_{+} = \ell;$ (III) $\ell_{+} = \ell_{-} = \ell.$

Using Lemma 5.8, we can $choose^{10}$ integral bases

$$\{p_1^+, \dots, p_{\ell_+}^+\} \cup \{D_j : j \in S_+\} \subset \widetilde{\mathbb{L}}_+^{\vee}$$

$$\{p_1^-, \dots, p_{\ell_-}^-\} \cup \{D_j : j \in S_-\} \subset \widetilde{\mathbb{L}}_-^{\vee}$$

$$(5.7)$$

of $\widetilde{\mathbb{L}}_{\pm}^{\vee}$ such that

- $p_1^+, \ldots, p_{\ell_+}^+$ lie in the nef cone $\overline{C'_+} \subset H^2(X_+; \mathbb{R});$
- p_1^-, \ldots, p_ℓ^- lie in the nef cone $\overline{C'_-} \subset H^2(X_-; \mathbb{R});$
- $p_i^+ = p_i^- \in \overline{C'_W}$ for $i = 1, \dots, \ell$.

These bases give co-ordinates on the toric charts (5.4). For $d \in \mathbb{L}$, we write y^d for the corresponding element in the group ring $\mathbb{C}[\mathbb{L}]$. The homomorphisms

$$\mathbb{C}[C^{\vee}_{+} \cap \mathbb{L}] \hookrightarrow \mathbb{C}[y_{1}, \dots, y_{\ell_{+}}, \{x_{j} : j \in S_{+}\}], \qquad \mathsf{y}^{d} \mapsto \prod_{i=1}^{\ell_{+}} y_{i}^{p_{i}^{+} \cdot d} \cdot \prod_{j \in S_{+}} x_{j}^{D_{j} \cdot d}$$
$$\mathbb{C}[C^{\vee}_{-} \cap \mathbb{L}] \hookrightarrow \mathbb{C}[\tilde{y}_{1}, \dots, \tilde{y}_{\ell_{-}}, \{\tilde{x}_{j} : j \in S_{-}\}], \qquad \mathsf{y}^{d} \mapsto \prod_{i=1}^{\ell_{-}} \tilde{y}_{i}^{p_{i}^{-} \cdot d} \cdot \prod_{j \in S_{-}} \tilde{x}_{j}^{D_{j} \cdot d}$$

define the two smooth co-ordinate charts

$$(y_i, x_j : 1 \le i \le \ell_+, j \in S_+)$$
 and $(\tilde{y}_i, \tilde{x}_j : 1 \le i \le \ell_-, j \in S_-)$

which are resolutions of (respectively) $\operatorname{Spec} \mathbb{C}[C_+^{\vee} \cap \mathbb{L}]$ and $\operatorname{Spec} \mathbb{C}[C_-^{\vee} \cap \mathbb{L}]$. We reorder the bases (5.7)

$$\{p_1^+, \dots, p_{\ell_+}^+\} \cup \{D_j : j \in S_+\} = \{\mathsf{p}_1^+, \dots, \mathsf{p}_{r-1}^+, \mathsf{p}_r^+\}$$

¹⁰ This can be seen from the fact that given a lattice L and a full-dimensional cone C in $L \otimes \mathbb{R}$, we can choose a basis of L that consists of elements of C.

T. Coates et al. / Advances in Mathematics 329 (2018) 1002-1087

$$\{p_1^-, \dots, p_{\ell_-}^-\} \cup \{D_j : j \in S_-\} = \{\mathbf{p}_1^-, \dots, \mathbf{p}_{r-1}^-, \mathbf{p}_r^-\}$$

in such a way that $\mathbf{p}_i^+ = \mathbf{p}_i^- \in W$ for $i = 1, \ldots, r-1$ and \mathbf{p}_r^{\pm} is the unique vector (in each basis) that does not lie on the wall W. Let

$$\{y_i, x_j : 1 \le i \le \ell_+, j \in S_+\} = \{y_1, \dots, y_r\}$$
$$\{\tilde{y}_i, \tilde{x}_j : 1 \le i \le \ell_-, j \in S_-\} = \{\tilde{y}_1, \dots, \tilde{y}_r\}$$

be the corresponding reordering of the co-ordinates. Then the change of co-ordinates is of the form:

$$\tilde{\mathbf{y}}_i = \begin{cases} \mathbf{y}_i \mathbf{y}_r^{c_i} & 1 \le i \le r-1\\ \mathbf{y}_r^{-c} & i = r \end{cases}$$
(5.8)

for some $c \in \mathbb{Q}_{>0}$ and $c_i \in \mathbb{Q}$. The numbers c_i , c here arise from the transition matrix of the two bases (5.7). We find a common denominator for c, c_i and write c = A/B and $c_i = A_i/B$, $1 \le i \le r - 1$ for some $A, B \in \mathbb{Z}_{>0}$ and $A_i \in \mathbb{Z}$. Then $y_r^{1/B} = \tilde{y}_r^{-1/A}$. The smooth manifold \mathcal{M}_{reg} is defined by gluing the two charts

$$U_{+} = \operatorname{Spec} \mathbb{C}[\mathsf{y}_{1}, \dots, \mathsf{y}_{r-1}, \mathsf{y}_{r}^{1/B}] \quad \text{and} \quad U_{-} = \operatorname{Spec} \mathbb{C}[\tilde{\mathsf{y}}_{1}, \dots, \tilde{\mathsf{y}}_{r-1}, \tilde{\mathsf{y}}_{r}^{1/A}]$$

via the change of variables (5.8). The large radius limit points $P_+ \in U_+$ and $P_- \in U_$ are given respectively by $y_1 = \cdots = y_r = 0$ and $\tilde{y}_1 = \cdots = \tilde{y}_r = 0$. Note that the last variables y_r , \tilde{y}_r correspond to the direction of $e \in \mathbb{L}$: one has $y^e = y_r^{p_r^+ \cdot e} = \tilde{y}_r^{p_r^- \cdot e}$.

The torus-invariant rational curve $C_{\text{reg}} \subset \mathcal{M}_{\text{reg}}$ associated to C_W is given by $y_1 = \cdots = y_{r-1} = 0$ on U_+ and by $\tilde{y}_1 = \cdots = \tilde{y}_{r-1} = 0$ on U_- . Let $\widehat{\mathcal{M}}_{\text{reg}}$ be the formal neighbourhood of \mathcal{C}_{reg} in \mathcal{M}_{reg} . Since the global quantum connection is an analytic object, we need to work with a suitable analytification of $\widehat{\mathcal{M}}_{\text{reg}}$: we include analytic functions in the last variable y_r in the structure sheaf and use the analytic topology on $\mathcal{C}_{\text{reg}} \cong \mathbb{P}^1$. The underlying topological space of $\widehat{\mathcal{M}}_{\text{reg}}$ is therefore $\mathbb{P}^{1,\text{an}}$; $\widehat{\mathcal{M}}_{\text{reg}}$ is covered by two charts \widehat{U}_+ and \widehat{U}_- with structure sheaves:

$$\mathcal{O}_{\widehat{U}_{+}} = \mathcal{O}_{\mathbb{C}_{+}}^{\mathrm{an}} \llbracket \mathsf{y}_{1}, \dots, \mathsf{y}_{r-1} \rrbracket \quad \text{and} \quad \mathcal{O}_{\widehat{U}_{-}} = \mathcal{O}_{\mathbb{C}_{-}}^{\mathrm{an}} \llbracket \widetilde{\mathsf{y}}_{1}, \dots, \widetilde{\mathsf{y}}_{r-1} \rrbracket \tag{5.9}$$

where \mathbb{C}_+ and \mathbb{C}_- denote the complex plane with co-ordinates $y_r^{1/B}$ and $\tilde{y}_r^{1/A}$ respectively and the superscript "an" means analytic (space or structure sheaf). In other words, we regard \hat{U}_+ , \hat{U}_- , and $\widehat{\mathcal{M}}_{\text{reg}}$ as ringed spaces respectively on \mathbb{C}_+ , \mathbb{C}_- and $\mathbb{P}^{1,\text{an}}$.

The same construction works over an arbitrary \mathbb{C} -algebra R. We define $\widehat{\mathcal{M}}_{reg}(R)$ by replacing the structure sheaves in (5.9) with $(\mathcal{O}_{\mathbb{C}_+}^{an} \otimes R)[\![y_1, \ldots, y_{r-1}]\!]$ and $(\mathcal{O}_{\mathbb{C}_-}^{an} \otimes R)[\![\tilde{y}_1, \ldots, \tilde{y}_{r-1}]\!]$. In the equivariant theory, we use $R = R_T[z] = H_T^{\bullet}(pt) \otimes \mathbb{C}[z]$ for the ground ring. The global equivariant quantum connection will be defined over $R_T[z]$ and on (a formal thickening of) a simply-connected open subset of $\mathbb{P}^{1,an}$ containing P_+ and P_- .

Remark 5.10. Taking an overlattice $\widetilde{\mathbb{L}}_{\pm}$ of \mathbb{L} corresponds to taking a finite cover of \mathcal{M} . This is necessary because the power series defining the I-function (see §5.4) is indexed by elements in \mathbb{L}_{\pm} . If one takes into consideration Galois symmetry [49] of the quantum connection, one can see that the quantum connection (near P_+) descends to the secondary toric variety with respect to the original lattice \mathbb{L} .

5.4. The I-function

Recall Givental's Lagrangian cone introduced in Definition 2.2. We consider the Givental cone $\mathcal{L}_{X_{\omega}}$ associated to the toric Deligne–Mumford stack X_{ω} . Under the decomposition (4.10) of $\mathbb{L} \otimes \mathbb{R}$, we decompose $d \in \mathbb{L} \otimes \mathbb{R}$ as:

$$d = \overline{d} + \sum_{j \in S_{\pm}} (D_j \cdot d) \xi_j$$

where \overline{d} is the $H_2(X_{\omega}; \mathbb{R})$ -component of d. Define the $H^{\bullet}_{\mathrm{CR},T}(X_{\omega})$ -valued hypergeometric series $I_{\omega}^{\text{temp}}(\sigma, x, z) \in H^{\bullet}_{\text{CR}, T}(X_{\omega}) \otimes_{R_T} R_T((z^{-1}))[[Q, \sigma, x]]$ by

$$I_{\omega}^{\text{temp}}(\sigma, x, z) := z e^{\sigma/z} \sum_{d \in \mathbb{X}} e^{\sigma \cdot \overline{d}} Q^{\overline{d}} \prod_{j \in S} x_j^{D_j \cdot d} \left(\prod_{j=1}^m \frac{\prod_{a:\langle a \rangle = \langle D_j \cdot d \rangle, a \leq 0} (u_j + az)}{\prod_{a:\langle a \rangle = \langle D_j \cdot d \rangle, a \leq D_j \cdot d} (u_j + az)} \right) \mathbf{1}_{[-d]}$$

where K is introduced in §4.6, $x = (x_j : j \in S)$ and $\sigma \in H^2_T(X_\omega)$ are variables, and [-d]is the equivalence class of -d in \mathbb{K}/\mathbb{L} (recall from §4.8 that \mathbb{K}/\mathbb{L} parametrizes inertia components). The subscript 'temp' reflects the fact that we are just about to change notation, by specializing certain parameters, and so this notation for the I-function is only temporary. One can see that the summand of I^{temp}_ω corresponding to $d\in\mathbb{K}$ vanishes unless $d \in C_{\omega}^{\vee}$. Therefore the summation ranges over all $d \in \mathbb{K}$ such that \overline{d} lies in the Mori cone $NE(X_{\omega}) = C'_{\omega}^{\vee}$ and $D_j \cdot d \ge 0$ for all $j \in S$. The Mirror Theorem for toric Deligne–Mumford stacks can be stated as follows:

Theorem 5.11 ([26,29]). $I_{\omega}^{\text{temp}}(\sigma, x, -z)$ is an $S_T[[Q, \sigma, x]]$ -valued point on $\mathcal{L}_{X_{\omega}}$.

We adapt the above theorem to the situation of toric wall-crossing. Let I_{\pm}^{temp} denote the *I*-function of X_{\pm} . We introduce a variant I_{\pm} of the *I*-function which gives a cohomology-valued function on a neighbourhood of P_{\pm} in \mathcal{M}_{reg} . The *I*-function I_{\pm} is obtained from I_{\pm}^{temp} by the following specialization:

- Q = 1;
- for I_+ , $\sigma = \sigma_+ := \theta_+ (\sum_{i=1}^r \mathsf{p}_i^+ \log \mathsf{y}_i) + c_0(\lambda);$ for I_- , $\sigma = \sigma_- := \theta_- (\sum_{i=1}^r \mathsf{p}_i^- \log \tilde{\mathsf{y}}_i) + c_0(\lambda);$

where $\theta_{\pm} \colon \mathbb{L}^{\vee} \otimes \mathbb{C} \to H^2_T(X_{\pm}; \mathbb{C})$ are the maps introduced in (4.8) and $c_0(\lambda) = \lambda_1 +$ $\cdots + \lambda_m$. Note that we have

T. Coates et al. / Advances in Mathematics 329 (2018) 1002-1087

$$\sigma_{+} = \sum_{i=1}^{\ell_{+}} \theta_{+}(p_{i}^{+}) \log y_{i} - \sum_{j \in S_{+}} \lambda_{j} \log x_{j} + c_{0}(\lambda)$$
(5.10)

since $\theta_+(D_j) = -\lambda_j$ for $j \in S_+$. More explicitly, one can write I_+ as:

$$I_{+}(\mathbf{y},z) := ze^{\sigma_{+}/z} \sum_{d \in \mathbb{K}_{+}} \mathbf{y}^{d} \left(\prod_{j=1}^{m} \frac{\prod_{a:\langle a \rangle = \langle D_{j} \cdot d \rangle, a \leq 0} (u_{j} + az)}{\prod_{a:\langle a \rangle = \langle D_{j} \cdot d \rangle, a \leq D_{j} \cdot d} (u_{j} + az)} \right) \mathbf{1}_{[-d]}$$
(5.11)

where recall that $(y_1, \ldots, y_r) = (y_i, x_j : 1 \le i \le \ell_+, j \in S_+)$ are co-ordinates on $\widehat{U}_+ \subset \widehat{\mathcal{M}}_{\text{reg}}$ and that

$$y^{d} = y_{1}^{p_{1}^{+} \cdot d} \cdots y_{r}^{p_{r}^{+} \cdot d} = \prod_{i=1}^{\ell_{+}} y_{i}^{p_{i}^{+} \cdot d} \prod_{j \in S_{+}} x_{j}^{D_{j} \cdot d}.$$

The *I*-function I_+ belongs to the space:

$$I_+ \in H^{\bullet}_{\mathrm{CR},T}(X_+) \otimes_{R_T} R_T[\log \mathsf{y}_1, \dots, \log \mathsf{y}_r]((z^{-1}))[[\mathsf{y}_1, \dots, \mathsf{y}_r]].$$

The series $e^{-\sigma_+/z}I_+(y,z)$ is homogeneous of degree two with respect to the (age-shifted) grading on $H^{\bullet}_{\text{CR},T}(X_+)$ and the degrees for variables given by:

deg
$$z = 2$$
 and $\sum_{i=1}^{r} (\deg y_i) \mathbf{p}_i^+ = 2 \sum_{i=1}^{m} D_i$ (5.12)

Note that deg $y_r = 0$ because $\sum_{i=1}^m D_i \in W$.

Remark 5.12. The extra factor $e^{c_0(\lambda)/z}$ in the *I*-function makes the mirror map compatible with Euler vector fields.

We now show that $I_+(y,z)$ is analytic in the last variable y_r , so that it defines an analytic function on $\widehat{\mathcal{M}}_{reg}$.

Lemma 5.13. Expand the I-function as

$$I_{+}(\mathbf{y},z) = ze^{\sigma_{+}/z} \sum_{k_{1}=0}^{\infty} \cdots \sum_{k_{r-1}=0}^{\infty} \sum_{i=0}^{N} \mathbf{y}_{1}^{k_{1}} \cdots \mathbf{y}_{r-1}^{k_{r-1}} I_{+;k_{1},\dots,k_{r-1}}^{i}(\mathbf{y}_{r},z)\phi_{i}$$

using an R_T -basis $\{\phi_i\}$ of $H^{\bullet}_{CR}(X_+)$. Let \mathfrak{c} be the rational number¹¹

$$\mathfrak{c} = \prod_{j:D_j \cdot e \neq 0} (D_j \cdot e)^{D_j \cdot e}$$
(5.13)

¹¹ This is the so-called *conifold point*.

and let \mathcal{U}_+ be the universal covering of the space $\{\mathbf{y}_r \in \mathbb{C} : \mathbf{y}_r^{\mathbf{p}_r^+ \cdot e} \neq \mathbf{c}\}$. Each coefficient $I^i_{+;k_1,\ldots,k_{r-1}}(\mathbf{y}_r,z) \in R_T[\![z^{-1}]\!][\![\mathbf{y}_r]\!]$ converges to give an analytic function on the region

$$\left\{ (\lambda_1, \dots, \lambda_m, z, \mathsf{y}_r) \in \mathbb{C}^m \times \mathbb{C} \times \mathcal{U}_+ : |\lambda_i| < \epsilon |z| \right\}$$
(5.14)

for some $\epsilon > 0$ that is independent of i, k_1, \ldots, k_{r-1} .

Proof. In §6.2 below we will compute the analytic continuation of the *I*-function after restriction to a fixed point. There we introduce a function called the *H*-function, which is related to the *I*-function by a constant linear transformation, and give a Mellin–Barnes integral representation for the *H*-function. This integral representation makes clear that $e^{-\sigma_+/(2\pi i)}H$ can be analytically continued to \mathcal{U}_+ . The linear transformation between the *I*-function and the *H*-function involves the factor $z^{-\mu}\widehat{\Gamma}_X$ (see equation (6.8)) and this factor has poles at non-zero $(\lambda_1/z, \ldots, \lambda_m/z)$. Therefore we obtain the analyticity of the *I*-function on a region of the form (5.14). \Box

An entirely parallel statement holds for $I_{-}(y, z)$.

5.5. Global equivariant quantum connection

In this section we use the I-function I_+ to construct a global quantum connection on the universal cover

$$\widetilde{\mathcal{M}}_{+} := \left(\left(\widehat{U}_{+} \setminus \{ \mathsf{y}^{e} = \mathfrak{c} \} \right) / \boldsymbol{\mu}_{B} \right)^{\widehat{}}$$

where \widehat{U}_+ is the open chart (5.9) of $\widehat{\mathcal{M}}_{reg}$ and $y^e = y_r^{p_r^+, e}$ is a function on \widehat{U}_+ . The action of μ_B on \widehat{U}_+ is by deck transformations of $y_r^{1/B} \mapsto y_r$. As in Lemma 5.13, we denote by \mathcal{U}_+ the universal cover of $\{y_r \in \mathbb{C} : y_r^{p_r^+, e} \neq \mathfrak{c}\}$. The space \mathcal{U}_+ is the underlying topological space of $\widetilde{\mathcal{M}}_+$, and $\widetilde{\mathcal{M}}_+$ is a formal thickening of \mathcal{U}_+ ; more precisely, $\widetilde{\mathcal{M}}_+$ is the ringed space $(\mathcal{U}_+, \mathcal{O})$ with structure sheaf $\mathcal{O} = \mathcal{O}_{\mathcal{U}_+}^{an} [y_1, \ldots, y_{r-1}]$. In a neighbourhood of P_+ , the global quantum connection that we will construct can be identified with the equivariant quantum connection of X_+ . The main result in this section is:

Theorem 5.14. There exist the following data:

- an open subset U[°]₊ ⊂ U⁺₊ such that P⁺₊ ∈ U[°]₊ and that the complement U⁺₊ \U[°]₊ is a discrete set; we write M[°]₊ = M[°]₊ |_{U[°]₊};
- a trivial $H^{\bullet}_{\operatorname{CR},T}(X_+)$ -bundle \mathbf{F}^+ over $\widetilde{\mathcal{M}}^{\circ}_+(R_T[z])$:

$$\mathbf{F}^+ = H^{\bullet}_{\mathrm{CR},T}(X_+) \otimes_{R_T} (\mathcal{O}_{\mathcal{U}^{\circ}_+} \otimes R_T[z]) \llbracket \mathsf{y}_1, \dots, \mathsf{y}_{r-1} \rrbracket;$$

• a flat connection $\nabla^+ = d + z^{-1} \mathbf{A}^+(\mathbf{y})$ on \mathbf{F}^+ of the form:

$$\mathbf{A}^{+}(\mathbf{y}) = \sum_{i=1}^{\ell_{+}} B_{i}(\mathbf{y}) \frac{dy_{i}}{y_{i}} + \sum_{j \in S_{+}} C_{j}(\mathbf{y}) dx_{j} - \sum_{j \in S_{+}} \lambda_{j} \frac{dx_{j}}{x_{j}}$$

with $B_i(\mathbf{y}), C_j(\mathbf{y}) \in \operatorname{End}(H^{\bullet}_{\operatorname{CR},T}(X_+)) \otimes_{R_T} (\mathcal{O}_{\mathcal{U}^{\circ}_+} \otimes R_T) \llbracket \mathbf{y}_1, \dots, \mathbf{y}_{r-1} \rrbracket;$

• a vector field \mathbf{E}^+ on $\widetilde{\mathcal{M}}_+(R_T)$, called the Euler vector field, defined by:

$$\mathbf{E}^{+} = \sum_{i=1}^{m} \lambda_{i} \frac{\partial}{\partial \lambda_{i}} + \sum_{i=1}^{r} \frac{1}{2} (\deg \mathsf{y}_{i}) \mathsf{y}_{i} \frac{\partial}{\partial \mathsf{y}_{i}};$$

• a mirror map $\tau_+ : \widetilde{\mathcal{M}}_+(R_T) \to H^{\bullet}_{\operatorname{CR},T}(X_+)$ of the form:

$$\tau_{+} = \sigma_{+} + \tilde{\tau}_{+} \qquad \tilde{\tau}_{+} \in H^{\bullet}_{\operatorname{CR},T}(X_{+}) \otimes_{R_{T}} (\mathcal{O}_{\mathcal{U}^{\circ}_{+}} \otimes R_{T}) \llbracket \mathsf{y}_{1}, \dots, \mathsf{y}_{r-1} \rrbracket$$
$$\tilde{\tau}_{+}|_{\mathsf{y}_{1}=\cdots=\mathsf{y}_{r}=0} = 0$$

such that ∇^+ equals the pull-back $\tau^*_+ \nabla^+$ of the equivariant quantum connection ∇^+ of X_+ by τ_+ , that is:

$$B_{i}(\mathbf{y}) = \sum_{k=0}^{N} \frac{\partial \tau_{+}^{k}(\mathbf{y})}{\partial \log y_{i}} (\phi_{k} \star_{\tau_{+}(\mathbf{y})}) \qquad 1 \le i \le \ell_{+}$$
$$C_{j}(\mathbf{y}) = \sum_{k=0}^{N} \frac{\partial \tilde{\tau}_{+}^{k}(\mathbf{y})}{\partial x_{j}} (\phi_{k} \star_{\tau_{+}(\mathbf{y})}) \qquad j \in S_{+}$$

and that the push-forward of \mathbf{E}^+ by τ_+ is the Euler vector field \mathcal{E}^+ for X_+ defined in equation (2.4). Moreover, there exists a global section $\Upsilon_0^+(\mathbf{y}, z)$ of \mathbf{F}^+ such that

$$I_{+}(\mathbf{y}, z) = zL_{+}(\tau_{+}(\mathbf{y}), z)^{-1}\Upsilon_{0}^{+}(\mathbf{y}, z)$$

where $L_{+}(\tau, z)$ is the fundamental solution for the quantum connection of X_{+} in Proposition 2.4.

Remark 5.15. Here the Novikov variables Q in the quantum product and the fundamental solution have been specialized to 1: see §3.2.

Remark 5.16. An entirely analogous result holds for X_{-} .

Remark 5.17. The data in Theorem 5.14 satisfy some compatibility equations. The connection matrices B_i , C_i are self-adjoint with respect to the equivariant orbifold Poincaré pairing (\cdot, \cdot) . Furthermore the grading operator $\mathbf{Gr}^+ = z \frac{\partial}{\partial z} + \mathbf{E}^+ + \mu^+$ on \mathbf{F}^+ (where μ^+ is the grading operator on $H^{\bullet}_{\mathbf{CR},T}(X_+)$ defined in equation (2.4)) satisfies

 $[\mathbf{Gr}^+, \mathbf{\nabla}_v^+] = \mathbf{\nabla}_{[\mathbf{E}^+, v]}^+$ for any vector field v. These properties are inherited from the quantum connection.

Remark 5.18 ([49, Remark 3.5]). By construction, the mirror map τ_+ here depends on the extended vectors b_j , $j \in S_+$, in the extended stacky fan. If we add sufficiently many extended vectors, we can make τ_+ submersive near P_+ and Theorem 5.14 gives an analytic continuation of the big quantum cohomology. In fact we have

$$\tau(\mathbf{y}) = c_0(\lambda) + \sum_{j \in S_+ \setminus S_0} \lambda_j \log x_j + \sum_{i=1}^{\ell_+} \theta(p_i^+) \log y_i + \sum_{j \in S_+} \alpha_j x_j + \text{higher order terms.}$$

Here $\alpha_j = \prod_{i \in I_j} u_i^{n_{ij}} \mathbf{1}_{[-\xi_j]}$, where $\xi_j \in \mathbb{K}_+$ is given in (4.9), $I_j \subset \{1, \ldots, m\} \setminus S_+$ is such that σ_{I_j} contains $\overline{b_j}$, and $\overline{b_j} = \sum_{i \in I_j} (n_{ij} + \epsilon_{ij}) \overline{b_i}$ with $\epsilon_{ij} \in [0, 1)$ and $n_{ij} \in \mathbb{Z}_{\geq 0}$. Note that $\mathbf{1}_{[-\xi_j]}$ corresponds to the Box element $b_j - \sum_{i \in I_j} n_{ij} b_i \in \text{Box}(X_+)$.

Remark 5.19. The logarithmic singularity of ∇^+ along $\prod_{j \in S_+} x_j = 0$ is not very important: this can be eliminated by shifting the mirror map τ by $\sum_{j \in S_+} \lambda_j \log x_j$; see (5.10).

The rest of this section is devoted to the proof of Theorem 5.14. First we recall how to compute the quantum connection of X_+ using the *I*-function (cf. [30]). By the Mirror Theorem 5.11, $I_+^{\text{temp}}(\sigma, x, -z)$ is a point on the Givental cone $\mathcal{L}_+ := \mathcal{L}_{X_+}$ for X_+ . Recall from Remark 2.5 that the cone \mathcal{L}_+ is ruled by its tangent spaces (multiplied by z):

$$\mathcal{L}_{+} = \bigcup_{\tau \in H^{\bullet}_{\mathbf{CR}, \tau}(X_{+})} z L_{+}(\tau, -z)^{-1} \mathcal{H}_{+}.$$

This implies that one has:

$$I_{+}^{\text{temp}}(\sigma, x, z) = zL_{+}(\tau, z)^{-1}\Upsilon_{0}^{+}$$

for some $\tau = \tau(\sigma, x) \in H^{\bullet}_{CR,T}(X_+) \otimes_{R_T} R_T[\![Q, \sigma, x]\!]$ and $\Upsilon^+_0 \in \mathcal{H}_+[\![\sigma, x]\!] = H^{\bullet}_{CR,T}(X_+) \otimes_{R_T} R_T[z][\![Q, \sigma, x]\!]$. The map $(\sigma, x) \mapsto \tau(\sigma, x)$ is called the *mirror map*: this will be determined below. In Lemma 5.22 we will construct differential operators $P_i = P_i(z\partial), i = 0, \ldots, N$ which depend polynomially on z and on the vector fields $z\partial_v$, $v \in H^2_T(X_+)$, and $z\partial_{x_j}, j \in S_+$, and which satisfy:

- $\phi_i = z^{-1} P_i I_+^{\text{temp}}|_{Q=\sigma=x=0}, \ 0 \le i \le N$, are independent of z;
- $\{\phi_i : 0 \le i \le N\}$ form a basis of $H^*_{\mathrm{CR},T}(X_+)$ over R_T ;
- $P_0 = 1$.

Then:

$$\begin{bmatrix} z^{-1}P_0I_+^{\text{temp}} & \cdots & z^{-1}P_NI_+^{\text{temp}} \end{bmatrix} = L_+(\tau, z)^{-1}\begin{bmatrix} | & | & | \\ \Upsilon_0^+ & \cdots & \Upsilon_N^+ \\ | & | \end{bmatrix}$$
(5.15)

for $\Upsilon_i^+ := P_i(z\tau^*\nabla)\Upsilon_0^+$. Here $\tau^*\nabla$ is the pull-back of the quantum connection of X_+ via the mirror map τ , and we used the fact that one has $\partial_v \circ L_+(\tau, z)^{-1} = L_+(\tau, z)^{-1} \circ (\tau^*\nabla)_v$ for any vector field v on (σ, x) -space. Note that:

- $\Upsilon_i^+ \in \mathcal{H}_+[\![\sigma, x]\!]$ does not contain negative powers of z;
- $L_{+}(\tau, z)$ does not contain positive powers of z; and
- $L_+(\tau, z) = \mathrm{id} + O(z^{-1}).$

Thus the right-hand side of (5.15) can be regarded as the *Birkhoff factorization* of the left-hand side (see [68]), when we view both sides as elements in the loop group LGL_{N+1} with z the loop parameter. The properties of P_i listed above ensure that the left-hand side of (5.15) is invertible at $Q = \sigma = x = 0$, and that its Birkhoff factorization can be determined recursively in powers of Q, σ and x (see Lemma 5.23). Thus the *I*-function determines $L_+(\tau, z)^{-1}$ as a function of (σ, x) , via Birkhoff factorization. The mirror map $\tau = \tau(\sigma, x)$ is determined by the asymptotics

$$L_{+}(\tau, z)^{-1} = 1 + \tau z^{-1} + O(z^{-2})$$

and $L_{+}(\tau, z)^{-1}$ determines the pulled-back quantum connection $\tau^* \nabla$.

We perform the above procedure globally on $\widehat{\mathcal{M}}_{\text{reg}}$, using the *I*-function I_+ obtained from I_+^{temp} by the specialization Q = 1, $\sigma = \sigma_+$. It will be convenient to assume the following condition.

Assumption 5.20. The set $\mathbf{N} \cap |\Sigma_+| = \{v \in \mathbf{N} : \overline{v} \in |\Sigma_+|\}$ of lattice points in the support $|\Sigma_+|$ of the fan is generated by $b_j, j = 1, \ldots, m$ as an additive monoid.

Remark 5.21. Assumption 5.20 is harmless: it can be always achieved by adding enough extended vectors to the extended stacky fan and in fact Theorem 5.14 holds without this assumption (see Remark 5.26).

Recall from §4.8 that $H^{\bullet}_{CR,T}(X_+)$ is the direct sum of sectors $H^{\bullet}_T(X^f_+)$, $f \in \mathbb{K}_+/\mathbb{L}$ and recall from §4.3 that each sector $H^{\bullet}_T(X^f_+)$ is generated by divisor classes. Thus we can take an R_T -basis of $H^{\bullet}_{CR,T}(X_+)$ of the form:

$$\phi_{f,i} = F_{f,i}\left(\theta(p_1^+), \dots, \theta(p_{\ell_+}^+)\right) \mathbf{1}_f \qquad f \in \mathbb{K}/\mathbb{L}, \ 1 \le i \le \dim H^{\bullet}(X_+^f)$$

where $F_{f,i}(a_1, \ldots, a_{\ell_+}) \in \mathbb{C}[a_1, \ldots, a_{\ell_+}]$ is a homogeneous polynomial. Recall from §4.8 that elements in \mathbb{K}_+/\mathbb{L} are in one-to-one correspondence with elements in $\text{Box}(X_+)$. Let $v_f \in \text{Box}(X_+)$ be the element corresponding to $f \in \mathbb{K}_+/\mathbb{L}$. By Assumption 5.20, there exist non-negative integers $n_{f,j}, j = 1, \ldots, m$, such that

$$v_f = \sum_{j=1}^m n_{f,j} b_j.$$
 (5.16)

On the other hand, taking a minimal cone σ_f in Σ_+ containing v_f , we can write

$$\overline{v_f} = \sum_{j \notin S_+, \overline{b_j} \in \sigma_f} c_{f,j} \overline{b_j}$$

for some $c_{f,j} \in [0,1)$. We set $c_{f,j} = 0$ if $j \in S_+$ or $b_j \notin \sigma_f$. Then $\sum_{j=1}^m (n_{f,j} - c_{f,j})\overline{b_j} = 0$ and by (4.3), there exists an element $d_f \in \mathbb{L} \otimes \mathbb{Q}$ such that $D_j \cdot d_f = n_{f,j} - c_{f,j}$. By definition of \mathbb{K}_+ , $d_f \in \mathbb{K}_+$ and $[-d_f] = f$ in \mathbb{K}_+/\mathbb{L} by (4.14) and (5.16). Set $D_j = \sum_{a=1}^r \mu_{ja} \mathbf{p}_a^+$ for some $\mu_{ja} \in \mathbb{Z}$. Define differential operators \mathcal{D}_j , Δ_f as

$$\mathcal{D}_{j} := \sum_{a=1}^{r} \mu_{ja} z \mathbf{y}_{a} \frac{\partial}{\partial \mathbf{y}_{a}}$$
$$\Delta_{f} := \mathbf{y}^{-d_{f}} \prod_{j=1}^{m} \prod_{\nu=0}^{n_{f,j}-1} (\mathcal{D}_{j} + \lambda_{j} - \nu z)$$

The following Lemma was proved in [49, Lemma 4.7], in the non-equivariant and compact case. The proof works verbatim here.

Lemma 5.22. Let $F_{f,i}$, $\phi_{f,i}$, Δ_f be as above. Define the differential operator $P_{f,i}^+$ by

$$P_{f,i}^+ := F_{f,i}\left(zy_1\frac{\partial}{\partial y_1}, \dots, zy_{\ell_+}\frac{\partial}{\partial y_{\ell_+}}\right)\Delta_f.$$

Then we have:

$$P_{f,i}^{+}I_{+}(\mathbf{y},z) = ze^{\sigma_{+}/z}(\phi_{f,i} + O(\mathbf{y})).$$

Applying the differential operators $P_{f,i}^+$, $f \in \mathbb{K}_+/\mathbb{L}$, $1 \leq i \leq \dim H^{\bullet}(X_+^f)$, to I_+ , we obtain a matrix of the form:

$$\begin{bmatrix} \cdots & z^{-1} P_{f,i}^+ I_+ & \cdots \\ & & \end{bmatrix} = e^{\sigma_+/z} \mathbb{I}_+(\mathbf{y}, z)$$
(5.17)

where I_+ is regarded as a column vector written in the basis $\{\phi_{f,i}\}$ of $H^{\bullet}_{\mathrm{CR},T}(X_+)$ and $\mathbb{I}_+(\mathbf{y},z) = \mathrm{id} + O(\mathbf{y})$ is a square matrix. We may also view $\mathbb{I}_+(\mathbf{y},z)$ as an $\mathrm{End}(H^{\bullet}_{\mathrm{CR}}(X_+))$ -valued function via the basis $\{\phi_{f,i}\}$. By the homogeneity of $e^{-\sigma_+/z}I_+$ and $P^+_{f,i}$, we find that the endomorphism $\mathbb{I}_+(\mathbf{y},z)$ is homogeneous of degree zero with respect to the degree (5.12) of variables and the grading on $H^{\bullet}_{\mathrm{CR}}(X_+)$, i.e. that:

$$\left(z\frac{\partial}{\partial z} + \mathbf{E}^{+} + \mathrm{ad}(\mu^{+})\right)\mathbb{I}_{+}(\mathbf{y}, z) = 0$$
(5.18)

As in (5.15), we consider the Birkhoff factorization of (5.17). Since $e^{\sigma_+/z} = id + O(z^{-1})$, it suffices to consider the Birkhoff factorization of $\mathbb{I}_+(y, z)$. Set:

$$\gamma(\mathbf{y}_r, z) := \mathbb{I}_+(\mathbf{y}, z) \Big|_{\mathbf{y}_1 = \cdots = \mathbf{y}_{r-1} = 0, \lambda_1 = \cdots = \lambda_m = 0}.$$

By Lemma 5.13, $z \mapsto \gamma(y_r, z)$ is a loop in $\operatorname{End}(H^{\bullet}_{\operatorname{CR}}(X_+))$ that depends analytically on $y_r \in \mathcal{U}_+$. We first consider the Birkhoff factorization of $\gamma(y_r, z)$. Since $\gamma(y_r, z)$ is homogeneous, it is a Laurent polynomial in z and both factors of the Birkhoff factorization $\gamma(y_r, z) = \gamma_-(z)\gamma_+(z)$ are also homogeneous if the factorization exists. Therefore the Birkhoff factorization is equivalent to the *block LU decomposition* of $\gamma(y_r, 1)$:

$$\gamma_{-}(1) = \begin{bmatrix} I_{r_{1}} & & \\ * & I_{r_{2}} & & 0 \\ \vdots & \vdots & \ddots & \\ * & * & \cdots & I_{r_{k}} \end{bmatrix} \qquad \gamma_{+}(1) = \begin{bmatrix} * & * & \cdots & * \\ & * & \cdots & * \\ & & \ddots & \vdots \\ 0 & & & * \end{bmatrix}$$

where each block corresponds to a homogeneous component of $H^{\bullet}_{CR}(X_+)$ and I_r denotes the identity matrix of size r. The block LU decomposition of $\gamma(y_r, 1)$ exists if and only if

$$H = (\gamma(\mathsf{y}_r, 1)H^{\leq p}) \oplus H^{> p}$$

holds for all $p \in \mathbb{Q}$, where $H = H^{\bullet}_{CR}(X_+)$ and $H^{\leq p}$ (resp. $H^{>p}$) denotes the subspace of degree less than or equal to p (resp. greater than p). This is a Zariski open condition for $\gamma(y_r, 1)$. Since $\gamma(y_r = 0, 1) = id$, it follows that $\gamma(y_r, z)$ admits a Birkhoff factorization on the complement \mathcal{U}°_+ of a discrete set in \mathcal{U}_+ . Clearly one has $P_+ \in \mathcal{U}^{\circ}_+$.

Lemma 5.23. Let $\gamma(z) \in LGL_{N+1}(\mathbb{C})$ be a Laurent polynomial loop admitting a Birkhoff factorization $\gamma = \gamma_{-}\gamma_{+}$. Let $\Gamma(s, z) \in End(\mathbb{C}^{N+1}) \otimes \mathbb{C}[z, z^{-1}][s_1, \ldots, s_l]$ be a formal loop such that $\Gamma|_{s=0} = \gamma$. Then $\Gamma(s, z)$ admits a unique Birkhoff factorization of the form

$$\Gamma(s,z) = \Gamma_{-}(s,z)\Gamma_{+}(s,z)$$

such that $\Gamma_{-}(s,z) \in \operatorname{End}(\mathbb{C}^{N+1}) \otimes \mathbb{C}[z^{-1}][\![s_1,\ldots,s_l]\!], \ \Gamma_{-}(s,\infty) = \operatorname{id} and \ \Gamma_{+}(s,z) \in \operatorname{End}(\mathbb{C}^{N+1}) \otimes \mathbb{C}[z][\![s_1,\ldots,s_l]\!].$

Proof. It suffices to show that $\Gamma' = \gamma_{-}^{-1} \Gamma \gamma_{+}^{-1}$ admits a Birkhoff factorization $\Gamma' = \Gamma'_{-} \Gamma'_{+}$. Expanding Γ' and Γ'_{\pm} in power series in s_1, \ldots, s_l , one can determine the coefficients recursively from the equation $\Gamma' = \Gamma'_{-} \Gamma'_{+}$. \Box

Applying the above lemma to $\mathbb{I}_+(y, z)$, we see that $\mathbb{I}_+(y, z)$ with $y_r \in \mathcal{U}_+^\circ$ admits a Birkhoff factorization

$$\mathbb{I}_{+}(\mathbf{y}, z) = \mathbf{L}_{+}(\mathbf{y}, z)^{-1} \Upsilon^{+}(\mathbf{y}, z)$$
(5.19)

where

$$\mathbf{L}_{+}(\mathsf{y}, z) \in \operatorname{End}(H^{\bullet}_{\operatorname{CR}}(X_{+})) \otimes \mathcal{O}_{\mathcal{U}^{\circ}_{+}}[z^{-1}] \llbracket \lambda_{1}, \dots, \lambda_{m}, \mathsf{y}_{1}, \dots, \mathsf{y}_{r-1} \rrbracket,$$

$$\Upsilon^{+}(\mathsf{y}, z) \in \operatorname{End}(H^{\bullet}_{\operatorname{CR}}(X_{+})) \otimes \mathcal{O}_{\mathcal{U}^{\circ}_{+}}[z] \llbracket \lambda_{1}, \dots, \lambda_{m}, \mathsf{y}_{1}, \dots, \mathsf{y}_{r-1} \rrbracket$$

and $\mathbf{L}_{+}(\mathbf{y}, \infty) = \mathrm{id.}$ Using the homogeneity equation (5.18), we find that the Birkhoff factors \mathbf{L}_{+} , Υ^{+} are also homogeneous of degree zero. Also the chosen R_{T} -basis $\{\phi_{f,i}\}$ of $H^{\bullet}_{\mathrm{CR},T}(X_{+})$ defines a splitting $H^{\bullet}_{\mathrm{CR},T}(X_{+}) \cong H^{\bullet}_{\mathrm{CR}}(X_{+}) \otimes R_{T}$, and via this splitting, one may naturally regard \mathbf{L}_{+} , Υ^{+} as $\mathrm{End}(H^{\bullet}_{\mathrm{CR},T}(X_{+}))$ -valued functions. It follows that:

$$\mathbf{L}_{+}(\mathbf{y}, z) \in \operatorname{End}(H^{\bullet}_{\operatorname{CR}, T}(X_{+})) \otimes_{R_{T}} (\mathcal{O}_{\mathcal{U}^{\circ}_{+}} \otimes R_{T}) \llbracket z^{-1} \rrbracket \llbracket \mathbf{y}_{1}, \dots, \mathbf{y}_{r-1} \rrbracket,$$

$$\Upsilon^{+}(\mathbf{y}, z) \in \operatorname{End}(H^{\bullet}_{\operatorname{CR}, T}(X_{+})) \otimes_{R_{T}} (\mathcal{O}_{\mathcal{U}^{\circ}_{+}} \otimes R_{T})[z] \llbracket \mathbf{y}_{1}, \dots, \mathbf{y}_{r-1} \rrbracket.$$

Comparing (5.19) with (5.15), we obtain

$$L_{+}(\tau_{+}(\mathbf{y}), z)^{-1}\big|_{Q=1} = e^{\sigma_{+}/z} \mathbf{L}_{+}(\mathbf{y}, z)^{-1}.$$
 (5.20)

The mirror map $\tau_+(y)$ is given by $\tau_+(y) = \sigma_+ + \tilde{\tau}_+(y)$ with $\tilde{\tau}_+(y)$ determined by:

$$\mathbf{L}_{+}(\mathbf{y}, z)^{-1}\mathbf{1} = \mathbf{1} + \tilde{\tau}_{+}(\mathbf{y})z^{-1} + O(z^{-2}).$$

We have $\tilde{\tau}_+(0) = 0$ and $\tilde{\tau}_+(\mathbf{y}) \in H^{\bullet}_{\mathrm{CR},T}(X_+) \otimes (\mathcal{O}_{\mathcal{U}^{\circ}_+} \otimes R_T) \llbracket \mathbf{y}_1, \dots, \mathbf{y}_{r-1} \rrbracket$. The first column of (5.19) gives $I_+(\mathbf{y}, z) = zL(\tau(\mathbf{y}), z)|_{Q=1}\Upsilon^+_0$, where Υ^+_0 is the first column of Υ^+ . (Here we assume that the first column corresponds to the basis vector $\phi_{0,1} = 1$ and the differential operator $P^+_{0,1} = 1$.)

Remark 5.24. Equation (5.20) is an equality in the ring:

$$\operatorname{End}(H^{\bullet}_{\operatorname{CR},T}(X_{+})) \otimes_{R_{T}} R_{T}[\log \mathsf{y}_{1},\ldots,\log \mathsf{y}_{r}][\![z^{-1}]\!][\![\mathsf{y}]\!].$$

Note that the substitution $\tau = \tau_+(y)$ in $L_+|_{Q=1}$ makes sense: see §3.2.

By equation (5.20), $\mathbf{L}_{+}(\mathbf{y}, z)$ determines the quantum connection pulled-back by the mirror map $\tau_{+}(\mathbf{y})$. Set $\tau_{+}^{*}\nabla^{+} = d + z^{-1}\mathbf{A}^{+}(\mathbf{y})$. The connection 1-form $\mathbf{A}^{+}(\mathbf{y})$ is computed by:

$$\begin{aligned} \mathbf{A}^{+}(\mathbf{y}) &:= -zd(\mathbf{L}_{+}(\mathbf{y},z)e^{-\sigma_{+}/z})e^{\sigma_{+}/z}\mathbf{L}_{+}(\mathbf{y},z)^{-1} \\ &= -z(d\mathbf{L}_{+}(\mathbf{y},z))\mathbf{L}_{+}(\mathbf{y},z)^{-1} + \mathbf{L}_{+}(\mathbf{y},z)(d\sigma_{+})\mathbf{L}_{+}(\mathbf{y},z)^{-1} \end{aligned}$$

where the term $d\sigma_+$ gives a logarithmic singularity (see equation (5.10)):

$$d\sigma_+ = \sum_{i=1}^{\ell_+} \theta_+(p_i^+) \frac{dy_i}{y_i} - \sum_{j \in S_+} \lambda_j \frac{dx_j}{x_j}.$$

Thus the connection form $\mathbf{A}^+(y)$ is a global 1-form on $\widetilde{\mathcal{M}}_+$ satisfying the properties in Theorem 5.14.

Remark 5.25. Note that $\mathbf{A}^+(\mathbf{y})$ is independent of z: in the formal neighbourhood of $P_+ = \{\mathbf{y}_1 = \cdots = \mathbf{y}_r = 0\}$ this follows from the fact that $d + z^{-1}\mathbf{A}^+(\mathbf{y})$ is the pulled-back quantum connection, and this is true everywhere by analytic continuation.

Finally we see that \mathbf{E}^+ corresponds to \mathcal{E}^+ . Choose a homogeneous R_T -basis $\{\phi_i\}$ of $H^{\bullet}_{\mathrm{CR},T}(X_+)$ such that $\phi_0 = 1$ and $\phi_i = \theta(p_i^+)$ for $1 \leq i \leq \ell_+$ and write $\tau^i_+(\mathsf{y})$ for the *i*th component of $\tau_+(\mathsf{y})$ with respect to this basis. One needs to check that $\mathbf{E}^+\tau^i_+(\mathsf{y}) = (1 - \frac{1}{2} \deg \phi_i)\tau^i_+(\mathsf{y}) + \rho^i$, where $\rho = \sum_{i=0}^N \rho^i \phi_i$. The homogeneity of \mathbf{L}^{-1}_+ shows that $\tilde{\tau}_+(\mathsf{y})$ is homogeneous of (real) degree two: this implies that $\mathbf{E}^+\tilde{\tau}^i_+(\mathsf{y}) = (1 - \frac{1}{2} \deg \phi_i)\tilde{\tau}^i_+(\mathsf{y})$. The rest is a straightforward computation. This completes the proof of Theorem 5.14.

Remark 5.26. For the existence of a global quantum connection in Theorem 5.14 and other main results in this paper, we do not need Assumption 5.20. Let us write $\widetilde{\mathcal{M}}^S_+$ for $\widetilde{\mathcal{M}}_+$ to emphasize the dependence on the extension data S. Then one has:

$$S \subset S' \implies \widetilde{\mathcal{M}}^S_+ \subset \widetilde{\mathcal{M}}^{S'}_+$$

Suppose that an S-extended stacky fan does not satisfy Assumption 5.20. By taking a bigger $S' \supset S$, we can achieve Assumption 5.20 and construct a global quantum connection on $\widetilde{\mathcal{M}}_{+}^{S'}$. Then we obtain a global quantum connection on $\widetilde{\mathcal{M}}_{+}^{S}$ by restriction. In this way, the global quantum connections form a projective system over all extension data S. Assumption 5.20 ensures that \mathbf{F}^+ is generated by a section Υ_0^+ and its covariant derivatives. For the convenience of discussion, we will sometimes use Assumption 5.20 in the rest of the paper, but this does not affect the final conclusion.

6. The Crepant Resolution Conjecture

We now come to the main result in this paper. In Theorem 5.14, we constructed a global quantum connection $(\mathbf{F}^+, \nabla^+, \mathbf{E}^+)$ for X_+ on $\widetilde{\mathcal{M}}_+^\circ$, where $\widetilde{\mathcal{M}}_+^\circ$ is an open subset

of the universal cover $\widetilde{\mathcal{M}}_+$ of $(\widehat{U}_+ \setminus \{\mathbf{y}^e = \mathfrak{c}\})/\boldsymbol{\mu}_B$. By applying Theorem 5.14 to X_- rather than X_+ , we obtain a global quantum connection $(\mathbf{F}^-, \nabla^-, \mathbf{E}^-)$ for X_- on $\widetilde{\mathcal{M}}_-^\circ$, where $\widetilde{\mathcal{M}}_-^\circ$ is an open subset of the universal cover $\widetilde{\mathcal{M}}_-$ of $(\widehat{U}_- \setminus \{\mathbf{y}^e = \mathfrak{c}\})/\boldsymbol{\mu}_A$. We now show that these two global quantum connections are gauge-equivalent on a common covering $\widetilde{\mathcal{M}}$: the universal cover of $\widehat{\mathcal{M}}_{\mathrm{reg}} \setminus \{\mathbf{y}^e = 0, \mathfrak{c}, \infty\}$.



Moreover, we show that the analytic continuation of flat sections is induced by a Fourier-Mukai transformation $\mathbb{FM}: K_T^0(X_-) \to K_T^0(X_+)$ through the equivariant integral structure in §3.1. We establish the gauge-equivalence of the two global quantum connections in several steps, beginning in §6.1 by expressing the gauge transformation involved as a linear symplectomorphism \mathbb{U} between the Givental spaces for X_+ and X_- . In §6.2 we use the Mellin-Barnes method to analytically continue the *I*-function I_+ , deducing from this a formula for \mathbb{U} . In §6.3 we construct a Fourier-Mukai transformation $\mathbb{FM}: K_T^0(X_-) \to K_T^0(X_+)$ associated to the toric birational transformation $X_+ \dashrightarrow X_-$. Finally in §6.4 and §6.5 we complete the proof of gauge-equivalence, and of the Crepant Resolution Conjecture in the toric case, by showing that the symplectic transformation \mathbb{U} coincides, via the equivariant integral structure, with the Fourier-Mukai transformation \mathbb{FM} .

6.1. The global quantum connections are gauge-equivalent

Let \mathcal{U}_{\pm} denote the underlying topological space of $\widetilde{\mathcal{M}}_{\pm}$. The space \mathcal{U}_{+} is the universal cover of $\{y_r \in \mathbb{C} : y_r^{\mathbf{p}_r^+ \cdot e} \neq \mathfrak{c}\}$ and \mathcal{U}_- is the universal cover of $\{\tilde{y}_r \in \mathbb{C} : \tilde{y}_r^{\mathbf{p}_r^- \cdot (-e)} \neq \mathfrak{c}^{-1}\}$. The underlying topological space of $\widetilde{\mathcal{M}}$ is the universal cover \mathcal{U} of $\mathcal{C}_{reg} \setminus \{y^e = 0, \mathfrak{c}, \infty\}$. We have natural maps $\pi_{\pm} : \mathcal{U} \to \mathcal{U}_{\pm}$ and set

$$\mathcal{U}^{\circ} := \pi_{+}^{-1}(\mathcal{U}_{+}^{\circ}) \cap \pi_{-}^{-1}(\mathcal{U}_{-}^{\circ}) \subset \mathcal{U}$$
$$\widetilde{\mathcal{M}}^{\circ} := \widetilde{\mathcal{M}}|_{\mathcal{U}^{\circ}}$$

where $\mathcal{U}_{\pm}^{\circ} \subset \mathcal{U}_{\pm}$ is the open dense subset from Theorem 5.14. Note that $\mathcal{U} \setminus \mathcal{U}^{\circ}$ is a discrete set. Since we use $P_{\pm} \in \mathcal{C}_{\text{reg}}$ as base points of the universal covers \mathcal{U}_{\pm} , we need to specify a path from P_{+} to P_{-} in $\mathcal{C}_{\text{reg}} \setminus \{y^{e} = \mathfrak{c}\}$ in order to identify the maps $\mathcal{U} \to \mathcal{U}_{\pm}$ between universal covers. We consider a path in the $\log(y^{e})$ -plane starting from $\log(y^{e}) = -\infty$



Fig. 1. The path γ of analytic continuation on the log(y^e)-plane.

and ending at $\log(y^e) = \infty$ such that it avoids $\log(\mathfrak{c}) + 2\pi \mathfrak{i}\mathbb{Z}$. We use a path γ as in Fig. 1 passing through the interval

$$\left(\log|\mathfrak{c}| + \pi \mathtt{i}(w-1), \log|\mathfrak{c}| + \pi \mathtt{i}(w+1)\right)$$

in the log(\mathbf{y}^e)-plane, where $w := -1 - \sum_{j:D_j \cdot e < 0} (D_j \cdot e) = -1 + \sum_{j:D_j \cdot e > 0} (D_j \cdot e).$

Theorem 6.1. Let $\mathcal{H}(X_{\pm}) = H^{\bullet}_{CR,T}(X_{\pm}) \otimes_{R_T} R_T((z^{-1}))$ denote Givental's symplectic vector space for X_{\pm} (see §2.5) without Novikov variables, i.e. with Q specialized to 1. There exists a degree-preserving¹² $R_T((z^{-1}))$ -linear symplectic transformation $\mathbb{U} \colon \mathcal{H}(X_-) \to \mathcal{H}(X_+)$ such that:

- (1) $I_+(\mathbf{y}, z) = \mathbb{U}I_-(\mathbf{y}, z)$ after analytic continuation in \mathbf{y}^e along the path γ in Fig. 1;
- (2) $\mathbb{U} \circ (g_{-}^{\star}v \cup) = (g_{+}^{\star}v \cup) \circ \mathbb{U}$ for all $v \in H_T^2(\overline{X}_0)$, where $g_{\pm} \colon X_{\pm} \to \overline{X}_0$ is the common blow-down appearing in the diagram (1.3);
- (3) there exists a Fourier–Mukai transformation \mathbb{FM} : $K_T^0(X_-) \to K_T^0(X_+)$ such that the following diagram commutes:

where the vertical map $\widetilde{\Psi}_{\pm} \colon K^0_T(X_{\pm}) \to \widetilde{\mathcal{H}}(X_{\pm})$ is the map

$$\widetilde{\Psi}_{\pm}(E) = z^{-\mu^{\pm}} z^{\rho^{\pm}} \left(\widehat{\Gamma}_{X_{\pm}} \cup (2\pi i)^{\frac{\deg_0}{2}} \operatorname{inv}^* \widetilde{\operatorname{ch}}(E) \right)$$

taking values¹³ in the "multi-valued Givental space":

¹³ Cf. Corollary 2.6.

¹² We use the usual grading on $H^{\bullet}_{CR,T}(X_{\pm})$, $R_T = H^{\bullet}_T(pt)$ and set deg z = 2.

$$\widetilde{\mathcal{H}}(X_{\pm}) = H^{\bullet}_{\mathrm{CR},T}(X_{\pm}) \otimes_{R_T} R_T[\log z]((z^{-1/k}))$$

Here $k \in \mathbb{N}$ is an integer such that all the eigenvalues of $k\mu^+$, $k\mu^-$ are integers.

Theorem 6.1 will be proved in §6.2 and §6.3. The Fourier–Mukai kernel will be described in §6.3: it is given by a toric common blow-up \widetilde{X} of X_{\pm} .

Notation 6.2. In Theorem 6.1, $\rho^{\pm} = c_1^T(TX_{\pm}) \in H^2_T(X_{\pm}), \ \mu^{\pm}$ is the grading operator (2.4) on $H^{\bullet}_{CR,T}(X_{\pm})$ and $\deg_0: H^{\bullet\bullet}_T(IX_{\pm}) \to H^{\bullet\bullet}_T(IX_{\pm})$ is the degree operator as in §3.1.

Theorem 6.3. Let $(\mathbf{F}^{\pm}, \nabla^{\pm}, \mathbf{E}^{\pm})$ be the global quantum connections for X_{\pm} over $\widetilde{\mathcal{M}}^{\circ}_{\pm}(R_T[z])$ from Theorem 5.14. We have that $\mathbf{E}^+ = \mathbf{E}^-$ on $\widetilde{\mathcal{M}}(R_T)$. There exists a gauge transformation

$$\Theta \in \operatorname{Hom}\left(H^{\bullet}_{\operatorname{CR},T}(X_{-}), H^{\bullet}_{\operatorname{CR},T}(X_{+})\right) \otimes_{R_{T}} (\mathcal{O}_{\mathcal{U}^{\circ}} \otimes R_{T})[z]\llbracket \mathsf{y}_{1}, \ldots, \mathsf{y}_{r-1}]$$

over $\widetilde{\mathcal{M}}^{\circ}(R_T[z])$ such that:

- ∇^- and ∇^+ are gauge-equivalent via Θ , i.e. $\nabla^+ \circ \Theta = \Theta \circ \nabla^-$;
- Θ is homogeneous of degree zero, i.e. $\mathbf{Gr}^+ \circ \Theta = \Theta \circ \mathbf{Gr}^-$ with $\mathbf{Gr}^{\pm} := z \frac{\partial}{\partial z} + \mathbf{E}^{\pm} + \mu^{\pm};$
- Θ preserves the orbifold Poincaré pairing, i.e. $(\Theta(\mathbf{y}, -z)\alpha, \Theta(\mathbf{y}, z)\beta) = (\alpha, \beta)$.

Moreover, the analytic continuation of the K-theoretic flat sections in Definition 3.1 (with Novikov variables Q set to be one, see §3.2) is induced by the Fourier–Mukai transformation:

$$\Theta\Big(\mathfrak{s}(E)(\tau_{-}(\mathsf{y}),z)\Big) = \mathfrak{s}(\mathbb{F}\mathbb{M}(E))(\tau_{+}(\mathsf{y}),z) \qquad \text{for all } E \in K^{0}_{T}(X_{-})$$

where τ_{\pm} are the mirror maps in Theorem 5.14.

Remark 6.4. The symplectic transformation \mathbb{U} in Theorem 6.1 and the gauge transformation Θ in Theorem 6.3 are related by

$$L_{+}(\tau_{+}(\mathbf{y}), z)^{-1} \circ \Theta = \mathbb{U} \circ L_{-}(\tau_{-}(\mathbf{y}), z)^{-1}$$
(6.2)

where L_{\pm} is the fundamental solution for the quantum connection of X_{\pm} in Proposition 2.4. The gauge transformation Θ sends the section $\Upsilon_0^- \in \mathbf{F}^-$ to the section $\Upsilon_0^+ \in \mathbf{F}^+$, where Υ_0^{\pm} are as in Theorem 5.14.

Remark 6.5. Theorems 6.1 and 6.3 can be interpreted as the statement that the symplectic transformation \mathbb{U} matches up the Givental cones \mathcal{L}_{\pm} associated to X_{\pm} after analytic continuation of \mathcal{L}_{\pm} :

$$\mathbb{U}(-z)\mathcal{L}_{-} = \mathcal{L}_{+}.\tag{6.3}$$

In fact, Remark 2.5 suggests that we may analytically continue the Lagrangian cones by the formula:

$$\mathcal{L}_{\pm} = \bigcup_{\mathbf{y} \in \widetilde{\mathcal{M}}^{\circ}} zL_{\pm}(\tau_{\pm}(\mathbf{y}), -z)^{-1}\mathcal{H}_{+}(X_{\pm})$$

and then equation (6.2) would imply (6.3). As discussed in the Introduction, to avoid subtleties in defining the analytic continuation of Givental cones in the equivariant setting, in this paper we state our results in terms of analytic continuation of the *I*-function (Theorem 6.1) or in terms of the equivariant quantum connection and gauge transformations (Theorem 6.3).

Remark 6.6. Theorem 6.3 implies that the global quantum connections of X_+ and X_- can be glued together to give a flat connection over $\widetilde{\mathcal{M}}^\circ$. This flat connection descends to the formal neighbourhood $\widehat{\mathcal{M}}$ of \mathcal{C} in the secondary toric variety \mathcal{M} via Galois symmetry as in Remark 5.10. This global connection, or D-module, on $\widehat{\mathcal{M}}$ can be described by explicit GKZ-type differential equations: it is a completed version of Borisov–Horja's *better-behaved GKZ system*¹⁴ [14]. In the papers [49,69], the toric quantum connection is described in terms of GKZ-type differential equations through mirror symmetry. The *I*-functions $I_{\pm}(q, z)$ are local solutions to these differential equations around the large radius limit points.

Proof that Theorem 6.1 implies Theorem 6.3. One can easily check that the change of variables (5.8) preserves degree, and that $\mathbf{E}^+ = \mathbf{E}^-$. By Theorem 6.1, we have

$$I_{+}(\mathbf{y}, z) = \mathbb{U}I_{-}(\mathbf{y}, z) \tag{6.4}$$

under analytic continuation along the path γ . The discussion in §5.5 (see Lemma 5.22, equation (5.17), and equation (5.19)) yields:

$$\begin{bmatrix} \cdots & z^{-1} P_{f,i}^+ I_+ & \cdots \\ & & \end{bmatrix} = e^{\sigma_+/z} \mathbf{L}_+(\mathbf{y}, z)^{-1} \Upsilon^+(\mathbf{y}, z).$$
(6.5)

Similarly, the discussion in §5.5 applied to X_- yields a global section Υ_0^- of \mathbf{F}^- and a (global) fundamental solution $\mathbf{L}_-(\mathbf{y}, z)e^{-\sigma_-/z}$ for $\nabla^- = d + z^{-1}\mathbf{A}^-(\mathbf{y})$ such that:

$$z^{-1}I_{-}(\mathbf{y}, z) = e^{\sigma_{-}/z} \mathbf{L}_{-}(\mathbf{y}, z)^{-1} \Upsilon_{0}^{-}(\mathbf{y}, z)$$

¹⁴ The better-behaved GKZ system is in general generated by several elements. In our case, by adding enough extended vectors that Assumption 5.20 is satisfied, we can make it generated by a single standard generator 1, and in this case the better-behaved GKZ system is the same as the original GKZ system [39].

Applying the differential operators $P_{f,i}^+$ to $z^{-1}I_-(y, z)$, we obtain

$$\begin{bmatrix} \cdots & z^{-1} P_{f,i}^+ I_- & \cdots \end{bmatrix} = e^{\sigma_-/z} \mathbf{L}_-(\mathbf{y}, z)^{-1} \begin{bmatrix} \cdots & P_{f,i}^+(z \nabla^-) \Upsilon_0^- & \cdots \\ | & | \end{bmatrix}$$
(6.6)

where $P_{f,i}^+(z\nabla^-)$ is obtained from $P_{f,i}$ by replacing $z\partial_v$ with $z\nabla_v^-$ for vector fields v. Let $\widetilde{\Upsilon}^-$ denote the matrix with column vectors $P_{f,i}^+(z\nabla^-)\Upsilon_0^-$. Comparing (6.5) with (6.6) and using (6.4), we obtain

$$e^{\sigma_+/z}\mathbf{L}_+(\mathbf{y},z)^{-1}\Upsilon^+ = \mathbb{U}e^{\sigma_-/z}\mathbf{L}_-(\mathbf{y},z)^{-1}\widetilde{\Upsilon}^-$$

since \mathbb{U} is independent of the base variables y. In particular, it follows that $\widetilde{\Upsilon}^-$ is invertible. Setting $\Theta = \Upsilon^+(\widetilde{\Upsilon}^-)^{-1}$, we obtain:

$$\left(e^{\sigma_+/z}\mathbf{L}_+(\mathbf{y},z)^{-1}\right)\Theta(\mathbf{y},z) = \mathbb{U}\left(e^{\sigma_-/z}\mathbf{L}_-(\mathbf{y},z)^{-1}\right).$$
(6.7)

Since $e^{\sigma_{\pm}/z} \mathbf{L}_{\pm}^{-1}$ are fundamental solutions for ∇^{\pm} , Θ gives a gauge transformation between ∇^{-} and ∇^{+} , i.e. $\Theta \circ \nabla^{-} = \nabla^{+} \circ \Theta$. One may assume that the first columns of Υ^{+} and $\widetilde{\Upsilon}^{-}$ are given respectively by Υ_{0}^{+} and Υ_{0}^{-} , and therefore $\Theta(\Upsilon_{0}^{-}) = \Upsilon_{0}^{+}$.

Next we see that Θ preserves the grading and the pairing. Part (2) in Theorem 6.1 implies that $\mathbb{U} \circ \theta_{-}(\mathbf{p}_{i}^{-}) = \theta_{+}(\mathbf{p}_{i}^{+}) \circ \mathbb{U}$ for $i = 1, \ldots, r-1$, since $\mathbf{p}_{i}^{+} = \mathbf{p}_{i}^{-}$ lies on the wall W for $1 \leq i \leq r-1$. Therefore

$$e^{-\sigma_+/z} \circ \mathbb{U} \circ e^{\sigma_-/z} = e^{-\theta_+(\mathbf{p}_r^+)\log \mathbf{y}_r/z} \circ \mathbb{U} \circ e^{\theta_-\left(\sum_{i=1}^r \mathbf{p}_i^-\log \tilde{\mathbf{y}}_i - \sum_{i=1}^{r-1} \mathbf{p}_i^+\log \mathbf{y}_i\right)/z}$$
$$= e^{-\theta_+(\mathbf{p}_r^+)\log \mathbf{y}_r/z} \circ \mathbb{U} \circ e^{\theta_-(\mathbf{p}_r^+)\log \mathbf{y}_r/z}$$

where we used $\sum_{i=1}^{r} \mathbf{p}_{i}^{+} \log \mathbf{y}_{i} = \sum_{i=1}^{r} \mathbf{p}_{i}^{-} \log \tilde{\mathbf{y}}_{i}$. This together with (6.7) implies that:

$$\mathbf{L}_{+}(\mathbf{y},z)^{-1}\Theta(\mathbf{y},z) = \left(e^{-\theta_{+}(\mathbf{p}_{r}^{+})\log \mathbf{y}_{r}/z} \circ \mathbb{U} \circ e^{\theta_{-}(\mathbf{p}_{r}^{+})\log \mathbf{y}_{r}/z}\right)\mathbf{L}_{-}(\mathbf{y},z)^{-1}$$

Since deg $y_r = 0$, we know that all of the factors in this equation except for Θ are homogeneous of degree zero; thus Θ is also homogeneous of degree zero. The fundamental solutions $e^{\sigma_{\pm}/z} \mathbf{L}_{\pm}^{-1}$ preserve the pairing by Proposition 2.4 (we saw in §5.5 that they coincide with the fundamental solutions from Proposition 2.4 via the mirror maps τ_{\pm}) and \mathbb{U} also preserves the pairing. Thus Θ preserves the pairing.

Finally we consider the analytic continuation of K-theoretic flat sections. Note that the flat section $\mathfrak{s}(E)(\tau_{-}(\mathsf{y}), z)$ is analytically continued along $\widetilde{\mathcal{M}}^{\circ}$ by the right-hand side of the formula

$$\mathfrak{s}(E)(\tau_{-}(\mathbf{y}), z) = \frac{1}{(2\pi)^{\dim X_{-}/2}} \mathbf{L}_{-}(\mathbf{y}, z) e^{-\sigma_{-}/z} \widetilde{\Psi}_{-}(E)$$

where $\widetilde{\Psi}_{-}$ is the map in Theorem 6.1. Using (6.7), we obtain:

$$\Theta(\mathfrak{s}(E)(\tau_{-}(\mathbf{y},z))) = \frac{1}{(2\pi)^{\dim X_{-}/2}} \mathbf{L}_{+}(\mathbf{y},z) e^{-\sigma_{+}/z} \mathbb{U}\left(\widetilde{\Psi}_{-}(E)\right)$$

Part (3) of Theorem 6.1 shows that this is equal to $\mathfrak{s}(\mathbb{FM}(E))(\tau_+(y), z)$. \Box

6.2. Mellin-Barnes analytic continuation

In this section, we compute the analytic continuation of the *I*-function and determine the linear transformation \mathbb{U} in Theorem 6.1.

6.2.1. The H-function

It will be convenient to introduce another cohomology-valued hypergeometric function called the *H*-function. Noting that coefficients of the *I*-function can be written in terms of ratios of Γ -functions:

$$I_{+}(\mathbf{y},z) := ze^{\sigma_{+}/z} \sum_{d \in \mathbb{K}_{+}} \frac{\mathbf{y}^{d}}{z^{(D_{1}+\dots+D_{m})\cdot d}} \left(\prod_{j=1}^{m} \frac{\Gamma\left(1 + \frac{u_{j}}{z} - \langle -D_{j} \cdot d \rangle\right)}{\Gamma\left(1 + \frac{u_{j}}{z} + D_{j} \cdot d\right)} \right) \frac{\mathbf{1}_{[-d]}}{z^{\iota_{[-d]}}}$$

we set:

$$H_{+}(\mathbf{y}) := e^{\frac{\sigma_{+}}{2\pi \mathbf{i}}} \sum_{d \in \mathbb{K}_{+}} \mathbf{y}^{d} \left(\prod_{j=1}^{m} \frac{1}{\Gamma\left(1 + \frac{u_{j}}{2\pi \mathbf{i}} + D_{j} \cdot d\right)} \right) \mathbf{1}_{[d]}$$

and similarly for H_{-} . Formally speaking, H_{+} belongs to the space:

$$\prod_{p} H_T^p(IX_+)[\log y_1, \dots, \log y_r][\![y_1, \dots, y_r]\!]$$

Noting that the *T*-equivariant Gamma class of X_+ is given by

$$\widehat{\Gamma}_{X_+} = \bigoplus_{f \in \mathbb{K}_+ / \mathbb{L}} \left(\prod_{j=1}^m \Gamma(1 + u_j - \langle D_j \cdot f \rangle) \right) \mathbf{1}_f$$

we obtain the relationship between the H-function and the I-function:

$$z^{-1}I_{+}(\mathsf{y},z) = z^{-\frac{c_{0}(\lambda)}{2\pi\mathrm{i}} - \frac{\dim X_{+}}{2}} z^{-\mu^{+}} z^{\rho^{+}} \left(\widehat{\Gamma}_{X_{+}} \cup (2\pi\mathrm{i})^{\frac{\deg_{0}}{2}} \operatorname{inv}^{*} H\left(z^{-\frac{\deg y}{2}}\mathsf{y}\right)\right)$$
(6.8)

where ρ^+ , μ^+ , deg₀ are as in Notation 6.2 and

$$z^{-\frac{\deg \mathbf{y}}{2}}\mathbf{y} = (z^{-\frac{\deg \mathbf{y}_1}{2}}\mathbf{y}_1, \dots, z^{-\frac{\deg \mathbf{y}_r}{2}}\mathbf{y}_r).$$

The relationship between H_{-} and I_{-} is similar.

Remark 6.7. The *H*-function H_+ has analytic properties analogous to those of the *I*-function stated in Lemma 5.13. Namely $e^{-\sigma_+/(2\pi i)}H_+(y)$ is a formal power series in y_1, \ldots, y_{r-1} with coefficients of the form $\sum_{i=0}^N f_i(\lambda, y_r)\phi_i$ where $\{\phi_i\}$ is an R_T -basis of $H_T^{\bullet}(IX_+)$ and $f_i(\lambda, y_r)$ is analytic in $(\lambda_1, \ldots, \lambda_r, y_r) \in \mathbb{C}^m \times \mathcal{U}_+$. Note that the *H*-function has better analytic behaviour with respect to λ . The analytic continuation of *H*-functions performed below should be understood as analytic continuation of the coefficient functions $f_i(\lambda, y_r)$.

6.2.2. Restriction of the H-function to T-fixed points

Recall that the *T*-fixed points on X_+ are indexed by minimal anticones $\delta \in \mathcal{A}_+$, and that the *T*-fixed points on the inertia stack IX_+ are indexed by pairs (δ, f) with $\delta \in \mathcal{A}_+$ a minimal anticone and $f \in \mathbb{K}_+/\mathbb{L}$ satisfying $D_i \cdot f \in \mathbb{Z}$ for all $i \in \delta$. The minimal anticone δ determines a *T*-fixed point x_δ on X_+ and the pair (δ, f) determines a *T*-fixed point $x_{(\delta,f)}$ on the component X_+^f of the inertia stack IX_+ . Let i_δ and $i_{(\delta,f)}$ denote the inclusion maps $x_\delta \to X_+$ and $x_{(\delta,f)} \to IX_+$ respectively. Set $u_j(\delta) = i_\delta^* u_j \in H_T^2(\mathrm{pt})$, noting that $u_j(\delta) = 0$ if and only if $j \in \delta$. We have that:

$$i_{(\delta,f)}^{\star}H_{+} = \sum_{d \in \mathbb{K}_{+}:[d]=f} \frac{\mathsf{y}^{d}}{\prod_{j \in \delta} \Gamma(1+D_{j} \cdot d)} \frac{e^{\frac{1}{2\pi \mathbf{i}}\sigma_{+}(\delta)}}{\prod_{j \notin \delta} \Gamma(1+\frac{u_{j}(\delta)}{2\pi \mathbf{i}}+D_{j} \cdot d)} \tag{6.9}$$

where $\sigma_+(\delta) := i_{\delta}^{\star} \sigma_+$. Consider the factor $\prod_{j \in \delta} \Gamma(1 + D_j \cdot d)^{-1}$ in the summand: since $d \equiv f \mod \mathbb{L}$ and since $D_j \cdot f \in \mathbb{Z}$ for all $j \in \delta$, the term $D_j \cdot d$ here is an integer. Thus the factor $\prod_{j \in \delta} \Gamma(1 + D_j \cdot d)^{-1}$ vanishes unless $d \in \delta^{\vee}$, where

$$\delta^{\vee} := \{ d \in \mathbb{L} \otimes \mathbb{Q} : D_j \cdot d \in \mathbb{Z}_{>0} \text{ for all } j \in \delta \}.$$

The *H*-function is a sum over the subset $\mathbb{K}^{\text{eff}}_+$ of \mathbb{K}_+ ,

$$\mathbb{K}^{\text{eff}}_{+} = \left\{ f \in \mathbb{L} \otimes \mathbb{Q} : \left\{ i \in \{1, 2, \dots, m\} : D_i \cdot f \in \mathbb{Z}_{\geq 0} \right\} \in \mathcal{A}_{+} \right\}$$

which is in general quite complicated, but the restriction $i^*_{(\delta,f)}H_+$ of H_+ to a *T*-fixed point in IX_+ is a sum over the much simpler set δ^{\vee} .

6.2.3. Analytic continuation of the H-function

The Localization Theorem in *T*-equivariant cohomology [3,8,45] implies that one can compute the analytic continuation of H_+ by computing the analytic continuation of the restriction $i^*_{(\delta,f)}H_+$ to each *T*-fixed point $x_{(\delta,f)} \in IX_+$. The restriction $i^*_{(\delta,f)}H_+$ is a $H_T^{\bullet\bullet}(\text{pt})$ -valued function. During the course of analytic continuation, we regard the equivariant parameters $\lambda_1, \ldots, \lambda_m$ as generic complex numbers. There are two cases:

- $\delta \in \mathcal{A}_+ \cap \mathcal{A}_-;$
- $\delta \in \mathcal{A}_+$ but $\delta \notin \mathcal{A}_-$.

The anticone δ determines a *T*-fixed point x_{δ} in X_+ , and in the first case it also determines a fixed point in X_- . In the first case the birational transformation $\varphi \colon X_+ \dashrightarrow X_-$ is an isomorphism in a neighbourhood of x_{δ} , and it is clear from (6.9) that $i^*_{(\delta,f)}H_+ = i^*_{(\delta,f)}H_-$, noting that $u_j(\delta)$ is the same for X_+ and X_- . In the second case x_{δ} lies in the flopping locus of φ , and we will see that the analytic continuation of $i^*_{(\delta,f)}H_+$ is a linear combination of restrictions $i^*_{(\delta_-,f_-)}H_-$ for appropriate $\delta_- \in \mathcal{A}_-$ and $f_- \in \mathbb{K}_-$. Note that in the second case, δ has the form $\{j_1, \ldots, j_{r-1}, j_+\}$ with $D_{j_1}, \ldots, D_{j_{r-1}} \in W$ and $i^{15} D_{j_+} \cdot e > 0$ (see Lemma 5.2).

Definition 6.8. Let $\delta_+ \in \mathcal{A}_+$ and $\delta_- \in \mathcal{A}_-$ be minimal anticones. We say that δ_+ is next to δ_- , written $\delta_+|\delta_-$, if $\delta_+ = \{j_1, \ldots, j_{r-1}, j_+\}$ and $\delta_- = \{j_1, \ldots, j_{r-1}, j_-\}$ with $D_{j_1}, \ldots, D_{j_{r-1}} \in W, D_{j_+} \cdot e > 0$, and $D_{j_-} \cdot e < 0$. In this case $\delta_+ \notin \mathcal{A}_-$ and $\delta_- \notin \mathcal{A}_+$.

Definition 6.9. Let (δ_+, f_+) index a *T*-fixed point on IX_+ and (δ_-, f_-) index a *T*-fixed point on IX_- . We say that (δ_+, f_+) is *next to* (δ_-, f_-) , written $(\delta_+, f_+)|(\delta_-, f_-)$, if $\delta_+|\delta_-$ and there exists $\alpha \in \mathbb{Q}$ such that $f_- = f_+ + \alpha e$ in $\mathbb{L} \otimes \mathbb{Q}/\mathbb{L}$.

The analytic continuation of $i^*_{(\delta,f)}H_+$ is a linear combination of $i^*_{(\delta_-,f_-)}H_-$ such that (δ, f) is next to (δ_-, f_-) .

Notation 6.10. Fix lifts $\mathbb{K}_+/\mathbb{L} \to \mathbb{K}_+$ and $\mathbb{K}_-/\mathbb{L} \to \mathbb{K}_-$ such that, for any pairs $(d_+, d_-) \in \mathbb{K}_+ \times \mathbb{K}_-$ with $d_+ - d_- \in \mathbb{Q}e$, the lifts of $[d_+]$ and $[d_-]$ differ by a rational multiple of e.

Lemma 6.11. Let $\delta_+ \in \mathcal{A}_+$ and $\delta_- \in \mathcal{A}_-$ be minimal anticones such that $\delta_+|\delta_-$, and let j_- be the element of δ_- such that $j_- \notin \delta_+$. Then for any j, one has:

$$u_j(\delta_+) = u_j(\delta_-) + \frac{D_j \cdot e}{D_{j_-} \cdot e} u_{j_-}(\delta_+).$$

Proof. Write $\delta_{-} = \{j_1, \ldots, j_{r-1}, j_{-}\}$. Since $D_{j_1}, \ldots, D_{j_{r-1}}, D_{j_{-}}$ form a basis of $\mathbb{L}^{\vee} \otimes \mathbb{Q}$, we can write:

$$D_j = c_1 D_{j_1} + \dots + c_{r-1} D_{j_{r-1}} + c_- D_{j_-}$$

Since $D_{j_1}, \ldots, D_{j_{r-1}}$ are on the wall, pairing with *e* yields:

$$D_j \cdot e = c_-(D_{j_-} \cdot e). \tag{6.10}$$

Applying the homomorphism θ_{\pm} from (4.8), we obtain

¹⁵ Recall that $e \in \mathbb{L}$ is the primitive lattice vector in W^{\perp} such that e > 0 on C_+ and e < 0 on C_- .

$$u_j - \lambda_j = c_1(u_{j_1} - \lambda_{j_1}) + \dots + c_{r-1}(u_{j_{r-1}} - \lambda_{j_{r-1}}) + c_-(u_{j_-} - \lambda_{j_-})$$

on both X_+ and X_- . Restricting to $x_{\delta_+} \in X_+$ and $x_{\delta_-} \in X_-$, we get the two relations:

$$u_{j}(\delta_{+}) - \lambda_{j} = -c_{1}\lambda_{j_{1}} - \dots - c_{r-1}\lambda_{j_{r-1}} + c_{-}(u_{j}(\delta_{+}) - \lambda_{j_{-}}),$$

$$u_{j}(\delta_{-}) - \lambda_{j} = -c_{1}\lambda_{j_{1}} - \dots - c_{r-1}\lambda_{j_{r-1}} - c_{-}\lambda_{j_{-}}.$$

Comparing the two equations, we get $u_j(\delta_+) = u_j(\delta_-) + c_- u_{j_-}(\delta_+)$. The conclusion now follows from equation (6.10). \Box

Corollary 6.12.

- (1) Let δ be a minimal anticone such that $\delta \in \mathcal{A}_+ \cap \mathcal{A}_-$. Then $\sigma_+(\delta) = \sigma_-(\delta)$.
- (2) Let $\delta_+ \in \mathcal{A}_+$, $\delta_- \in \mathcal{A}_-$ be minimal anticones such that $\delta_+|\delta_-$ and let $j_- \in \delta_-$ be an element such that $j_- \notin \delta_+$. Then:

$$\sigma_+(\delta_+) = \sigma_-(\delta_-) + \frac{\log y^e}{D_{j_-} \cdot e} u_{j_-}(\delta_+)$$

Proof.

- (1) As we discussed, $u_j(\delta)$ is the same for X_+ and X_- whenever $\delta \in \mathcal{A}_+ \cap \mathcal{A}_-$. Therefore $i^*_{\delta}\theta_+(D_j) = i^*_{\delta}\theta_-(D_j)$ for all j. In particular $i^*_{\delta}\theta_+(p) = i^*_{\delta}\theta_-(p)$ for every $p \in \mathbb{L}^{\vee} \otimes \mathbb{C}$. Setting $p = \sum_{i=1}^r \mathsf{p}^+_i \log \mathsf{y}_i = \sum_{i=1}^r \mathsf{p}^-_i \log \tilde{\mathsf{y}}_i$, we obtain (1).
- (2) Lemma 6.11 shows that

$$i_{\delta_{+}}^{\star}\theta_{+}(p) = i_{\delta_{-}}^{\star}\theta_{-}(p) + \frac{p \cdot e}{D_{j_{-}} \cdot e}u_{j_{-}}(\delta_{+})$$
(6.11)

for all $p \in \mathbb{L}^{\vee} \otimes \mathbb{C}$. Setting again $p = \sum_{i=1}^{r} \mathsf{p}_{i}^{+} \log \mathsf{y}_{i} = \sum_{i=1}^{r} \mathsf{p}_{i}^{-} \log \tilde{\mathsf{y}}_{i}$, we obtain (2). \Box

Theorem 6.13. Let (δ_+, f_+) index a *T*-fixed point on IX_+ . If $\delta_+ \in \mathcal{A}_+ \cap \mathcal{A}_-$ then:

$$i^{\star}_{(\delta_+,f_+)}H_+ = i^{\star}_{(\delta_+,f_+)}H_-.$$

Otherwise, after analytic continuation along the path γ in Fig. 1, we have:

$$i_{(\delta_{+},f_{+})}^{\star}H_{+} = \sum_{\substack{(\delta_{-},f_{-}):\\(\delta_{+},f_{+})\mid (\delta_{-},f_{-})}} C_{\delta_{+},f_{+}}^{\delta_{-},f_{-}} i_{(\delta_{-},f_{-})}^{\star}H_{-}$$

where:

$$C_{\delta_{+},f_{+}}^{\delta_{-},f_{-}} = e^{\frac{\pi i w}{D_{j_{-}} \cdot e} \left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i} + D_{j_{-}} \cdot (f_{+} - f_{-})\right)} \times \frac{\sin \pi \left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i} + D_{j_{-}} \cdot (f_{+} - f_{-})\right)}{(-D_{j_{-}} \cdot e) \sin \frac{\pi}{-D_{j_{-}} \cdot e} \left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i} + D_{j_{-}} \cdot (f_{+} - f_{-})\right)} \prod_{\substack{j: D_{j} \cdot e < 0 \\ j \neq j_{-}}} \frac{\sin \pi \left(\frac{u_{j}(\delta_{+})}{2\pi i} + D_{j} \cdot f_{+}\right)}{\sin \pi \left(\frac{u_{j}(\delta_{-})}{2\pi i} + D_{j} \cdot f_{-}\right)}$$

with $w := -1 - \sum_{j:D_j \cdot e < 0} D_j \cdot e = -1 + \sum_{j:D_j \cdot e > 0} D_j \cdot e$ and $j_- \in \delta_-$ given by the unique element such that $D_{j_-} \cdot e < 0$.

Remark 6.14. The coefficient $C_{\delta_+,f_+}^{\delta_-,f_-}$ does not depend on the choice of lifts $f_+ \in \mathbb{K}_+$ and $f_- \in \mathbb{K}_-$ such that $f_+ - f_- \in \mathbb{Q}e$ (see Notation 6.10).

Proof of Theorem 6.13. The first statement follows immediately from (6.9) and Corollary 6.12. In this case, $i^*_{(\delta_+,f_+)}H_+$ (respectively $i^*_{(\delta_+,f_+)}H_-$) is a formal power series in y_1, \ldots, y_{r-1} (respectively in $\tilde{y}_1, \ldots, \tilde{y}_{r-1}$) with coefficients that are *polynomials* in y_r (respectively in \tilde{y}_r), and the series $i^*_{(\delta_+,f_+)}H_+$, $i^*_{(\delta_+,f_+)}H_-$ match under the change (5.8) of co-ordinates. Consider now

$$i^{\star}_{(\delta_{+},f_{+})}H_{+} = e^{\frac{\sigma_{+}(\delta_{+})}{2\pi i}} \sum_{\substack{d \in \delta^{\vee}_{+}:\\[d]=f_{+}}} y^{d} \frac{1}{\prod_{j=1}^{m} \Gamma\left(1 + \frac{u_{j}(\delta_{+})}{2\pi i} + D_{j} \cdot d\right)}$$

where $\delta_+ \in \mathcal{A}_+$ but $\delta_+ \notin \mathcal{A}_-$. We can write $d \in \delta_+^{\vee}$ uniquely as $d = d_+ + ke$ with k a non-negative integer, $d_+ \in \delta_+^{\vee}$, and $d_+ - e \notin \delta_+^{\vee}$. Then:

$$i_{(\delta_{+},f_{+})}^{\star}H_{+} = \sum_{\substack{d_{+}\in\delta_{+}^{\vee}:\\d_{+}-e\notin\delta_{+}^{\vee}\\[d_{+}]=f_{+}}} y^{d_{+}} \sum_{k=0}^{\infty} \frac{e^{\frac{\sigma_{+}(\delta_{+})}{2\pi i}}(y^{e})^{k}}{\prod_{j=1}^{m}\Gamma\left(1+\frac{u_{j}(\delta_{+})}{2\pi i}+D_{j}\cdot d_{+}+kD_{j}\cdot e\right)}$$
(6.12)

Consider the second sum here. This is:

$$\sum_{k=0}^{\infty} e^{\frac{\sigma_{+}(\delta_{+})}{2\pi \mathbf{i}}} (\mathbf{y}^{e})^{k} \prod_{j:D_{j} \cdot e < 0} \frac{(-1)^{kD_{j} \cdot e} \sin \pi \left(-\frac{u_{j}(\delta_{+})}{2\pi \mathbf{i}} - D_{j} \cdot d_{+} \right)}{\pi} \times \frac{\prod_{j:D_{j} \cdot e < 0} \Gamma \left(-\frac{u_{j}(\delta_{+})}{2\pi \mathbf{i}} - D_{j} \cdot d_{+} - kD_{j} \cdot e \right)}{\prod_{j:D_{j} \cdot e \ge 0} \Gamma \left(1 + \frac{u_{j}(\delta_{+})}{2\pi \mathbf{i}} + D_{j} \cdot d_{+} + kD_{j} \cdot e \right)} \quad (6.13)$$

where we used $\Gamma(y)\Gamma(1-y) = \pi/(\sin \pi y)$. Thus (6.13) is:



Fig. 2. The contour C.

$$\sum_{k=0}^{\infty} e^{\frac{\sigma_{+}(\delta_{+})}{2\pi \mathbf{i}}} \operatorname{Res}_{s=k} \Gamma(s) \Gamma(1-s) e^{\pi \mathbf{i}s} (\mathbf{y}^{e})^{s} \prod_{j:D_{j} \cdot e < 0} \frac{e^{\pi \mathbf{i}s(D_{j} \cdot e)} \sin \pi \left(-\frac{u_{j}(\delta_{+})}{2\pi \mathbf{i}} - D_{j} \cdot d_{+}\right)}{\pi} \times \frac{\prod_{j:D_{j} \cdot e < 0} \Gamma \left(-\frac{u_{j}(\delta_{+})}{2\pi \mathbf{i}} - D_{j} \cdot d_{+} - sD_{j} \cdot e\right)}{\prod_{j:D_{j} \cdot e \ge 0} \Gamma \left(1 + \frac{u_{j}(\delta_{+})}{2\pi \mathbf{i}} + D_{j} \cdot d_{+} + sD_{j} \cdot e\right)} ds. \quad (6.14)$$

Consider now the contour integral

$$e^{\frac{\sigma_{+}(\delta_{+})}{2\pi \mathrm{i}}} \int_{C} \Gamma(s)\Gamma(1-s) \frac{\prod_{j:D_{j}\cdot e < 0} \Gamma\left(-\frac{u_{j}(\delta_{+})}{2\pi \mathrm{i}} - D_{j}\cdot d_{+} - sD_{j}\cdot e\right)}{\prod_{j:D_{j}\cdot e \ge 0} \Gamma\left(1 + \frac{u_{j}(\delta_{+})}{2\pi \mathrm{i}} + D_{j}\cdot d_{+} + sD_{j}\cdot e\right)} \left(e^{-\pi \mathrm{i}w} \mathsf{y}^{e}\right)^{s} ds$$

$$(6.15)$$

where the contour C, shown in Fig. 2, is chosen such that the poles at s = n are on the right of C and the poles at s = -1 - n and at

$$s = \frac{1}{-D_{j_-} \cdot e} \left(\frac{u_{j_-}(\delta_+)}{2\pi i} + D_{j_-} \cdot d_+ - n \right) \qquad \text{for } j_- \text{ such that } D_{j_-} \cdot e < 0 \quad (6.16)$$

are on the left of C; here n is a non-negative integer. Note that all poles of the integrand are simple. By assumption we have that $\sum_{j=1}^{m} D_j \in W$, and hence that $\sum_{j=1}^{m} D_j \cdot e = 0$. Let $\mathfrak{c} \in \mathbb{C}$ be the conifold point (5.13). Lemma A.6 in [12] implies that:

- the contour integral (6.15) is convergent and analytic as a function of y^e in the domain {y^e : | arg(y^e) − wπ| < π};
- for $|y^e| < |c|$, the integral is equal to the sum of residues on the right of C; and
- for $|y^e| > |c|$, the integral is equal to minus the sum of residues on the left of C.

The residues at s = -1 - n vanish, where n is a non-negative integer: each such residue contains a factor

$$\prod_{j\in\delta_+}\Gamma(1+D_j\cdot(d_+-(n+1)e))^{-1}$$

and $d_+ - (n+1)e \notin \delta_+^{\vee}$, so at least one of the Γ -functions is evaluated at a negative integer. After analytic continuation in y^e , therefore, (6.14) becomes minus the sum of residues at the poles (6.16). The residue at the pole

$$p = \frac{1}{-D_{j_{-}} \cdot e} \left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i} + D_{j_{-}} \cdot d_{+} - n \right)$$

is:

$$-e^{\frac{\sigma_{+}(\delta_{+})}{2\pi i}}(\mathbf{y}^{e})^{p}e^{\pi i p(1+D_{j_{-}}\cdot e)}\frac{\sin \pi \left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i}+D_{j_{-}}\cdot d_{+}\right)}{\sin \pi p}\frac{1}{(-D_{j_{-}}\cdot e)}\frac{(-1)^{n}}{n!}$$
$$\prod_{\substack{j:D_{j}\cdot e<0\\j\neq j_{-}}}\frac{e^{\pi i p(D_{j}\cdot e)}\sin \pi \left(\frac{u_{j}(\delta_{+})}{2\pi i}+D_{j}\cdot d_{+}\right)}{\sin \pi \left(\frac{u_{j}(\delta_{+})}{2\pi i}+D_{j}\cdot d_{+}+p(D_{j}\cdot e)\right)}\prod_{j:j\neq j_{-}}\frac{1}{\Gamma\left(1+\frac{u_{j}(\delta_{+})}{2\pi i}+D_{j}\cdot d_{+}+p(D_{j}\cdot e)\right)}$$
(6.17)

This simplifies dramatically. Set $n = k(-D_{j_{-}} \cdot e) + l$ with $0 \le l < (-D_{j_{-}} \cdot e)$,

$$d_{-} = d_{+} + \frac{D_{j-} \cdot d_{+} - l}{-D_{j-} \cdot e}e$$

and $\delta_{-} = \{j_1, \ldots, j_{r-1}, j_-\}$, where $\delta_{+} = \{j_1, \ldots, j_{r-1}, j_+\}$ with $D_{j_1} \cdot e = \cdots = D_{j_{r-1}} \cdot e = 0$. Note that $D_{j_-} \cdot d_- = l \in \mathbb{Z}_{\geq 0}$ but $D_{j_-} \cdot (d_- + e) < 0$, and therefore $d_- \in \delta_-^{\vee}$ but $d_- + e \notin \delta_-^{\vee}$. Lemma 6.11 implies that:

$$\frac{u_j(\delta_+)}{2\pi i} + D_j \cdot d_+ + p(D_j \cdot e) = \frac{u_j(\delta_-)}{2\pi i} + D_j \cdot d_- - k(D_j \cdot e)$$

and thus the residue (6.17) is:

$$-e^{\frac{\sigma_{-}(\delta_{-})}{2\pi i}}y^{d_{-}-d_{+}-ke}e^{\frac{\pi iw}{D_{j_{-}}\cdot e}\left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i}+D_{j_{-}}\cdot(d_{+}-d_{-})\right)} \times \frac{\sin\pi\left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i}+D_{j_{-}}\cdot(d_{+}-d_{-})\right)}{\left(-D_{j_{-}}\cdot e\right)\sin\frac{\pi}{-D_{j_{-}}\cdot e}\left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i}+D_{j_{-}}\cdot(d_{+}-d_{-})\right)} \times \prod_{\substack{j:D_{j}\cdot e<0\\j\neq j_{-}}}\frac{\sin\pi\left(\frac{u_{j}(\delta_{+})}{2\pi i}+D_{j}\cdot d_{+}\right)}{\sin\pi\left(\frac{u_{j}(\delta_{-})}{2\pi i}+D_{j}\cdot d_{-}\right)}\prod_{j=1}^{m}\frac{1}{\Gamma\left(1+\frac{u_{j}(\delta_{-})}{2\pi i}+D_{j}\cdot d_{-}-k(D_{j}\cdot e)\right)}$$
(6.18)

where we used $p = \frac{1}{-D_{j_-} \cdot e} \left(\frac{u_{j_-}(\delta_+)}{2\pi i} + D_{j_-} \cdot (d_+ - d_-) \right) - k$ and Corollary 6.12.

Let f_- denote the equivalence class of d_- in \mathbb{K}_-/\mathbb{L} , noting that $(\delta_+, f_+)|(\delta_-, f_-)$ and that

$$d_{+} = f_{+} - f_{-} + Ne + d_{-}$$

for some integer N. (Here we used Notation 6.10.) The dependence of (6.18) on N cancels, giving:

$$-e^{\frac{\sigma_{-}(\delta_{-})}{2\pi i}} y^{d_{-}-d_{+}-ke} \frac{1}{\prod_{j=1}^{m} \Gamma\left(1+\frac{u_{j}(\delta_{-})}{2\pi i}+D_{j}\cdot d_{-}-k(D_{j}\cdot e)\right)} C^{\delta_{-},f_{-}}_{\delta_{+},f_{+}}$$

and minus the sum of these residues gives the analytic continuation of (6.14). After analytic continuation in $y^e = y_r^{p_r^+,e}$, therefore, we have that:

$$= \sum_{\substack{(\delta_{-},f_{-}):\\(\delta_{+},f_{+})|(\delta_{-},f_{-})}} \sum_{\substack{d_{-}\in\delta_{-}^{\vee}:\\(\delta_{+},f_{+})|(\delta_{-},f_{-})}} \sum_{\substack{d_{-}\in\delta_{-}^{\vee}:\\d_{-}+e\notin\delta_{-}^{\vee}}} y^{d_{-}} \sum_{k=0}^{\infty} \frac{e^{\frac{\sigma_{-}(\delta_{-})}{2\pi i}}(y^{e})^{-k}}{\prod_{j=1}^{m}\Gamma\left(1+\frac{u_{j}(\delta_{-})}{2\pi i}+D_{j}\cdot d_{-}-kD_{j}\cdot e\right)} C^{\delta_{-},f_{-}}_{\delta_{+},f_{+}}$$

Comparing with (6.12) gives the result. \Box

6.2.4. Analytic continuation of the I-function and the symplectic transformation $\mathbb U$

Set $\widehat{R}_T = H_T^{\bullet \bullet}(\mathrm{pt})$ and let \widehat{S}_T be the completion of S_T in §3.1. Define an \widehat{S}_T -linear transformation $\mathbb{U}_H : H_T^{\bullet \bullet}(IX_-) \otimes_{\widehat{R}_T} \widehat{S}_T \to H_T^{\bullet \bullet}(IX_+) \otimes_{\widehat{R}_T} \widehat{S}_T$ by

$$\mathbb{U}_{H}(\alpha) = \sum_{\substack{(\delta,f):\\\delta\in\mathcal{A}_{+}\cap\mathcal{A}_{-}}} (i_{(\delta,f)}^{\star}\alpha) \cdot \frac{\mathbf{1}_{\delta,f}}{e_{T}(N_{\delta,f})} + \sum_{\substack{(\delta_{+},f_{+}):\\\delta_{+}\in\mathcal{A}_{+}\setminus\mathcal{A}_{-}}} \sum_{\substack{(\delta_{-},f_{-}):\\(\delta_{-},f_{-})}} C_{\delta_{-},f_{-}}^{\delta_{+},f_{-}} \cdot (i_{(\delta_{-},f_{-})}^{\star}\alpha) \cdot \frac{\mathbf{1}_{\delta_{+},f_{+}}}{e_{T}(N_{\delta_{+},f_{+}})}}$$
(6.19)

where (δ, f) and (δ_+, f_+) index *T*-fixed points in IX_+ , $\mathbf{1}_{\delta,f} = i_{(\delta,f)\star}\mathbf{1}$ and $N_{\delta,f} := T_{x_{(\delta,f)}}X_+^f$. Then Theorem 6.13 can be restated as:

$$H_+ = \mathbb{U}_H H_-$$

Define the linear transformation \mathbb{U} so that the following diagram commutes:

where the vertical maps are defined by $\widetilde{\Psi}'_{\pm}(\alpha) = z^{-\mu^{\pm}} z^{\rho^{\pm}}(\widehat{\Gamma}_{X_{\pm}} \cup (2\pi i)^{\frac{\deg_0}{2}} \operatorname{inv}^* \alpha)$ and $k \in \mathbb{N}$ is as in Theorem 6.1. The relationship (6.8) between the *H*-function and the *I*-function implies part (1) of Theorem 6.1:

$$I_{+} = \mathbb{U}I_{-}.\tag{6.21}$$

Since the *I*-function contains neither $\log z$ nor non-integral powers of z, it follows that \mathbb{U} is in fact a linear transformation:

$$\mathbb{U} \colon H^{\bullet}_{\mathrm{CR},T}(X_{-}) \otimes_{R_{T}} S_{T}((z^{-1})) \to H^{\bullet}_{\mathrm{CR},T}(X_{+}) \otimes_{R_{T}} S_{T}((z^{-1}))$$

Diagram (6.20) gives that \mathbb{U} is automatically degree-preserving. We show that \mathbb{U} satisfies part (2) of Theorem 6.1. Noting that $\mathbf{p}_i^+ = \mathbf{p}_i^-$, $i = 1, \ldots, r-1$ are on the wall W, it suffices to show that $\theta_+(\mathbf{p}_i^+) \circ \mathbb{U} = \mathbb{U} \circ \theta_-(\mathbf{p}_i^-)$ for $1 \leq i \leq r-1$. This follows from equation (6.21) and the monodromy properties of the *I*-functions:

$$\begin{split} I_+ \big|_{\mathbf{y}_j \mapsto e^{2\pi \mathbf{i}} \mathbf{y}_j} &= e^{2\pi \mathbf{i} \theta_+ (\mathbf{p}_j^+)/z} I_+ \\ I_- \big|_{\tilde{\mathbf{y}}_j \mapsto e^{2\pi \mathbf{i}} \tilde{\mathbf{y}}_j} &= e^{2\pi \mathbf{i} \theta_- (\mathbf{p}_j^-)/z} I_- \end{split}$$

for $1 \leq j \leq r-1$. Note that $y_j \to e^{2\pi i} y_j$ corresponds to $\tilde{y}_j \to e^{2\pi i} \tilde{y}_j$ under the change (5.8) of variables. It remains to show that:

- U is symplectic;
- U is defined over $R_T((z^{-1}))$, i.e. that U admits a non-equivariant limit.

These properties follow from the identification of \mathbb{U}_H with the Fourier–Mukai transformation defined in the next section. We will discuss these points in §6.5 below.

6.3. The Fourier–Mukai transform

We now construct a diagram (1.1) canonically associated to the toric birational transformation $\varphi: X_+ \dashrightarrow X_-$, where \widetilde{X} is a toric Deligne–Mumford stack and f_+ , f_- are toric blow-ups, and compute the Fourier–Mukai transformation:

$$\mathbb{FM}: K_T^0(X_-) \to K_T^0(X_+) \qquad \mathbb{FM}:= (f_+)_*(f_-)^*$$

In §6.4 below we will see that this transformation coincides, via the equivariant integral structure in Definition 3.1, with the transformation \mathbb{U} from §6.2.4 given by analytic continuation.

6.3.1. The common blow-up of X_+ and X_-

Recall from §4.2 that X_+ and X_- are defined in terms of an exact sequence:

$$0 \longrightarrow \mathbb{L} \longrightarrow \mathbb{Z}^m \xrightarrow{\beta} \mathbf{N} \longrightarrow 0$$

where the map $\mathbb{L} \to \mathbb{Z}^m$ is given by (D_1, \ldots, D_m) . This sequence defines an action of $K = (\mathbb{C}^{\times})^r$ on \mathbb{C}^m , and $X_{\pm} = [U_{\omega_{\pm}}/K]$ for appropriate stability conditions $\omega_+, \omega_- \in \mathbb{L}^{\vee} \otimes \mathbb{R}$. Let b_1, \ldots, b_m denote the images of the standard basis elements for \mathbb{Z}^m under the map β . Consider now the action of $K \times \mathbb{C}^{\times}$ on \mathbb{C}^{m+1} defined by the exact sequence:

$$0 \longrightarrow \mathbb{L} \oplus \mathbb{Z} \longrightarrow \mathbb{Z}^m \oplus \mathbb{Z} \xrightarrow{\tilde{\beta}} \mathbf{N} \longrightarrow 0$$

where the map $\mathbb{L} \oplus \mathbb{Z} \to \mathbb{Z}^m \oplus \mathbb{Z}$ is given by $(\widetilde{D}_1, \ldots, \widetilde{D}_{m+1})$,

$$\widetilde{D}_j = \begin{cases} D_j \oplus 0 & \text{if } j < m+1 \text{ and } D_j \cdot e \leq 0\\ D_j \oplus (-D_j \cdot e) & \text{if } j < m+1 \text{ and } D_j \cdot e > 0\\ 0 \oplus 1 & \text{if } j = m+1 \end{cases}$$

The map $\tilde{\beta}$ is the direct sum of β with the map $\mathbb{Z} \to \mathbf{N}$ defined by the element

$$b_{m+1} = \sum_{j:D_j \cdot e > 0} (D_j \cdot e) b_j$$

so the images of the standard basis elements for $\mathbb{Z}^m \oplus \mathbb{Z}$ under the map $\tilde{\beta}$ are b_1, \ldots, b_{m+1} . Consider the chambers \tilde{C}_+ , \tilde{C}_- , and \tilde{C} in $(\mathbb{L} \oplus \mathbb{Z})^{\vee} \otimes \mathbb{R}$ that contain, respectively, the stability conditions

$$\tilde{\omega}_{+} = (\omega_{+}, 1)$$
 $\tilde{\omega}_{-} = (\omega_{-}, 1)$ and $\tilde{\omega} = (\omega_{0}, -\varepsilon)$

where ω_0 is a point in the relative interior of $W \cap \overline{C_+} = W \cap \overline{C_-}$ as in §5.1, and ε is a very small positive real number. Let \widetilde{X} denote the toric Deligne–Mumford stack defined by the stability condition $\tilde{\omega}$.

Lemma 6.15. Recall the notation \mathcal{A}_{\pm} , \mathcal{A}_0 , $\mathcal{A}_0^{\text{thick}}$, $\mathcal{A}_0^{\text{thin}}$, M_0 , M_{\pm} in Lemma 5.2. The set of anticones for the stability conditions $\tilde{\omega}_{\pm}$, $\tilde{\omega}$ are given by

$$\mathcal{A}_{\tilde{\omega}_{\pm}} = \{ I \sqcup \{ m+1 \} : I \in \mathcal{A}_{\pm} \}$$
$$\mathcal{A}_{\tilde{\omega}} = \{ I \sqcup \{ m+1 \} : I \in \mathcal{A}_{0}^{\text{thick}} \} \sqcup \{ I \in \mathcal{A}_{0}^{\text{thick}} : I \cap M_{0} \in \mathcal{A}_{0}^{\text{thin}} \}$$

Proof. Straightforward. \Box

Lemma 6.16. We have the following statements.

- (1) The toric Deligne–Mumford stack corresponding to the chamber \widetilde{C}_+ is X_+ .
- (2) The toric Deligne–Mumford stack corresponding to the chamber \widetilde{C}_{-} is X_{-} .
- (3) There is a commutative diagram as in (1.1), where:
 - f₊: X̃ → X₊ is a toric blow-up, arising from wall-crossing from the chamber C̃ to C̃₊; and
 - $f_-: \widetilde{X} \to X_-$ is a toric blow-up, arising from wall-crossing from the chamber \widetilde{C} to \widetilde{C}_- .

Proof. In view of §4.1, the description of $\mathcal{A}_{\tilde{\omega}_{\pm}}$ in Lemma 6.15 proves (1) and (2). The birational transformations $f_+: \tilde{X} \dashrightarrow X_+$ and $f_-: \tilde{X} \dashrightarrow X_-$ determined by the toric wall-crossings are each morphisms which contract the toric divisor defined by the (m + 1)-st homogeneous co-ordinate. Indeed, f_+ is induced by the identity birational map $U_{\tilde{\omega}} \dashrightarrow U_{\tilde{\omega}_+}$, and a point $(z_1, \ldots, z_m, z_{m+1}) \in U_{\tilde{\omega}_+}$ is equivalent to the point $(z_1 z_{m+1}^{l_1}, \ldots, z_m z_{m+1}^{l_m}, 1) \in U_{\omega_+} \times \{1\}$ under the action of the \mathbb{C}^{\times} -subgroup of $K \times \mathbb{C}^{\times}$ corresponding to $e \oplus 1 \in \mathbb{L} \oplus \mathbb{Z}$, where we set $l_i := \max(-D_i \cdot e, 0)$ for $1 \leq i \leq m$. Therefore f_+ is induced by a morphism

$$U_{\tilde{\omega}} \to U_{\omega_{+}} \qquad (z_{1}, \dots, z_{m}, z_{m+1}) \mapsto (z_{1} z_{m+1}^{l_{1}}, \dots, z_{m} z_{m+1}^{l_{m}}) \tag{6.22}$$

which is equivariant with respect to the group homomorphism (quotient by the \mathbb{C}^{\times} -subgroup given by $e \oplus 1$)

$$\phi_+ \colon K \times \mathbb{C}^{\times} \to K \qquad (g, \lambda) \mapsto g \cdot \lambda^{-e}. \tag{6.23}$$

Using Lemma 6.15, one can easily check that the map (6.22) indeed sends $U_{\tilde{\omega}}$ to U_{ω_+} . We obtain a similar description for f_- by considering the \mathbb{C}^{\times} -subgroup given by $0\oplus 1 \in \mathbb{L} \oplus \mathbb{Z}$ instead of $e \oplus 1$. \Box

Remark 6.17. Torus fixed points on \widetilde{X} lying on the exceptional divisor $\{z_{m+1} = 0\}$ of f_{\pm} correspond to minimal anticones $\widetilde{\delta} \in \mathcal{A}_{\widetilde{\omega}}$ such that $\widetilde{\delta} \in \mathcal{A}_{0}^{\text{thick}}$ and $\widetilde{\delta} \cap M_{0} \in \mathcal{A}_{0}^{\text{thin}}$. These minimal anticones take the form

$$\tilde{\delta} = \{j_1, \dots, j_{r-1}, j_+, j_-\}$$

where $D_{j_1}, \ldots, D_{j_{r-1}} \in W$, $D_{j_+} \cdot e > 0$ and $D_{j_-} \cdot e < 0$. The birational morphism f_{\pm} maps the corresponding torus fixed point $x_{\delta} \in \widetilde{X}$ to the torus fixed point $x_{\delta_{\pm}} \in X_{\pm}$ with

$$\delta_+ = \{j_1, \dots, j_{r-1}, j_+\} \in \mathcal{A}_+, \quad \delta_- = \{j_1, \dots, j_{r-1}, j_-\} \in \mathcal{A}_-.$$

Torus fixed points on \widetilde{X} lying away from the exceptional divisor $\{z_{m+1} = 0\}$ corresponds to minimal anticones $\widetilde{\delta} \in \mathcal{A}_{\widetilde{\omega}}$ of the form $\widetilde{\delta} = \delta \cup \{m+1\}, \ \delta \in \mathcal{A}_0^{\text{thick}} = \mathcal{A}_+ \cap \mathcal{A}_-$. The morphisms f_{\pm} are isomorphisms in neighbourhoods of these fixed points, and the torus fixed point $x_{\widetilde{\delta}}$ maps to the fixed point x_{δ} in X_+ or in X_- .

Remark 6.18. The stacky fan $\widetilde{\Sigma}$ for \widetilde{X} is obtained from the stacky fans Σ_{\pm} for X_{\pm} by adding the extra ray $b_{m+1} = \sum_{j:D_j \cdot e > 0} (D_j \cdot e) b_j$ where

$$\sum_{j:D_j \cdot e > 0} (D_j \cdot e)b_j = \sum_{j:D_j \cdot e < 0} (-D_j \cdot e)b_j$$

is a minimal linear relation (or circuit) in Σ_{\pm} , see Remark 5.3. So our discussion here is a rephrasing in terms of GIT data of the material in [12, §5].

6.3.2. A basis for localized T-equivariant K-theory

Recall that $T = (\mathbb{C}^{\times})^m$ acts on X_{\pm} through the diagonal *T*-action on \mathbb{C}^m . We consider the *T*-action on \widetilde{X} induced from the inclusion $T = T \times \{1\} \subset T \times \mathbb{C}^{\times}$ and the $(T \times \mathbb{C}^{\times})$ -action on \mathbb{C}^{m+1} . Then all the maps in the diagram (1.1) are *T*-equivariant. The *T*-equivariant *K*-groups $K_T^0(X_{\pm})$, $K_T^0(\widetilde{X})$ are modules over $K_T^0(\text{pt})$, which is the ring $\mathbb{Z}[T]$ of regular functions (over \mathbb{Z}) on the algebraic torus *T*.

The *T*-invariant divisor $\{z_i = 0\}$ on X_{ω} defined in (4.6) determines a *T*-equivariant line bundle $\mathcal{O}(\{z_i = 0\})$ on X_{ω} , and we denote the class of this line bundle in *T*-equivariant *K*-theory by R_i . For the spaces X_+ , X_- , and \widetilde{X} we write these classes as

$$R_1^+, \dots, R_m^+ \in K_T^0(X_+) \qquad R_1^-, \dots, R_m^- \in K_T^0(X_-) \quad \text{and} \quad \widetilde{R}_1, \dots, \widetilde{R}_{m+1} \in K_T^0(\widetilde{X}).$$

Let us write:

$$S_j^+ := (R_j^+)^{-1} \qquad S_j^- := (R_j^-)^{-1} \qquad \text{and} \qquad \widetilde{S}_j := \widetilde{R}_j^{-1}$$

An irreducible K-representation $p \in \text{Hom}(K, \mathbb{C}^{\times}) = \mathbb{L}^{\vee}$ defines a line bundle $L(p) \to X_{\omega}$:

$$L(p) = U_{\omega} \times \mathbb{C}/(z, v) \sim (g \cdot z, p(g)v), \ g \in K$$

This line bundle is equipped with the *T*-linearization $[z, v] \mapsto [t \cdot z, v], t \in T$ and thus defines a class in $K_T^0(X_\omega)$. We write $L_+(p)$ for the corresponding line bundle on X_+ and $L_-(p)$ for the corresponding line bundle on X_- . We have

$$R_i^{\pm} = L_{\pm}(D_i) \otimes e^{\lambda_i}$$

where $e^{\lambda_i} \in \mathbb{C}[T]$ stands for the irreducible *T*-representation given by the *i*th projection $T \to \mathbb{C}^{\times}$. In particular we have $c_1^T(L_{\pm}(p)) = \theta_{\pm}(p)$ for the map θ in (4.8). Similarly

a character $(p,n) \in \text{Hom}(K \times \mathbb{C}^{\times}, \mathbb{C}^{\times}) = \mathbb{L}^{\vee} \oplus \mathbb{Z}$ defines a *T*-equivariant line bundle $L(p,n) \to \widetilde{X}$ and we have:

$$\widetilde{R}_i = L(\widetilde{D}_i) \otimes e^{\lambda_i} \qquad 1 \le i \le m$$
$$\widetilde{R}_{m+1} = L(\widetilde{D}_{m+1}) = L(0, 1)$$

The classes $L_{\pm}(p)$ (respectively the classes L(p, n)) generate the equivariant K-group $K_T^0(X_{\pm})$ (respectively $K_T^0(\widetilde{X})$) over $\mathbb{Z}[T]$.

Let $\delta_{-} \in \mathcal{A}_{-}$ be a minimal anticone, $x_{\delta_{-}}$ be the corresponding *T*-fixed point on X_{-} , $i_{\delta_{-}} : x_{\delta_{-}} \to X_{-}$ be the inclusion of the fixed point, and $G_{\delta_{-}}$ be the isotropy group of $x_{\delta_{-}}$. We have that $x_{\delta_{-}} \cong BG_{\delta_{-}}$, and that $i_{\delta_{-}}^{\star}R_{i} = 1$ for $i \in \delta_{-}$. A basis for $K_{T}^{0}(X_{-})$, after inverting non-zero elements of $\mathbb{Z}[T]$, is given by:

$$\{(i_{\delta_{-}})_{\star}\varrho: \varrho \text{ an irreducible representation of } G_{\delta_{-}}, \, \delta_{-} \in \mathcal{A}_{-}\}$$
(6.24)

We need to specify a *T*-linearization on $(i_{\delta_{-}})_{\star}\varrho$. Choosing a lift $\hat{\varrho} \in \text{Hom}(K, \mathbb{C}^{\times}) = \mathbb{L}^{\vee}$ of each $G_{\delta_{-}}$ -representation $\varrho \colon G_{\delta_{-}} \to \mathbb{C}^{\times}$, we write any element in (6.24) in the form:

$$e_{\delta_{-},\varrho} := L_{-}(\hat{\varrho}) \prod_{i \notin \delta_{-}} \left(1 - S_{i}^{-}\right)$$

Then $\{e_{\delta_{-},\varrho}\}$ is a basis for the localized *T*-equivariant *K*-theory of X_{-} . There is an entirely analogous basis $\{e_{\delta_{+},\varrho}\}$ for the localized *T*-equivariant *K*-theory of X_{+} . We will describe the action of the Fourier–Mukai transform in terms of these bases.

6.3.3. Computing the Fourier–Mukai transform

Consider the diagram (1.1) and the associated Fourier–Mukai transform \mathbb{FM} : $K_T^0(X_-) \to K_T^0(X_+)$. In this section we prove:

Theorem 6.19. If $\delta_{-} \in \mathcal{A}_{-}$ is a minimal anticone such that $\delta_{-} \in \mathcal{A}_{+}$ then

$$\mathbb{FM}(e_{\delta_{-},\varrho}) = e_{\delta_{-},\varrho}$$

where on the left-hand side of the equality δ_{-} is regarded as a minimal anticone for X_{-} and on the right-hand side δ_{-} is regarded as a minimal anticone for X_{+} . If δ_{-} is a minimal anticone in \mathcal{A}_{-} such that $\delta_{-} \notin \mathcal{A}_{+}$ then $\mathbb{FM}(e_{\delta_{-},o})$ is equal to

$$\frac{1}{l} \sum_{t \in \mathcal{T}} \left(\frac{1 - S_{j_{-}}^{+}}{1 - t^{-1}} \cdot L_{+}(\hat{\varrho}) t^{\hat{\varrho} \cdot e} \cdot \prod_{\substack{j \notin \delta_{-} \\ D_{j} \cdot e < 0}} (1 - S_{j}^{+}) \cdot \prod_{\substack{i \notin \delta_{-} \\ D_{i} \cdot e \ge 0}} (1 - t^{-D_{i} \cdot e} S_{i}^{+}) \right)$$

where j_{-} is the unique element of δ_{-} such that $D_{j_{-}} \cdot e < 0, \ l = -D_{j_{-}} \cdot e$ and

$$\mathcal{T} = \left\{ \zeta \cdot (R_{j_{-}}^{+})^{1/l} : \zeta \in \boldsymbol{\mu}_{l} \right\}.$$

Remark 6.20. We have

$$\frac{1}{l}\sum_{t\in\mathcal{T}}t^n = \begin{cases} (R_{j_-}^+)^{n/l} & \text{if } l \text{ divides } n;\\ 0 & \text{otherwise.} \end{cases}$$

Thus $\frac{1}{l} \sum_{t \in \mathcal{T}} f(t)$ makes sense as an element of $K_T^0(X_+)$ for a Laurent polynomial f(t) in t. Note that each summand appearing in the formula for $\mathbb{FM}(e_{\delta_{-},\varrho})$ is in fact a Laurent polynomial in t, since the factor $1 - t^{-1}$ divides $1 - S_{i_-}^+ = 1 - t^{-l}$.

Borisov–Horja have computed how non-equivariant versions of the classes R_i^- change under pullback [13, Proposition 8.1]. We have parallel results in the equivariant setting.

Proposition 6.21. For $p \in \mathbb{L}$, we have:

$$f^{\star}_{-}(L_{-}(p)) = L(p,0)$$
 and $f^{\star}_{+}(L_{+}(p)) = L(p,-p \cdot e)$

Let $k_i := \max(D_i \cdot e, 0)$ and $l_i := \max(-D_i \cdot e, 0)$. Then:

$$f_{-}^{\star}R_{i}^{-} = \widetilde{R}_{i}\widetilde{R}_{m+1}^{k_{i}} \qquad and \qquad f_{+}^{\star}R_{i}^{+} = \widetilde{R}_{i}\widetilde{R}_{m+1}^{l_{i}}.$$

Proof. These statements follow from the description of $f_{\pm} : \widetilde{X} \to X_{\pm}$ in the proof of Lemma 6.16; see (6.22) and (6.23). \Box

We now analyze the push-forward of classes supported on torus fixed points of X.

Proposition 6.22. Consider minimal anticones

$$\widetilde{\delta} = \{j_1, \dots, j_{r-1}, j_+, j_-\} \in \mathcal{A}_{\widetilde{\omega}} \qquad for \ \widetilde{X}$$
$$\delta_+ = \{j_1, \dots, j_{r-1}, j_+\} \in \mathcal{A}_+ \qquad for \ X_+$$

such that $\{j_1, \ldots, j_{r-1}, j_+, j_-\} \subset \{1, \ldots, m\}, D_{j_1} \cdot e = \cdots = D_{j_{r-1}} \cdot e = 0, D_{j_-} \cdot e < 0$ and $D_{j_+} \cdot e > 0$. Let $i_{\tilde{\delta}} \colon BG_{\tilde{\delta}} \to \tilde{X}$ and $i_{\delta_+} \colon BG_{\delta_+} \to X_+$ denote the inclusions of the corresponding T-fixed points and let $f_{+,\tilde{\delta}} \colon BG_{\tilde{\delta}} \to BG_{\delta_+}$ denote the map induced on the fixed points.

(1) The map $f_{+,\tilde{\delta}}$ exhibits $BG_{\tilde{\delta}}$ as a μ_l -gerbe over BG_{δ_+} , where $l = -D_{j_-} \cdot e$.

(2) We have:

$$(f_{+,\tilde{\delta}})_{\star}(i_{\tilde{\delta}})^{\star}L(p,n) = \begin{cases} (i_{\delta_{+}})^{\star}L_{+}(p)(R_{j_{-}}^{+})^{(p\cdot e+n)/l} & \text{if } l \text{ divides } p \cdot e+n; \\ 0 & \text{otherwise.} \end{cases}$$

(3) Let g be a Laurent polynomial in m + 1 variables. Then:

$$(f_{+,\tilde{\delta}})_{\star}(i_{\tilde{\delta}})^{\star}L(p,n)g\big(\widetilde{R}_{1},\ldots,\widetilde{R}_{m+1}\big)$$

= $(i_{\delta_{+}})^{\star}\frac{1}{l}\sum_{t\in\mathcal{T}}L_{+}(p)t^{p\cdot e+n}g\big(t^{-l_{1}}R_{1}^{+},\ldots,t^{-l_{m}}R_{m}^{+},t\big)$

Proof. The stabilizers G_{δ} and G_{δ_+} are given, as subgroups of $K \times \mathbb{C}^{\times}$ and K, by

$$G_{\bar{\delta}} = \{ (g,\lambda) \in K \times \mathbb{C}^{\times} : D_j(g)\lambda^{-D_j \cdot e} = 1 \text{ for all } j \in \delta_+, D_{j_-}(g) = 1 \}$$

$$G_{\delta_+} = \{ h \in K : D_j(h) = 1 \text{ for all } j \in \delta_+ \}$$

where we regard D_j as a character of K. The homomorphism $G_{\tilde{\delta}} \to G_{\delta_+}$ is induced by $\phi_+: (g, \lambda) \mapsto h = g \cdot \lambda^{-e}$ in (6.23). The kernel of the homomorphism is $\{(\lambda^e, \lambda) : \lambda \in \mu_l\}$ and we obtain an exact sequence:

$$1 \longrightarrow \boldsymbol{\mu}_l \longrightarrow G_{\tilde{\delta}} \longrightarrow G_{\delta_+} \longrightarrow 1$$

Therefore $f_{+,\tilde{\delta}}$ exhibits $BG_{\tilde{\delta}}$ as a μ_l -gerbe over BG_{δ_+} .

For part (2), notice that $(f_{+,\tilde{\delta}})_{\star}$ maps a $G_{\tilde{\delta}}$ -representation to its μ_l -invariant part. The character $(p,n) \in \operatorname{Hom}(K \times \mathbb{C}^{\times}, \mathbb{C}^{\times})$ induces a μ_l -character $\lambda \mapsto \lambda^{p \cdot e + n}$ via the inclusion $\mu_l \subset G_{\tilde{\delta}} \subset K \times \mathbb{C}^{\times}$. Therefore $(f_{+,\tilde{\delta}})_{\star}(\iota_{\tilde{\delta}})^{\star}L(p,n)$ vanishes if l does not divide $p \cdot e + n$. On the other hand, Proposition 6.21 gives $(f_+)^{\star}R_{j_-}^+ = \widetilde{R}_{j_-}\widetilde{R}_{m+1}^l$ and hence, if l divides $p \cdot e + n$,

$$(f_{+,\tilde{\delta}})^{*}(i_{\delta_{+}})^{*}L_{+}(p)(R_{j_{-}}^{+})^{(p\cdot e+n)/l} = (i_{\tilde{\delta}})^{*}L(p,-p\cdot e)(\widetilde{R}_{j_{-}})^{(p\cdot e+n)/l}(\widetilde{R}_{m+1})^{p\cdot e+n}$$
$$= (i_{\tilde{\delta}})^{*}L(p,-p\cdot e)(\widetilde{R}_{m+1})^{p\cdot e+n} = (i_{\tilde{\delta}})^{*}L(p,n).$$

Therefore the Projection Formula gives $(f_{+,\tilde{\delta}})_{\star}(i_{\tilde{\delta}})^{\star}L(p,n) = (i_{\delta_{+}})^{\star}L_{+}(p)(R_{j_{-}}^{+})^{(p\cdot e+n)/l}$. This proves (2).

For part (3) it suffices to take g to be a monomial: $g(\widetilde{R}_1, \ldots, \widetilde{R}_{m+1}) = \prod_{i=1}^{m+1} \widetilde{R}_i^{n_i}$. In this case:

$$L(p,n)g(\widetilde{R}_{1},\ldots,\widetilde{R}_{m+1}) = L(p + \sum_{i=1}^{m} n_{i}D_{i}, n + n_{m+1} - \sum_{i=1}^{m} n_{i}k_{i}) \otimes e^{\sum_{i=1}^{m} n_{i}\lambda_{i}}$$
(6.25)

Part (2) can be restated as:

$$(f_{+,\tilde{\delta}})_{\star}(i_{\tilde{\delta}})^{\star}L(p,n) = (i_{\delta_{+}})^{\star}\frac{1}{l}\sum_{t\in\mathcal{T}}L_{+}(p)t^{p\cdot e+n}$$

Combining this with (6.25) yields (3). \Box

Proof of Theorem 6.19. Suppose first that $\delta_{-} \in \mathcal{A}_{+} \cap \mathcal{A}_{-}$. Then, as discussed, φ gives an isomorphism between neighbourhoods of the fixed points corresponding to δ_{-} . Thus $\mathbb{FM}(e_{\delta_{-},\varrho}) = e_{\delta_{-},\varrho}$.

Suppose now that $\delta_{-} \in \mathcal{A}_{-}$ but $\delta_{-} \notin \mathcal{A}_{+}$, so that $\delta_{-} = \{j_{1}, \ldots, j_{r-1}, j_{-}\}$ with $D_{j_{1}} \cdot e = \cdots = D_{j_{r-1}} \cdot e = 0$ and $D_{j_{-}} \cdot e < 0$. Proposition 6.21 gives:

$$(f_{-})^{\star}e_{\delta_{-},\varrho} = L(\hat{\varrho},0)\prod_{i\notin\delta_{-}} \left(1-\widetilde{S}_{m+1}^{k_{i}}\widetilde{S}_{i}\right)$$

where the index i in the product satisfies $i \leq m$. This restricts to zero at a fixed point $x_{\tilde{\delta}} \in \tilde{X}$ unless $x_{\tilde{\delta}}$ is in $f_{+}^{-1}(x_{\delta_{-}})$, that is, unless $\tilde{\delta}$ has the form $\delta_{-} \cup \{j_{+}\}$ with $D_{j_{+}} \cdot e > 0$. The Localization Theorem in *T*-equivariant *K*-theory [32] gives:

$$(f_{-})^{\star}e_{\delta_{-},\varrho} = \sum_{\tilde{\delta}} (i_{\tilde{\delta}})_{\star} (i_{\tilde{\delta}})^{\star} \left[\frac{L(\hat{\varrho},0)\prod_{i\notin\delta_{-}} \left(1-\widetilde{S}_{m+1}^{k_{i}}\widetilde{S}_{i}\right)}{(1-\widetilde{S}_{m+1})\prod_{j\notin\delta_{-},j\neq j_{+}} (1-\widetilde{S}_{j})} \right]$$
(6.26)

where $i, j \leq m$ and the sum runs over $\tilde{\delta} = \delta_- \cup \{j_+\}$ such that $D_{j_+} \cdot e > 0$. Restricted to such a *T*-fixed point, \tilde{S}_{j_+} becomes trivial, so the numerator in (6.26) contains a factor $(i_{\tilde{\delta}})^*(1-\tilde{S}_{m+1}^{k_{j_+}})$ that is divisible by $(i_{\tilde{\delta}})^*(1-\tilde{S}_{m+1})$. Thus (6.26) depends polynomially on \tilde{S}_{m+1} . Now:

$$(f_{+})_{\star}(f_{-})^{\star}e_{\delta_{-},\varrho} = \sum_{\delta_{+}:\delta_{+}|\delta_{-}} (i_{\delta_{+}})_{\star}(f_{+,\tilde{\delta}})_{\star}(i_{\tilde{\delta}})^{\star} \left[\frac{L(\hat{\varrho},0)\prod_{i\notin\delta_{-}}\left(1-\widetilde{S}_{m+1}^{k_{i}}\widetilde{S}_{i}\right)}{(1-\widetilde{S}_{m+1})\prod_{j\notin\delta_{-},j\neq j_{+}}(1-\widetilde{S}_{j})} \right]$$
$$= \sum_{\delta_{+}:\delta_{+}|\delta_{-}} (i_{\delta_{+}})_{\star}(i_{\delta_{+}})^{\star} \left[\frac{1}{l} \sum_{t\in\mathcal{T}} \frac{L_{+}(\hat{\varrho})t^{\hat{\varrho}\cdot e}\prod_{i\notin\delta_{-}}\left(1-t^{l_{i}-k_{i}}S_{i}^{+}\right)}{(1-t^{-1})\prod_{j\notin\delta_{-},j\neq j_{+}}\left(1-t^{l_{j}}S_{j}^{+}\right)} \right]$$

where we used part (3) of Proposition 6.22. This is:

$$\sum_{\delta_{+}:\delta_{+}|\delta_{-}} (i_{\delta_{+}})_{\star} (i_{\delta_{+}})^{\star} \left[\frac{\frac{1}{l} \sum_{t \in \mathcal{T}} \frac{1-S_{j_{-}}^{+}}{1-t^{-1}} \cdot L_{+}(\hat{\varrho}) t^{\hat{\varrho} \cdot e} \cdot \prod_{j \notin \delta_{-}} \left(1-t^{-k_{j}} S_{j}^{+}\right)}{\prod_{j \notin \delta_{+}} (1-S_{j}^{+})} \right]$$

Applying the Localization Theorem again gives the result. Here we need to check that the restriction of

$$\frac{1-S_{j_{-}}^{+}}{1-t^{-1}} \cdot \prod_{j \notin \delta_{-}} \left(1-t^{-k_{j}}S_{j}^{+}\right)$$

to the fixed point corresponding to $\delta \in \mathcal{A}_+ \cap \mathcal{A}_-$ vanishes. If there exists $j \in \delta$ with $j \notin \delta_-$ and $D_j \cdot e \leq 0$ then the restriction vanishes since $i^*_{\delta}S^+_j = 1$. Otherwise one has

 $\delta \setminus \delta_{-} \subset M_{+}$. In this case $j_{-} \in \delta$ and there exists $j_{0} \in \delta \cap M_{+}$. Thus the restriction contains the factor

$$i_{\delta}^{\star} \left[\frac{1 - S_{j_{-}}^{+}}{1 - t^{-1}} (1 - t^{-D_{j_{0}} \cdot e} S_{j_{0}}^{+}) \right] = i_{\delta}^{\star} \left[(1 - S_{j_{-}}^{+}) \frac{1 - t^{-D_{j_{0}} \cdot e}}{1 - t^{-1}} \right]$$

which vanishes. \Box

6.4. The Fourier–Mukai transform matches with analytic continuation

We now show that the analytic continuation formula in Theorem 6.13 matches with the Fourier–Mukai transform in Theorem 6.19. More precisely we show:

Theorem 6.23. Let \mathbb{U}_H be the linear transformation in §6.2.4 given by the analytic continuation of H-functions. Then \mathbb{U}_H induces a map $\mathbb{U}_H: H_T^{\bullet\bullet}(IX_-) \to H_T^{\bullet\bullet}(IX_+)$ and the following diagram commutes:

$$K_T^0(X_-) \xrightarrow{\mathbb{FM}} K_T^0(X_+)$$

$$\stackrel{\widetilde{ch}}{\to} \bigvee_{T} (IX_-) \xrightarrow{\mathbb{U}_H} H_T^{\bullet\bullet}(IX_+) \qquad (6.27)$$

We start by computing the Chern characters of certain line bundles. It is easy to see that:

$$\begin{split} \widetilde{\mathrm{ch}}(L_{\pm}(\hat{\varrho})) &= \bigoplus_{f \in \mathbb{K}_{\pm}/\mathbb{L}} e^{2\pi \mathrm{i}\hat{\varrho} \cdot f} e^{\theta_{\pm}(\hat{\varrho})} \mathbf{1}_{f} \\ \widetilde{\mathrm{ch}}(S_{j}^{\pm}) &= \bigoplus_{f \in \mathbb{K}_{\pm}/\mathbb{L}} e^{-2\pi \mathrm{i}D_{j} \cdot f} e^{-u_{j}} \mathbf{1}_{f} \end{split}$$

In view of this, we define

$$\widetilde{\mathrm{ch}}(t) := \bigoplus_{f \in \mathbb{K}_+/\mathbb{L}} \zeta e^{2\pi \mathrm{i} D_{j_-} \cdot f/l} e^{u_{j_-}/l} \mathbf{1}_f$$

for $t = \zeta(R_{j_-}^+)^{1/l} \in \mathcal{T}$ appearing in Theorem 6.19. Here we fix lifts $\mathbb{K}_+/\mathbb{L} \to \mathbb{K}_+$, $\mathbb{K}_-/\mathbb{L} \to \mathbb{K}_-$ as in Notation 6.10 and identify $f \in \mathbb{K}_+/\mathbb{L}$ with its lift in \mathbb{K}_+ .

Lemma 6.24. Suppose that (δ_+, f_+) indexes a *T*-fixed point on X_+ , that (δ_-, f_-) indexes a *T*-fixed point on X_- , and that $(\delta_+, f_+)|(\delta_-, f_-)$. Let $j_- \in \delta_-$ be the unique index such that $D_{j_-} \cdot e < 0$ and write $l = -D_{j_-} \cdot e$. Setting $t = e^{-2\pi i D_{j_-} \cdot f_-/l} (R_{j_-}^+)^{1/l}$, we have:

T. Coates et al. / Advances in Mathematics 329 (2018) 1002-1087

$$i_{(\delta_{+},f_{+})}^{\star} \widetilde{\operatorname{Ch}} \left(L_{+}(\hat{\varrho}) t^{\hat{\varrho} \cdot e} \right) = i_{(\delta_{-},f_{-})}^{\star} \widetilde{\operatorname{Ch}} \left(L_{-}(\hat{\varrho}) \right)$$

$$i_{(\delta_{+},f_{+})}^{\star} \widetilde{\operatorname{Ch}} \left(S_{j}^{+} t^{-D_{j} \cdot e} \right) = i_{(\delta_{-},f_{-})}^{\star} \widetilde{\operatorname{Ch}} \left(S_{j}^{-} \right)$$
(6.28)

We also have:

$$i_{(\delta_+,f_+)}^{\star} \widetilde{\mathrm{ch}} \left[L_+(\hat{\varrho}) t^{\hat{\varrho} \cdot e} \prod_{j \notin \delta_-} (1 - t^{-D_j \cdot e} S_j^+) \right] = i_{(\delta_-,f_-)}^{\star} \widetilde{\mathrm{ch}}(e_{\delta_-,\varrho}) \tag{6.29}$$

and:

$$i_{(\delta_{+},f_{+})}^{\star} \widetilde{\mathrm{ch}} \left[\frac{1 - S_{j_{-}}^{+}}{l(1 - t^{-1})} \cdot \prod_{\substack{j \notin \delta_{-} \\ D_{j} \cdot e < 0}} \frac{1 - S_{j}^{+}}{1 - S_{j}^{+} t^{-D_{j} \cdot e}} \right] = C_{(\delta_{+},f_{+})}^{(\delta_{-},f_{-})}$$
(6.30)

where $C_{(\delta_+,f_+)}^{(\delta_-,f_-)}$ are the coefficients appearing in Theorem 6.13.

Proof. This is just a calculation. Recall from Notation 6.10 that $f_- = f_+ + \alpha e$ for some $\alpha \in \mathbb{Q}$. Then $D_j \cdot (f_+ - f_-) = -\alpha D_j \cdot e$ and $D_{j_-} \cdot (f_+ - f_-) = l\alpha$. The formulae (6.28) easily follow from Lemma 6.11 and (6.11). The formula (6.29) is an easy consequence of (6.28). To see (6.30), we calculate, using (6.28),

$$\begin{split} \text{LHS} &= \frac{1}{l} \frac{1 - e^{-u_{j_{-}}(\delta_{+}) - 2\pi i (D_{j_{-}} \cdot f_{+})}}{1 - e^{-\frac{1}{l}(u_{j_{-}}(\delta_{+}) + 2\pi i D_{j_{-}} \cdot (f_{+} - f_{-}))}} \prod_{\substack{j \notin \delta_{-} \\ D_{j} \cdot e < 0}} \frac{1 - e^{-u_{j}(\delta_{+}) - 2\pi i D_{j} \cdot f_{+}}}{1 - e^{-u_{j}(\delta_{-}) - 2\pi i D_{j} \cdot f_{-}}} \\ &= \frac{1}{l} \frac{\sin \pi \left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i} + D_{j_{-}} \cdot (f_{+} - f_{-})\right)}{\sin \frac{\pi}{l} \left(\frac{u_{j_{-}}(\delta_{+})}{2\pi i} + D_{j_{-}} \cdot (f_{+} - f_{-})\right)} \prod_{\substack{j \notin \delta_{-} \\ D_{j} \cdot e < 0}} \frac{\sin \pi \left(\frac{u_{j}(\delta_{+})}{2\pi i} + D_{j} \cdot f_{+}\right)}{\sin \pi \left(\frac{u_{j}(\delta_{-})}{2\pi i} + D_{j} \cdot f_{-}\right)} \\ &\times e^{-\frac{1}{2}(1 - \frac{1}{l})(u_{j_{-}}(\delta_{+}) + 2\pi i D_{j_{-}} \cdot (f_{+} - f_{-})) + \sum_{j \notin \delta_{-}, D_{j} \cdot e < 0} \left(\frac{1}{2}(u_{j}(\delta_{-}) - u_{j}(\delta_{+})) + \pi i D_{j} \cdot (f_{-} - f_{+})\right)} \end{split}$$

where we used the fact that $D_{j_-} \cdot f_- \in \mathbb{Z}$. Using Lemma 6.11 again to calculate the exponential factor, we arrive at the expression for $C_{(\delta_+,f_+)}^{(\delta_-,f_-)}$ in Theorem 6.13. \Box

Proof of Theorem 6.23. We first show that the commutative diagram holds over \widehat{S}_T . Then it follows that \mathbb{U}_H has a non-equivariant limit, as \mathbb{FM} does. Consider the element $e_{\delta,\varrho} \in K_T^0(X_-)$ with $\delta \in \mathcal{A}_+ \cap \mathcal{A}_-$. Theorem 6.19 and the definition (6.19) of \mathbb{U}_H show that

$$\widetilde{\mathrm{ch}}(\mathbb{FM}(e_{\delta,\varrho})) = \widetilde{\mathrm{ch}}(e_{\delta,\varrho}) = \mathbb{U}_H(\widetilde{\mathrm{ch}}(e_{\delta,\varrho})).$$

Consider now $e_{\delta_{-},\varrho} \in K^0_T(X_{-})$ for $\delta_{-} \in \mathcal{A}_{-} \setminus \mathcal{A}_{+}$. It is clear that $\widetilde{ch}(\mathbb{FM}(e_{\delta_{-},\varrho}))$ is supported only on fixed points $x_{(\delta_{+},f_{+})} \in IX_{+}$ such that $\delta_{+}|\delta_{-}$. By the definition (6.19) of \mathbb{U}_H , it suffices to show that:
T. Coates et al. / Advances in Mathematics 329 (2018) 1002-1087

$$i_{(\delta_{+},f_{+})}^{\star} \widetilde{\operatorname{ch}}\left(\mathbb{FM}(e_{\delta_{-},\varrho})\right) = \sum_{\substack{f_{-} \in \mathbb{K}_{-}/\mathbb{L}:\\(\delta_{+},f_{+}) \mid (\delta_{-},f_{-})}} C_{\delta_{+},f_{+}}^{\delta_{-},f_{-}} \cdot i_{(\delta_{-},f_{-})}^{\star} \widetilde{\operatorname{ch}}(e_{\delta_{-},\varrho})$$
(6.31)

We may rewrite the result in Theorem 6.19 as

$$\mathbb{FM}(e_{\delta_{-},\varrho}) = \frac{1}{l} \sum_{t \in \mathcal{T}} \left(\frac{1 - S_{j_{-}}^{+}}{1 - t^{-1}} \prod_{\substack{j \notin \delta_{-} \\ D_{j} \cdot e < 0}} \frac{1 - S_{j}^{+}}{1 - t^{-D_{j} \cdot e} S_{j}^{+}} \cdot L_{+}(\hat{\varrho}) t^{\hat{\varrho} \cdot e} \prod_{i \notin \delta_{-}} (1 - t^{-D_{i} \cdot e} S_{i}^{+}) \right)$$

$$(6.32)$$

We have a one-to-one correspondence between the index of summation f_{-} in (6.31) and the index of summation $t \in \mathcal{T}$ in (6.32) given by

$$f_{-} \iff t = e^{-2\pi i D_{j_{-}} \cdot f_{-}/l} (R_{j_{-}}^{+})^{1/l}$$

where $j_{-} \in \delta_{-}$ is the unique element satisfying $D_{j_{-}} \cdot e < 0$ and $l = -D_{j_{-}} \cdot e$. Therefore (6.31) follows from (6.32), (6.29) and (6.30). The Theorem is proved. \Box

6.5. Completing the proof of Theorem 6.1

Combining the commutative diagrams (6.20) and (6.27), we obtain the commutative diagram (6.1) in Theorem 6.1. Since the Fourier–Mukai transformation FM can be defined non-equivariantly, U also admits a non-equivariant limit. Finally we show that U is symplectic, i.e. that $(\mathbb{U}(-z)\alpha, \mathbb{U}(z)\beta) = (\alpha, \beta)$ for all α, β . Since FM is induced by an equivalence of derived categories [32], it preserves the Euler pairing $\chi(E, F)$ given in (3.5). The proof of Proposition 3.2 shows that the vertical maps $\widetilde{\Psi}_{\pm}$ in (6.1) preserve the pairing in the sense that:

$$\left(\widetilde{\Psi}_{\pm}(E)|_{z \to e^{-\pi i}z}, \widetilde{\Psi}_{\pm}(F)\right) = \chi_z(E, F).$$

The commutative diagram (6.1) now shows that \mathbb{U} is symplectic. This completes the proof of Theorem 6.1.

7. Toric complete intersections

We now turn to the Crepant Transformation Conjecture for toric complete intersections. Consider toric Deligne–Mumford stacks X_{\pm} of the form $[\mathbb{C}^m/\!/_{\omega}K]$, where $K = (\mathbb{C}^{\times})^r$ is a complex torus, and consider a K-equivalence $\varphi \colon X_+ \dashrightarrow X_-$ determined by a wall-crossing in the space of stability conditions ω as in §5. We use notation as there, so that $\mathbb{L} = \text{Hom}(\mathbb{C}^{\times}, K)$ is the lattice of cocharacters of K; the space of stability conditions is $\mathbb{L}^{\vee} \otimes \mathbb{R}$; and the birational map φ is induced by the wall-crossing from

a chamber $C_+ \subset \mathbb{L}^{\vee} \otimes \mathbb{R}$ to a chamber $C_- \subset \mathbb{L}^{\vee} \otimes \mathbb{R}$, where C_+ and C_- are separated by a wall W. Consider characters E_1, \ldots, E_k of K such that:

- each E_i lies in $W \cap \overline{C_+} = W \cap \overline{C_-};$
- for each *i*, the line bundle $L_{X_+}(E_i) \to X_+$ corresponding to E_i is a pull-back from the coarse moduli space $|X_+|$;
- for each *i*, the line bundle $L_{X_-}(E_i) \to X_-$ corresponding to E_i is a pull-back from the coarse moduli space $|X_-|$;

where $L_{X_{\pm}}(E_i)$ are the line bundles on X_{\pm} associated to the character E_i in §6.3.2. Let:

$$E_{+} := \bigoplus_{i=1}^{k} L_{X_{+}}(E_{i}) \qquad E_{-} := \bigoplus_{i=1}^{k} L_{X_{-}}(E_{i})$$

Let s_+ , s_- be regular sections of, respectively, the vector bundles $E_+ \to X_+$ and $E_- \to X_-$ such that:

- s_+ and s_- are compatible via $\varphi \colon X_+ \dashrightarrow X_-$;
- the zero loci of s_{\pm} intersect the flopping locus of φ transversely;

and let $Y_+ \subset X_+$, $Y_- \subset X_-$ be the complete intersection substacks defined by s_+ , s_- . The birational transformation φ then induces a K-equivalence $\varphi: Y_+ \dashrightarrow Y_-$. In this section we establish the Crepant Transformation Conjecture for $\varphi: Y_+ \dashrightarrow Y_-$.

7.1. The ambient part of quantum cohomology

Under our standing hypotheses on the ambient toric stacks X_{\pm} , the complete intersections Y_{\pm} automatically have semi-projective coarse moduli spaces, and so the (non-equivariant) quantum products on $H^{\bullet}_{CR}(Y_{\pm})$ are well-defined. Thus we have a welldefined quantum connection

$$\nabla = d + z^{-1} \sum_{i=0}^{N} (\phi_i \star_\tau) d\tau^i$$
(7.2)

(7.1)

where \star_{τ} is the non-equivariant big quantum product, defined exactly as in (2.2). This is a pencil ∇ of flat connections on the trivial $H^{\bullet}_{CR}(Y_{\pm})$ -bundle over an open set in $H^{\bullet}_{CR}(Y_{\pm})$; here, as in the equivariant case, $z \in \mathbb{C}^{\times}$ is the pencil variable, $\tau \in H^{\bullet}_{CR}(Y_{\pm})$ is the co-ordinate on the base of the bundle, ϕ_0, \ldots, ϕ_N are a basis for $H^{\bullet}_{CR}(Y_{\pm})$, and τ^0, \ldots, τ^N are the corresponding co-ordinates of $\tau \in H^{\bullet}_{CR}(Y_{\pm})$, so that $\tau = \sum_{i=0}^N \tau^i \phi_i$. We consider now a similar structure on the *ambient part* of $H^{\bullet}_{CR}(Y_{\pm})$, that is, on:

$$H^{\bullet}_{\mathrm{amb}}(Y_{\pm}) := \operatorname{im} \iota^{\star}_{\pm} \subset H^{\bullet}_{\mathrm{CR}}(Y_{\pm})$$

where $\iota_{\pm} \colon Y_{\pm} \to X_{\pm}$ are the inclusion maps. If $\tau \in H^{\bullet}_{amb}(Y_{\pm})$ then the big quantum product \star_{τ} preserves $H^{\bullet}_{amb}(Y_{\pm})$ [51, Corollary 2.5], and so (7.2) restricts to give a welldefined quantum connection on the ambient part of $H^{\bullet}_{CR}(Y_{\pm})$. The restriction of the fundamental solution $L_{\pm}(\tau, z)$ for (7.2), defined exactly as in (2.8), gives a fundamental solution $L^{amb}_{\pm}(\tau, z)$ for the quantum connection on the ambient part.

There is also an ambient part of $K^0(Y_{\pm})$, given by $K^0_{amb}(Y_{\pm}) := \operatorname{im} \iota_{\pm}^*$, and an ambient K-group framing (cf. Definition 3.1)

$$\mathfrak{s} \colon K^0_{\mathrm{amb}}(Y_{\pm}) \to H^{\bullet}_{\mathrm{amb}}(Y_{\pm}) \otimes \mathbb{C}[\log z]((z^{-1/k})) \llbracket Q, \tau \rrbracket$$

given by

$$\mathfrak{s}(E)(\tau,z) = \frac{1}{(2\pi)^{\dim Y_{\pm}/2}} L_{\pm}^{\mathrm{amb}}(\tau,z) z^{-\mu} z^{\rho} \left(\widehat{\Gamma}_{Y_{\pm}} \cup (2\pi \mathtt{i})^{\frac{\deg_0}{2}} \operatorname{inv}^* \widetilde{\mathrm{ch}}(E)\right)$$

where μ and ρ are the grading operator and first Chern class for Y_{\pm} , $k \in \mathbb{N}$ is such that the eigenvalues of $k\mu$ are integers, and $\widehat{\Gamma}_{Y_{\pm}}$ is the non-equivariant $\widehat{\Gamma}$ -class of Y_{\pm} . As in §3, the image of \mathfrak{s} is contained in the space of flat sections for the quantum connection on the ambient part of $H^{\bullet}_{CR}(Y_{\pm})$ which are homogeneous of degree zero.

7.2. I-functions for toric complete intersections

Recall from §5.4 that the GIT data for X_+ determine a cohomology-valued hypergeometric function I_+ . The *I*-function $I_{X_+} := I_+$ is a multi-valued function of y_1, \ldots, y_r , depending analytically on y_r and formally on y_1, \ldots, y_{r-1} , defined near the large-radius limit point $(y_1, \ldots, y_r) = (0, \ldots, 0)$ in $\widehat{\mathcal{M}}_{reg}$. The GIT data for the total space of E_+^{\vee} (regarded as a non-compact toric stack) is obtained from the GIT data for X_+ by adding extra toric divisors $-E_1, \ldots, -E_k$. It is easy to see that the corresponding *I*-function $I_{E_+^{\vee}}$ is also a multi-valued function of y_1, \ldots, y_r , depending analytically on y_r and formally on y_1, \ldots, y_{r-1} , which is defined near the same large-radius limit point $(y_1, \ldots, y_r) = (0, \ldots, 0)$ in $\widehat{\mathcal{M}}_{reg}$. The global quantum connections for X_+ and E_+^{\vee} were constructed, in §5.5, using the *I*-functions I_{X_+} and $I_{E_+^{\vee}}$. We now introduce a closely-related *I*-function, defined in terms of GIT data for X_+ and the characters E_1, \ldots, E_k , that will allow us to globalize the quantum connection on the ambient part of $H_{CR}^{\circ}(Y_{\pm})$.

With notation as in §5.4, except with u_i now denoting the non-equivariant class Poincaré-dual to the *i*th toric divisor (4.6) and with $v_j \in H^2(X_+)$, $1 \leq j \leq k$, given by the non-equivariant first Chern class of the line bundle corresponding to the character E_j , define a $H^{\bullet}_{CR}(X_+)$ -valued hypergeometric series $I^{temp}_{X_+,Y_+}(\sigma, x, z) \in$ $H^{\bullet}_{CR}(X_+) \otimes \mathbb{C}((z^{-1}))[[Q, \sigma, x]]$ by: T. Coates et al. / Advances in Mathematics 329 (2018) 1002-1087

$$I_{X_{+},Y_{+}}^{\text{temp}}(\sigma,x,z) = ze^{\sigma/z} \sum_{d \in \mathbb{K}} e^{\sigma \cdot \overline{d}} Q^{\overline{d}} \prod_{j \in S} x_{j}^{D_{j} \cdot d} \left(\prod_{j=1}^{m} \frac{\prod_{a:\langle a \rangle = \langle D_{j} \cdot d \rangle, a \leq 0} (u_{j} + az)}{\prod_{a:\langle a \rangle = \langle D_{j} \cdot d \rangle, a \leq D_{j} \cdot d} (u_{j} + az)} \right) \times \left(\prod_{j=1}^{k} \prod_{a=1}^{E_{j} \cdot d} (v_{j} + az) \right) \mathbf{1}_{[-d]}$$

Note that for each $d \in \mathbb{K}$ and each $j \in \{1, 2, \ldots, k\}$, $E_j \cdot d$ is a non-negative integer. (The subscript 'temp' here again reflects the fact that this notation for the *I*-function is only temporary: we are just about to change notation, by specializing certain parameters.) Under our hypotheses (7.1) on the line bundles $L_{X_+}(E_j)$, we have a Mirror Theorem for the toric complete intersection Y_+ :

Theorem 7.1 ([30]). $\iota_{+}^{\star}I_{X_{+},Y_{+}}^{\text{temp}}(\sigma, x, -z)$ is a $\mathbb{C}\llbracket Q, \sigma, x \rrbracket$ -valued point on $\mathcal{L}_{Y_{+}}$.

We define the *I*-function I_{X_+,Y_+} to be the function obtained from $I_{X_+,Y_+}^{\text{temp}}$ by the specialization Q = 1, $\sigma = \sigma_+ := \theta_+ (\sum_{i=1}^r \mathsf{p}_i^+ \log \mathsf{y}_i)$ where θ_+ is as in (4.8). Thus:

$$I_{X_+,Y_+}(\mathbf{y},z) \\ := ze^{\sigma_+/z} \sum_{d \in \mathbb{K}_+} \mathbf{y}^d \left(\prod_{j=1}^m \frac{\prod_{a:\langle a \rangle = \langle D_j \cdot d \rangle, a \le 0} (u_j + az)}{\prod_{a:\langle a \rangle = \langle D_j \cdot d \rangle, a \le D_j \cdot d} (u_j + az)} \right) \left(\prod_{j=1}^k \prod_{a=1}^{E_j \cdot d} (v_j + az) \right) \mathbf{1}_{[-d]}$$

where (y_1, \ldots, y_r) are as in §5.4. Repeating the analysis in Lemma 5.13 shows that I_{X_+,Y_+} , just like I_{X_+} and $I_{E_+^{\vee}}$, is a multi-valued function of y_1, \ldots, y_r that depends analytically on y_r and formally on y_1, \ldots, y_{r-1} , defined near the large-radius limit point $(y_1, \ldots, y_r) = (0, \ldots, 0)$ in $\widehat{\mathcal{M}}_{reg}$. The arguments in §5.5 can now be applied verbatim to $I_{Y_+} := \iota_+^* I_{X_+,Y_+}$, and thus we construct a global version of the quantum connection on the ambient part $H^{\bullet}_{amb}(Y_+)$, defined over the base $\widetilde{\mathcal{M}}_+^{\circ}$. The analog of Theorem 5.14 holds, with the same proof:

Theorem 7.2. There exist the following data:

- an open subset U^o₊ ⊂ U₊ such that P₊ ∈ U^o₊ and that the complement U₊ \ U^o₊ is a discrete set; we write M^o₊ = M₊|_{U^o₊};
- a trivial $H^{\bullet}_{amb}(Y_+)$ -bundle \mathbf{F}^+ over $\widetilde{\mathcal{M}}^{\circ}_+(\mathbb{C}[z])$:

$$\mathbf{F}^+ = H^{\bullet}_{\mathrm{amb}}(Y_+) \otimes \mathcal{O}_{\mathcal{U}^{\circ}_+}[z] \llbracket \mathsf{y}_1, \dots, \mathsf{y}_{r-1} \rrbracket;$$

• a flat connection $\nabla^+ = d + z^{-1} \mathbf{A}^+(\mathbf{y})$ on \mathbf{F}^+ of the form:

$$\mathbf{A}^{+}(\mathbf{y}) = \sum_{i=1}^{\ell_{+}} B_{i}(\mathbf{y}) \frac{dy_{i}}{y_{i}} + \sum_{j \in S_{+}} C_{j}(\mathbf{y}) dx_{j}$$

with $B_i(\mathbf{y}), C_j(\mathbf{y}) \in \operatorname{End}(H^{\bullet}_{\operatorname{amb}}(Y_+)) \otimes \mathcal{O}_{\mathcal{U}^{\circ}_+}[\![\mathbf{y}_1, \ldots, \mathbf{y}_{r-1}]\!];$

• a vector field \mathbf{E}^+ on $\widetilde{\mathcal{M}}_+$, called the Euler vector field, defined by:

$$\mathbf{E}^{+} = \sum_{i=1}^{r} \frac{1}{2} (\deg \mathsf{y}_{i}) \mathsf{y}_{i} \frac{\partial}{\partial \mathsf{y}_{i}};$$

• a mirror map $\tau_+ \colon \widetilde{\mathcal{M}}_+ \to H^{\bullet}_{\mathrm{amb}}(Y_+)$ of the form:

$$\begin{aligned} \tau_{+} &= \iota_{+}^{\star} \sigma_{+} + \tilde{\tau}_{+} \qquad \tilde{\tau}_{+} \in H^{\bullet}_{\mathrm{amb}}(Y_{+}) \otimes \mathcal{O}_{\mathcal{U}^{\bullet}_{+}}[\![\mathbf{y}_{1}, \dots, \mathbf{y}_{r-1}]\!] \\ & \tilde{\tau}_{+}|_{\mathbf{y}_{1} = \dots = \mathbf{y}_{r} = 0} = 0 \end{aligned}$$

such that ∇^+ equals the pull-back $\tau^*_+ \nabla^+$ of the (non-equivariant) quantum connection ∇^+ on the ambient part of $H^{\bullet}_{CR}(Y_+)$ by τ_+ , that is:

$$B_{i}(\mathbf{y}) = \sum_{k=0}^{N} \frac{\partial \tau_{+}^{k}(\mathbf{y})}{\partial \log y_{i}} (\phi_{k} \star_{\tau_{+}(\mathbf{y})}) \qquad 1 \le i \le \ell_{+}$$
$$C_{j}(\mathbf{y}) = \sum_{k=0}^{N} \frac{\partial \tilde{\tau}_{+}^{k}(\mathbf{y})}{\partial x_{j}} (\phi_{k} \star_{\tau_{+}(\mathbf{y})}) \qquad j \in S_{+}$$

and that the push-forward of \mathbf{E}^+ by τ_+ is the (non-equivariant) Euler vector field \mathcal{E}^+ on the ambient part $H^{\bullet}_{amb}(Y_+)$. Moreover, there exists a global section $\Upsilon^+_0(\mathbf{y}, z)$ of \mathbf{F}^+ such that

$$I_{Y_{+}}(\mathbf{y}, z) = z L_{+}^{\text{amb}}(\tau_{+}(\mathbf{y}), z)^{-1} \Upsilon_{0}^{+}(\mathbf{y}, z)$$

where $L^{\text{amb}}_{+}(\tau, z)$ is the ambient fundamental solution from §7.1.

Remark 7.3. Here, as in Theorem 5.14, the Novikov variable Q has been specialized to 1.

Remark 7.4. Entirely parallel results hold for Y_{-} .

7.3. Analytic continuation of I-functions

To prove the Crepant Transformation Conjecture in this context, we need to establish the analog of Theorem 6.1. To do this, we will compare the analytic continuation of the *I*-functions $I_{X_{\pm},Y_{\pm}}$ with the analytic continuation of $I_{E_{\pm}^{\vee}}$. Let $T = (\mathbb{C}^{\times})^m$ denote the torus acting on X_{\pm} , and $\tilde{T} = (\mathbb{C}^{\times})^{m+k}$ denote the torus acting on E_{\pm}^{\vee} . The splitting $\tilde{T} = T \times (\mathbb{C}^{\times})^k$ gives $R_{\tilde{T}} = R_T[\kappa_1, \ldots, \kappa_k]$ where $\kappa_j, 1 \leq j \leq k$, is the character of $(\mathbb{C}^{\times})^k$ given by projection to the *j*th factor of the product $(\mathbb{C}^{\times})^k$. We regard \tilde{T} as acting on X_{\pm} via the given action of $T \subset \tilde{T}$ and the trivial action of $(\mathbb{C}^{\times})^k \subset \tilde{T}$, so that:

$$\mathbb{Z}[\widetilde{T}] = \mathbb{Z}[T][e^{\pm\kappa_1}, \dots, e^{\pm\kappa_k}] \qquad \text{and} \qquad K^0_{\widetilde{T}}(X_{\pm}) = K^0_T(X_{\pm}) \otimes_{\mathbb{Z}[T]} \mathbb{Z}[\widetilde{T}]$$

Lemma 7.5. The Fourier–Mukai transformations

$$\mathbb{FM}: K^0(X_-) \to K^0(X_+) \qquad \mathbb{FM}: K^0(E_-^{\vee}) \to K^0(E_+^{\vee})$$

coincide under the natural identification of $K^0(X_{\pm})$ with $K^0(E_{\pm}^{\vee})$. The same statement holds equivariantly.

Proof. Consider the fibre diagram:



where the bottom triangle is (1.1) and the top triangle is the analog of (1.1) for E_{\pm}^{\vee} , and apply the flat base change theorem. \Box

Let $\mathbb{U}_{E^{\vee}}$ be the symplectic transformation from Theorem 6.1 applied to E_{\pm}^{\vee} . Combining Lemma 7.5 with Theorem 6.1 gives a commutative diagram:

where $\rho_{\pm} \in H^2_{\widetilde{T}}(E_{\pm}^{\vee})$ is the \widetilde{T} -equivariant first Chern class of E_{\pm}^{\vee} and μ_{\pm} are the \widetilde{T} -equivariant grading operators. Recall that

$$\widehat{\Gamma}_{E_{\pm}^{\vee}} = \widehat{\Gamma}_{X_{\pm}} \widehat{\Gamma}(E_{\pm}^{\vee}) \qquad \rho_{\pm} = \rho_{X_{\pm}} + c_1^{\widetilde{T}}(E_{\pm}^{\vee})$$

and that the Chern roots of E_{\pm}^{\vee} are pulled back from the common blow-down \overline{X}_0 of X_{\pm} . Part (2) of Theorem 6.1 thus implies that we can factor out the contributions of $\widehat{\Gamma}(E_{\pm}^{\vee})$ and $c_1^{\widetilde{T}}(E_{\pm}^{\vee})$ from the vertical maps in (7.3), replacing the vertical arrows by:

$$z^{-\mu_{X_{\pm}}} z^{\rho_{X_{\pm}}} \widehat{\Gamma}_{X_{\pm}} \cup (2\pi i)^{\frac{\deg_0}{2}} \operatorname{inv}^* \widetilde{\operatorname{ch}}(-)$$

This proves:

Lemma 7.6. The transformations $\mathbb{U}_X : \mathcal{H}(X_-) \to \mathcal{H}(X_+)$ and $\mathbb{U}_{E^{\vee}} : \mathcal{H}(E_-^{\vee}) \to \mathcal{H}(E_+^{\vee})$ coincide under the natural identifications of $\mathcal{H}(X_{\pm})$ with $\mathcal{H}(E_{\pm}^{\vee})$. In particular, $\mathbb{U}_{E^{\vee}}$ is independent of $\kappa_1, \ldots, \kappa_k$.

The *I*-functions I_{X_+,Y_+} and $I_{E_{\perp}^{\vee}}$ are related¹⁶ by:

$$I_{E_+^{\vee}}(\mathbf{y})\Big|_{\lambda=0,\kappa=-z}=e^{\pi\mathrm{i}c_1(E_+^{\vee})/z}I_{X_+,Y_+}(\pm\mathbf{y})$$

where the subscript on the left-hand side denotes the specialization:

$$\begin{cases} \lambda_i = 0 & 1 \le i \le m \\ \kappa_j = -z & 1 \le j \le k \end{cases}$$
(7.4)

and the \pm on the right-hand side denotes the change of variables:

$$\log \mathbf{y}_i \mapsto \log \mathbf{y}_i - \pi \mathbf{i} \sum_{j=1}^k l_{ij} \qquad 1 \le i \le r \qquad \text{with} \qquad E_j = \sum_{i=1}^r l_{ij} \mathbf{p}_i \qquad (7.5)$$

The specialization (7.4) is given by a shift $\mathbb{S}: \kappa_j \mapsto \kappa_j - z$ in the equivariant parameters followed by passing to the non-equivariant limit. Note that the change of variables (7.5) maps y^d to $(-1)^{-c_1(E_+^{\vee}) \cdot d} y^d$.

Recall from Theorem 6.1 that, after analytic continuation, we have $I_{E_{+}^{\vee}} = \mathbb{U}_{E^{\vee}} I_{E_{-}^{\vee}}$. Since $\mathbb{U}_{E^{\vee}}$ is independent of κ_j , $1 \leq j \leq k$, it follows that $\mathbb{U}_{E^{\vee}}$ commutes with the shift S. Since the Chern roots of E^{\vee} are pulled back from the common blow-down \overline{X}_0 of X_{\pm} , it follows that

$$\mathbb{U}_{E^{\vee}} e^{\pi \mathbf{i} c_1(E^{\vee}_-)/z} = e^{\pi \mathbf{i} c_1(E^{\vee}_+)/z} \mathbb{U}_{E^{\vee}}$$

Setting $\lambda = 0$ and $\kappa_j = -z$ in the equality $I_{E_+^{\vee}} = \mathbb{U}_{E^{\vee}} I_{E_-^{\vee}}$, and replacing $\mathcal{H}(E_{\pm}^{\vee})$ and $\mathbb{U}_{E^{\vee}}$ with their non-equivariant limits

$$\mathcal{H}(E_{\pm}^{\vee}) := H^{\bullet}_{\mathrm{CR}}(E_{\pm}^{\vee}) \otimes \mathbb{C}((z^{-1})) \qquad \text{and} \qquad \mathbb{U}_{E^{\vee}} : \mathcal{H}(E_{-}^{\vee}) \to \mathcal{H}(E_{\pm}^{\vee})$$

we find that

$$I_{X_+,Y_+} = \mathbb{U}_{E^{\vee}} I_{X_-,Y_-}$$

after analytic continuation. Thus:

$$I_{X_+,Y_+} = \mathbb{U}_X I_{X_-,Y_-}$$

 $^{^{16}\,}$ An analogous relationship holds between I_{X_-,Y_-} and $I_{E_-^\vee}.$

after analytic continuation; here \mathbb{U}_X denotes the non-equivariant limit of \mathbb{U}_X in the lemma above.

7.4. Compatibility of Fourier–Mukai transformations

For the analogue of part (3) of Theorem 6.1, we need to compare the Fourier–Mukai transformation associated to $X_+ \dashrightarrow X_-$ with the Fourier–Mukai transformation associated to $Y_+ \dashrightarrow Y_-$. This is a base change question (cf. Lemma 7.5), but this time we do not have flatness. By assumption, we have:



where the vertical maps are inclusions, the bottom triangle is (1.1) and the top triangle is the analog of (1.1) for Y_{\pm} . The substacks \tilde{Y} is defined by the vanishing of a section $\tilde{s}: \tilde{X} \to \tilde{E}$, where $\tilde{E} \to \tilde{X}$ is the direct sum of line bundles

$$\widetilde{E} := \bigoplus_{i=1}^{k} L_{\widetilde{X}}(E_i)$$

The line bundles E_- , \tilde{E} , and E_+ are all canonically identified via f_-^* and f_+^* , since they are all pulled back from the common blow-down \overline{X}_0 of X_{\pm} . The section \tilde{s} coincides both with the pullback of the section s_+ via f_+ and with the pullback of the section s_- via f_- . Since the zero loci of s_{\pm} are assumed to intersect the flopping locus transversely, \tilde{s} is a regular section of \tilde{E} and the substack $\tilde{Y} \subset \tilde{X}$ is smooth.

Lemma 7.7. The following diagram commutes:

where the top horizontal arrow is the Fourier–Mukai transformation $(f_+)_*(f_-)^*$ from (7.6), and the bottom horizontal arrow is the Fourier–Mukai transformation $(F_+)_*(F_-)^*$ from (7.6).

Proof. The pullback along f_+ of the Koszul resolution of \mathcal{O}_{Y_+} in X_+ gives the Koszul resolution of $\mathcal{O}_{\tilde{Y}}$ in \tilde{X} . This implies that, in the right-hand square in (7.6), \tilde{X} and Y_+ are Tor-independent over X_+ [72, Tag 08IA]. Tor-independent base-change [72, Tag 08IB] now implies that:

$$(F_+)_{\star} \circ \tilde{\iota}^{\star} = \iota_+^{\star} \circ (f_+)_{\star}$$

Since $F_{-}^{\star} \circ \iota_{-}^{\star} = \tilde{\iota}^{\star} \circ f_{-}^{\star}$, it follows that

$$(\iota_+)^*(f_+)_*(f_-)^* = (F_+)_*(F_-)^*(\iota_-)_*$$

which is the result. \Box

Remark 7.8. This argument in fact proves that the analog of diagram (7.7) for derived categories is commutative, but we only need the statement at the level of K-theory.

7.5. Completing the proof

Denote by \mathbb{U}_X the transformation from the non-equivariant version of Theorem 6.1 applied to X_{\pm} . This is a map $\mathbb{U}_X : \mathcal{H}(X_-) \to \mathcal{H}(X_+)$ between the non-equivariant Givental spaces for X_{\pm} :

$$\mathcal{H}(X_{\pm}) := H^{\bullet}_{\mathrm{CR}}(X_{\pm}) \otimes \mathbb{C}((z^{-1}))$$

Let us remark again that the Chern roots of E_{\pm} are pulled back from the common blow-down \overline{X}_0 of X_{\pm} ; the second part of Theorem 6.1 therefore gives:

$$\mathbb{U}_X \widehat{\Gamma}(E_-) = \widehat{\Gamma}(E_+) \mathbb{U}_X \tag{7.8}$$

The results from §7.3 and §7.4 combine to give a commutative diagram:



where $\widetilde{\mathcal{H}}(Y_{\pm})_{\text{amb}}$ is the ambient part of the multi-valued Givental space:

$$\widetilde{\mathcal{H}}(Y_{\pm})_{\mathrm{amb}} := H^{\bullet}_{\mathrm{amb}}(Y_{\pm}) \otimes \mathbb{C}[\log z]((z^{-1/k}))$$
(7.9)

with k as in the statement of Theorem 6.1, and:

- the top diagonal maps are the K-theory framing maps from Definition 3.1 but with $\widehat{\Gamma}_{X_{\pm}}$ replaced by $\widehat{\Gamma}_{X_{\pm},Y_{\pm}} := \widehat{\Gamma}_{X_{\pm}}\widehat{\Gamma}(E_{\pm})^{-1}$;
- the bottom diagonal maps are the ambient K-group framing maps from §7.1.

Here:

- the top face is commutative, by Theorem 6.1 and (7.8);
- the back face is commutative, by Lemma 7.7;
- the sides are commutative, by the definition of the framing maps;

and we want to define the dotted arrow so that all faces commute. Define \mathbb{U}_Y : $\widetilde{\mathcal{H}}(Y_-)_{\mathrm{amb}} \to \widetilde{\mathcal{H}}(Y_+)_{\mathrm{amb}}$ to be the unique map such that the bottom face commutes. Chasing diagrams shows that the front face commutes also. Since $I_{X_+,Y_+} = \mathbb{U}_X I_{X_-,Y_-}$ after analytic continuation and since $I_{Y_{\pm}} := \iota_{\pm}^* I_{X_{\pm},Y_{\pm}}$, we conclude that $I_{Y_+} = \mathbb{U}_Y I_{Y_-}$ after analytic continuation.

Theorem 7.9. Consider the ambient part of the (non-equivariant) Givental space for Y_{\pm} with the Novikov variable Q specialized to 1:

$$\mathcal{H}(Y_{\pm})_{\mathrm{amb}} = H^{\bullet}_{\mathrm{amb}}(Y_{\pm}) \otimes \mathbb{C}((z^{-1}))$$

Regard $\mathcal{H}(Y_{\pm})_{\text{amb}}$ as a graded vector space, where we use the age-shifted grading on $H^{\bullet}_{\text{amb}}(Y_{\pm})$ and set deg z = 2. There exists a degree-preserving $\mathbb{C}((z^{-1}))$ -linear transformation

$$\mathbb{U}_Y \colon \mathcal{H}(Y_-)_{\mathrm{amb}} \to \mathcal{H}(Y_+)_{\mathrm{amb}}$$

such that:

- (1) $I_{Y_+}(\mathbf{y}, z) = \mathbb{U}_Y I_{Y_-}(\mathbf{y}, z)$ after analytic continuation in \mathbf{y}^e along the path γ in Fig. 1;
- (2) $\mathbb{U}_Y \circ (g_-^* v \cup) = (g_+^* v \cup) \circ \mathbb{U}_Y$ for all $v \in H^2(\overline{X}_0)$, where \overline{X}_0 is the common blow-down of X_{\pm} and $g_{\pm} \colon Y_{\pm} \to \overline{X}_0$ is the composition of the inclusion $\iota_{\pm} \colon Y_{\pm} \to X_{\pm}$ with the blow-down $X_{\pm} \to \overline{X}_0$;

(3) there is a commutative diagram



where \mathbb{FM} is the Fourier-Mukai transformation given by the top triangle in (7.6) and the vertical arrows are the ambient K-group framing defined in §7.1.

If Y_{\pm} is compact then \mathbb{U}_Y intertwines the (possibly-degenerate) symplectic pairings on $\mathcal{H}(Y_{\pm})_{amb}$.

Proof. Everything has been proved except the statement that, if Y_{\pm} is compact, then \mathbb{U}_Y intertwines the pairings on $\mathcal{H}(Y_{\pm})_{\text{amb}}$. But:

$$\begin{split} \left(\mathbb{U}_{Y}(-z)\iota_{-}^{\star}\alpha, \mathbb{U}_{Y}(z)\iota_{-}^{\star}\beta \right)_{Y_{+}} &= \left(\iota_{+}^{\star}\mathbb{U}_{X}(-z)\alpha, \iota_{+}^{\star}\mathbb{U}_{X}(z)\beta \right)_{Y_{+}} \\ &= \left(\mathbb{U}_{X}(-z)\alpha, e(E_{+})\mathbb{U}_{X}(z)\beta \right)_{X_{+}} \\ &= \left(\mathbb{U}_{X}(-z)\alpha, \mathbb{U}_{X}(z)e(E_{-})\beta \right)_{X_{+}} \quad \text{by Theorem 6.1(2)} \\ &= \left(\alpha, e(E_{-})\beta \right)_{X_{-}} = \left(\iota_{-}^{\star}\alpha, \iota_{-}^{\star}\beta \right)_{Y_{-}} \quad \Box \end{split}$$

Remark 7.10. If Y_{\pm} is compact then the Givental space for Y_{\pm} has a well-defined symplectic pairing, but the restriction of this pairing to the ambient part is non-degenerate if and only if $(\iota_{\pm})_{\star}: H^{\bullet}_{amb}(Y_{\pm}) \to H^{\bullet}_{CR}(X_{\pm})$ is injective. This holds by the Hard Lefschetz Theorem when E_{\pm} is a direct sum of ample line bundles, but our assumption only ensures that the line bundles are semiample and the question is more subtle in general. Injectivity holds when Y_{\pm} is a regular semiample hypersurface by a result of Mavlyutov [64, Theorem 5.1].

Theorem 7.9 is the analog, for toric complete intersections, of Theorem 6.1. The analog of Theorem 6.3 also holds:

Theorem 7.11. Let $(\mathbf{F}^{\pm}, \nabla^{\pm}, \mathbf{E}^{\pm})$ be the global quantum connections for the ambient parts $H^{\bullet}_{amb}(Y_{\pm})$ over $\widetilde{\mathcal{M}}^{\circ}_{\pm}(\mathbb{C}[z])$ from Theorem 7.2. We have that $\mathbf{E}^{+} = \mathbf{E}^{-}$ on $\widetilde{\mathcal{M}}$. There exists a gauge transformation

$$\Theta_Y \in \operatorname{Hom}\left(H^{\bullet}_{\operatorname{amb}}(Y_{-}), H^{\bullet}_{\operatorname{amb}}(Y_{+})\right) \otimes \mathcal{O}_{\mathcal{U}^{\circ}}[z]\llbracket \mathsf{y}_1, \dots, \mathsf{y}_{r-1}\rrbracket$$

over $\widetilde{\mathcal{M}}^{\circ}(\mathbb{C}[z])$ such that:

- ∇^- and ∇^+ are gauge-equivalent via Θ_Y , i.e. $\nabla^+ \circ \Theta_Y = \Theta_Y \circ \nabla^-$;
- Θ_Y is homogeneous of degree zero, i.e. $\mathbf{Gr}^+ \circ \Theta_Y = \Theta_Y \circ \mathbf{Gr}^-$ with $\mathbf{Gr}^{\pm} := z \frac{\partial}{\partial z} + \mathbf{E}^{\pm} + \mu^{\pm};$
- if Y_± are compact then Θ_Y preserves the (possibly-degenerate) orbifold Poincaré pairing on H[●]_{amb}(Y_±), i.e. (Θ_Y(y, -z)α, Θ_Y(y, z)β) = (α, β).

Moreover, the analytic continuation of flat sections coincides, via the ambient K-group framing defined in ^{§7.1}, with the Fourier–Mukai transformation:

$$\Theta_Y(\mathfrak{s}(E)(\tau_-(\mathsf{y}), z)) = \mathfrak{s}(\mathbb{F}\mathbb{M}(E))(\tau_+(\mathsf{y}), z) \qquad \text{for all } E \in K^0(Y_-)_{\text{amb}}$$

where τ_{\pm} are the mirror maps from Theorem 7.2.

Theorem 7.11 follows from Theorem 7.9 exactly as Theorem 6.3 follows from Theorem 6.1. The transformation \mathbb{U}_Y in Theorem 7.9 and the gauge transformation Θ_Y in Theorem 7.11 are related by

$$L^{\mathrm{amb}}_{+}(\tau_{+}(\mathsf{y}), z)^{-1} \circ \Theta_{Y} = \mathbb{U}_{Y} \circ L^{\mathrm{amb}}_{-}(\tau_{-}(\mathsf{y}), z)^{-1}$$

where L_{\pm} is the ambient fundamental solution from §7.1. The gauge transformation Θ_Y sends the section $\Upsilon_0^- \in \mathbf{F}^-$ to the section $\Upsilon_0^+ \in \mathbf{F}^+$, where Υ_0^{\pm} are as in Theorem 7.2.

Acknowledgments

We thank Alessio Corti, Alexander Kasprzyk, Yuan-Pin Lee, Nathan Priddis and Mark Shoemaker for a number of useful conversations. This research was supported by a Royal Society University Research Fellowship; the Leverhulme Trust; ERC Starting Investigator Grant 240123; EPSRC Mathematics Platform grant EP/I019111/1; JSPS Kakenhi Grants 17H06127, 16H06335, 16H06337, 16K05127, 26610008, 25400069, 24224001, 23224002; NFGRF, University of Kansas; and Simons Foundation Collaboration Grant 311837. Yunfeng Jiang is also supported by NSF Grant DMS-1600997.

References

- D. Abramovich, T. Graber, A. Vistoli, Algebraic orbifold quantum products, in: Orbifolds in Mathematics and Physics, Madison, WI, 2001, in: Contemp. Math., vol. 310, Amer. Math. Soc., Providence, RI, 2002, pp. 1–24.
- [2] D. Abramovich, T. Graber, A. Vistoli, Gromov–Witten theory of Deligne–Mumford stacks, Amer. J. Math. 130 (5) (2008) 1337–1398.
- [3] M.F. Atiyah, R. Bott, The moment map and equivariant cohomology, Topology 23 (1) (1984) 1–28.
- [4] M. Audin, Torus Actions on Symplectic Manifolds, revised edition, Progress in Mathematics, vol. 93, Birkhäuser Verlag, Basel, 2004.
- [5] S. Barannikov, Quantum periods. I. Semi-infinite variations of Hodge structures, Int. Math. Res. Not. IMRN (23) (2001) 1243–1264.
- [6] E.W. Barnes, A new development of the theory of the hypergeometric functions, Proc. Lond. Math. Soc. S2-6 (1) (1908) 141.

- [7] V.V. Batyrev, Dual polyhedra and mirror symmetry for Calabi–Yau hypersurfaces in toric varieties, J. Algebraic Geom. 3 (3) (1994) 493–535.
- [8] N. Berline, M. Vergne, Classes caractéristiques équivariantes. Formule de localisation en cohomologie équivariante, C. R. Acad. Sci. Paris Sér. I Math. 295 (9) (1982) 539–541.
- [9] S. Boissière, É. Mann, F. Perroni, The cohomological crepant resolution conjecture for P(1, 3, 4, 4), Internat. J. Math. 20 (6) (2009) 791–801.
- [10] S. Boissière, É. Mann, F. Perroni, Computing certain Gromov–Witten invariants of the crepant resolution of P(1, 3, 4, 4), Nagoya Math. J. 201 (2011) 1–22.
- [11] L.A. Borisov, L. Chen, G.G. Smith, The orbifold Chow ring of toric Deligne–Mumford stacks, J. Amer. Math. Soc. 18 (1) (2005) 193–215 (electronic).
- [12] L.A. Borisov, R.P. Horja, Mellin–Barnes integrals as Fourier–Mukai transforms, Adv. Math. 207 (2) (2006) 876–927.
- [13] L.A. Borisov, R.P. Horja, On the K-theory of smooth toric DM stacks, in: Snowbird Lectures on String Geometry, in: Contemp. Math., vol. 401, Amer. Math. Soc., Providence, RI, 2006, pp. 21–42.
- [14] L.A. Borisov, R.P. Horja, On the better behaved version of the GKZ hypergeometric system, Math. Ann. 357 (2) (2013) 585–603.
- [15] A. Braverman, D. Maulik, A. Okounkov, Quantum cohomology of the Springer resolution, Adv. Math. 227 (1) (2011) 421–458.
- [16] A. Brini, R. Cavalieri, D. Ross, Crepant resolutions and open strings, arXiv:1309.4438 [math.AG], 2013.
- [17] J. Bryan, A. Gholampour, Hurwitz–Hodge integrals, the E_6 and D_4 root systems, and the crepant resolution conjecture, Adv. Math. 221 (4) (2009) 1047–1068.
- [18] J. Bryan, A. Gholampour, The quantum McKay correspondence for polyhedral singularities, Invent. Math. 178 (3) (2009) 655–681.
- [19] J. Bryan, T. Graber, The crepant resolution conjecture, in: Algebraic Geometry—Seattle 2005. Part 1, in: Proc. Sympos. Pure Math., vol. 80, Amer. Math. Soc., Providence, RI, 2009, pp. 23–42.
- [20] J. Bryan, T. Graber, R. Pandharipande, The orbifold quantum cohomology of C²/Z₃ and Hurwitz– Hodge integrals, J. Algebraic Geom. 17 (1) (2008) 1–28.
- [21] P. Candelas, X.C. de la Ossa, P.S. Green, L. Parkes, A pair of Calabi–Yau manifolds as an exactly soluble superconformal theory, Nuclear Phys. B 359 (1) (1991) 21–74.
- [22] B. Chen, A.-M. Li, X. Li, G. Zhao, Ruan's conjecture on singular symplectic flops of mixed type, Sci. China Math. 57 (6) (2014) 1121–1148.
- [23] B. Chen, A.-M. Li, Q. Zhang, G. Zhao, Singular symplectic flops and Ruan cohomology, Topology 48 (1) (2009) 1–22.
- [24] W. Chen, Y. Ruan, Orbifold Gromov-Witten theory, in: Orbifolds in Mathematics and Physics, Madison, WI, 2001, in: Contemp. Math., vol. 310, Amer. Math. Soc., Providence, RI, 2002, pp. 25–85.
- [25] W. Chen, Y. Ruan, A new cohomology theory of orbifold, Comm. Math. Phys. 248 (1) (2004) 1–31.
- [26] D. Cheong, I. Ciocan-Fontanine, B. Kim, Orbifold quasimap theory, Math. Ann. 363 (3–4) (2015) 777–816.
- [27] T. Coates, On the crepant resolution conjecture in the local case, Comm. Math. Phys. 287 (3) (2009) 1071–1108.
- [28] T. Coates, A. Corti, H. Iritani, H.-H. Tseng, Computing genus-zero twisted Gromov–Witten invariants, Duke Math. J. 147 (3) (2009) 377–438.
- [29] T. Coates, A. Corti, H. Iritani, H.-H. Tseng, A mirror theorem for toric stacks, Compos. Math. 10 (151) (2015) 1878–1912.
- [30] T. Coates, A. Corti, H. Iritani, H.-H. Tseng, Some applications of the mirror theorems for toric stacks, arXiv:1401.2611 [math.AG], 2014.
- [31] T. Coates, A. Givental, Quantum Riemann-Roch, Lefschetz and Serre, Ann. of Math. (2) 165 (1) (2007) 15–53.
- [32] T. Coates, H. Iritani, Y. Jiang, E. Segal, K-theoretic and categorical properties of toric Deligne-Mumford stacks, Pure Appl. Math. Q. 11 (2) (2015) 239–266.
- [33] T. Coates, H. Iritani, H.-H. Tseng, Wall-crossings in toric Gromov–Witten theory. I. Crepant examples, Geom. Topol. 13 (5) (2009) 2675–2744.
- [34] T. Coates, Y. Ruan, Quantum cohomology and crepant resolutions: a conjecture, Ann. Inst. Fourier (Grenoble) 63 (2) (2013) 431–478.
- [35] D.A. Cox, J.B. Little, H.K. Schenck, Toric Varieties, Graduate Studies in Mathematics, vol. 124, American Mathematical Society, Providence, RI, 2011.

- [36] I.V. Dolgachev, Y. Hu, Variation of geometric invariant theory quotients, Publ. Math. Inst. Hautes Études Sci. 87 (1998) 5–56. With an appendix by Nicolas Ressayre.
- [37] B. Fantechi, E. Mann, F. Nironi, Smooth toric Deligne–Mumford stacks, J. Reine Angew. Math. 648 (2010) 201–244.
- [38] I.M. Gel'fand, M.M. Kapranov, A.V. Zelevinskiĭ, Discriminants, Resultants and Multidimensional Determinants, Modern Birkhäuser Classics, Birkhäuser Boston, Inc., Boston, MA, 2008.
- [39] I.M. Gel'fand, A.V. Zelevinskiĭ, M.M. Kapranov, Hypergeometric functions and toric varieties, Funktsional. Anal. i Prilozhen. 23 (2) (1989) 12–26.
- [40] W.D. Gillam, The crepant resolution conjecture for three-dimensional flags modulo an involution, Comm. Algebra 41 (2) (2013) 736–764.
- [41] A.B. Givental, Equivariant Gromov–Witten invariants, Int. Math. Res. Not. IMRN 13 (1996) 613–663.
- [42] A.B. Givental, Elliptic Gromov-Witten invariants and the generalized mirror conjecture, in: Integrable Systems and Algebraic Geometry, Kobe/Kyoto, 1997, World Sci. Publ., River Edge, NJ, 1998, pp. 107–155.
- [43] A.B. Givental, Symplectic geometry of Frobenius structures, in: Frobenius Manifolds, in: Aspects Math., vol. E36, Friedr. Vieweg, Wiesbaden, 2004, pp. 91–112.
- [44] E. Gonzalez, C.T. Woodward, A wall-crossing formula for Gromov–Witten invariants under variation of GIT quotient, arXiv:1208.1727 [math.AG], 2012.
- [45] M. Goresky, R. Kottwitz, R. MacPherson, Equivariant cohomology, Koszul duality, and the localization theorem, Invent. Math. 131 (1) (1998) 25–83.
- [46] B.R. Greene, M.R. Plesser, Duality in Calabi–Yau moduli space, Nuclear Phys. B 338 (1) (1990) 15–37.
- [47] M. Herbst, K. Hori, D. Page, Phases of N = 2 theories in 1 + 1 dimensions with boundary, arXiv: 0803.2045 [hep-th], 2008;
 A short summary published as: B-type D-branes in toric Calabi–Yau varieties, in: Homological Mirror Symmetry, in: Lecture Notes in Phys., vol. 757, Springer, Berlin, 2009, pp. 27–44.
- [48] K. Hori, M. Romo, Exact results in two-dimensional (2, 2) supersymmetric gauge theories with boundary, arXiv:1308.2438 [hep-th], 2013.
- [49] H. Iritani, An integral structure in quantum cohomology and mirror symmetry for toric orbifolds, Adv. Math. 222 (3) (2009) 1016–1079.
- [50] H. Iritani, Ruan's conjecture and integral structures in quantum cohomology, in: New Developments in Algebraic Geometry, Integrable Systems and Mirror Symmetry, RIMS, Kyoto, 2008, in: Adv. Stud. Pure Math., vol. 59, Math. Soc. Japan, Tokyo, 2010, pp. 111–166.
- [51] H. Iritani, Quantum cohomology and periods, Ann. Inst. Fourier (Grenoble) 61 (7) (2011) 2909–2958.
- [52] Y. Iwao, Y.-P. Lee, H.-W. Lin, C.-L. Wang, Invariance of Gromov–Witten theory under a simple flop, J. Reine Angew. Math. 663 (2012) 67–90.
- [53] Y. Jiang, The orbifold cohomology ring of simplicial toric stack bundles, Illinois J. Math. 52 (2) (2008) 493–514.
- [54] L. Katzarkov, M. Kontsevich, T. Pantev, Hodge theoretic aspects of mirror symmetry, in: From Hodge Theory to Integrability and TQFT tt*-Geometry, in: Proc. Sympos. Pure Math., vol. 78, Amer. Math. Soc., Providence, RI, 2008, pp. 87–174.
- [55] Y.-P. Lee, H.-W. Lin, F. Qu, C.-L. Wang, Invariance of quantum rings under ordinary flops: III, arXiv:1401.7097 [math.AG], 2014.
- [56] Y.-P. Lee, H.-W. Lin, C.-L. Wang, Flops, motives, and invariance of quantum rings, Ann. of Math. (2) 172 (1) (2010) 243–290.
- [57] Y.-P. Lee, H.-W. Lin, C.-L. Wang, Analytic continuations of quantum cohomology, in: Fifth International Congress of Chinese Mathematicians. Part 1, 2, in: AMS/IP Stud. Adv. Math., vol. 2, Amer. Math. Soc., Providence, RI, 2012, pp. 269–282, 51, pt. 1.
- [58] Y.-P. Lee, H.-W. Lin, C.-L. Wang, Invariance of quantum rings under ordinary flops II: A quantum Leray–Hirsch theorem, Algebr. Geom. 3 (5) (2016) 615–653.
- [59] Y.-P. Lee, N. Priddis, M. Shoemaker, A proof of the Landau–Ginzburg/Calabi–Yau correspondence via the Crepant Transformation Conjecture, arXiv:1410.5503 [math.AG], 2014.
- [60] Y.-P. Lee, M. Shoemaker, A mirror theorem for the mirror quintic, Geom. Topol. 18 (3) (2014) 1437–1483.
- [61] T.-J. Li, Y. Ruan, Symplectic birational geometry, in: New Perspectives and Challenges in Symplectic Field Theory, in: CRM Proc. Lecture Notes, vol. 49, Amer. Math. Soc., Providence, RI, 2009, pp. 307–326.

- [62] Y.I. Manin, Frobenius Manifolds, Quantum Cohomology, and Moduli Spaces, American Mathematical Society Colloquium Publications, vol. 47, American Mathematical Society, Providence, RI, 1999.
- [63] D. Maulik, A. Okounkov, Quantum groups and quantum cohomology, arXiv:1211.1287 [math.AG], 2012.
- [64] A.R. Mavlyutov, On the chiral ring of Calabi–Yau hypersurfaces in toric varieties, Compos. Math. 138 (3) (2003) 289–336.
- [65] A. Okounkov, R. Pandharipande, The quantum differential equation of the Hilbert scheme of points in the plane, Transform. Groups 15 (4) (2010) 965–982.
- [66] R. Pandharipande, Rational curves on hypersurfaces (after A. Givental), Séminaire Bourbaki, vol. 1997/1998, Astérisque 252 (1998) 307–340, Exp. No. 848, 5.
- [67] F. Perroni, Chen-Ruan cohomology of ADE singularities, Internat. J. Math. 18 (9) (2007) 1009–1059.
- [68] A. Pressley, G. Segal, Loop Groups, Oxford Mathematical Monographs, The Clarendon Press, Oxford University Press, Oxford Science Publications, New York, 1986.
- [69] T. Reichelt, C. Sevenheck, Logarithmic Frobenius manifolds, hypergeometric systems and quantum D-modules, J. Algebraic Geom. 24 (2) (2015) 201–281.
- [70] Y. Ruan, The cohomology ring of crepant resolutions of orbifolds, in: Gromov–Witten Theory of Spin Curves and Orbifolds, in: Contemp. Math., vol. 403, Amer. Math. Soc., Providence, RI, 2006, pp. 117–126.
- [71] M. Thaddeus, Geometric invariant theory and flips, J. Amer. Math. Soc. 9 (3) (1996) 691–723.
- [72] The Stacks Project Authors, Stacks Project, http://stacks.math.columbia.edu, 2014.
- [73] H.-H. Tseng, Orbifold quantum Riemann–Roch, Lefschetz and Serre, Geom. Topol. 14 (1) (2010) 1–81.
- [74] J. Wise, The genus zero Gromov–Witten invariants of the symmetric square of the plane, Comm. Anal. Geom. 19 (5) (2011) 923–974.
- [75] J. Zhou, Crepant resolution conjecture in all genera for type A singularities, arXiv:0811.2023 [math.AG], 2008.